

REVIEW OF METHODOLOGIES AND INSTRUMENTS FOR DIAGNOSTICS OF TRANSPORT INFRASTRUCTURE



IM-SAFE



This project has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 958171. © 2021 IM-SAFE-Proiect | TNO. The Netherlands. All right reserved |



Document Information

Name	Remark
Document Name	Review of methodologies and instruments for diagnostics of transport infrastructure
Version No.	V1.0
Due date Annex I	15/07/2020
Report date	30/04/2022 (last updated in 29/04/2022)
Number of pages	158 excluding Appendixes
Lead Author	Monica Longo (SAC)
Other Authors	Serena Negri (SAC), Ana Sánchez Rodríguez (UVIGO), Ángel Santiago López Marroquín, (UVIGO), Andrés Justo Domínguez (UVIGO), Óscar Bouzas Rodríguez, (UVIGO), Patrycja Sanecka (MOW), Chiara Marchiori (IBM), Mattia Rigotti, (IBM)
Dissemination level	Public

Document history

Ver.	Date	Remark	Author	Checked by
Rev. 0.1	08/10/2021	Creation of the document	M. Longo	S. Negri
Rev. 0.2	11/02/2022	First draft	All	All
Rev. 0.3	16/02/2022	Second draft for revision	All	All
Rev. 0.4	17/02/2022	Amendments on content	All	A. Sánchez
Rev. 0.5	21/02/2022	Amendments on structure	A. Sánchez	
Rev. 0.6	28/02/2022	Document for reviewers	All	J. Köhler, K. Bergmeister and D. Allaix
Rev. 0.7	28/04/2022	Amendments from reviewers	M. Longo	S. Negri
Rev. 0.8	29/04/2022	Document for approval	All	F. Cinquini and A. Bigaj-van Vliet
Rev. 1.0	30/04/2022	Document Uploaded	All	A. Bigaj-van Vliet

Document approval

Ver.	Date	Approval	Project's Role	Beneficiary
Rev. 1.0	30/04/2022	A. Bigaj-van Vliet	Coordinator	TNO





Preface

This report is part of the H2020 CSA IM-SAFE project results and is the outcome of WP2 (Diagnostics of structures based on inspection, monitoring and testing) Task 2.2 (Review of methodologies and instruments for diagnostics of transport infrastructure) activities, listed as delivery D2.2. It constructs a part of the technical background for the formulation of the proposal for the mandate to CEN for a further amendment to the existing EU standards on monitoring and maintenance strategies, taking into account inspection, testing and monitoring data in safety assessment and maintenance approaches.

Task Leader: SAFECERTIFIEDSTRUCTURE INGEGNERIA S.R.L (SAC)

Contributors: UNIVERSIDAD DE VIGO (UVIGO),

MOSTOSTAL WARSZAWA S.A. (MOS)

IBM RESEARCH GMBH (IBM)

WP2 contributes to the identification of the normative gaps with regard to standards in monitoring and maintenance, damage detection and maintenance decision-making based on review of the current state-of-the-art as represented by standards, guidelines, other regulations as well as current practice and research. The particular focus of Task 2.2 is on diagnostics of structures based on survey data.

This report includes the review of the methodologies for inspection, the review of the damage detection indicators (DIs), procedures used for determining DIs based on condition survey data and KPIs from diagnostic testing and actions evaluation procedures. The conclusions from the review are used to formulate conclusion with regard to developments in standardization of concepts and methods for diagnostics of existing structures with includes a proposal for the development of a Damage Classification database.





Table of Content

DOCUMENT INFORMATION DOCUMENT HISTORY DOCUMENT APPROVAL PREFACE TABLE OF CONTENT	2 2 2 3
	4
	8
I ABLE OF I ABLES	10
1 PROBLEM STATEMENT	12
	10
	12
1.2 DIAGNOSTICS OF STRUCTURES BASED ON SURVEY DATA	13
1.3 IM-SAFE PROJECT SCOPE	14
1.4 OBJECTIVES OF THE DELIVERABLE	15
1.5 REPORT CONTENTS	15
2 OVERVIEW OF DAMAGE AND DEGRADATION MECHANISMS	17
21 GENEDAL	17
2.1 DENEINE 2.2 DAMAGE AND DECRADATION MECHANISMS	10
	20
	20
	22
2.3 DAMAGE CAUSES AND EFFECTS	23
2.3.1 DAMAGE CAUSES	23
2.3.2 ELEMENTS AFFECTED BY DAMAGE	25
2.4 BASIS FOR DAMAGE EVALUATION	26
2.4.1 DAMAGE EVALUATION APPROACHES	27
2.5 DAMAGE DETECTION PROCEDURE	28
2.5.1 DAMAGE IDENTIFICATION AND LOCALIZATION	28
2.5.2 DAMAGE QUANTIFICATION	28
2.5.3 DAMAGE MODELLING AND PREDICTION	28
2.5.4 DAMAGE MONITORING	28
2.6 DAMAGE AND PERFORMANCE INDICATORS	29
2.7 SYSTEM LIMITATIONS	30
2.8 SURVEY TECHNIQUES AND DATA ANALYSIS METHODS FOR DAMAGE IDENTIFICATION,	
LOCALIZATION, QUANTIFICATION, AND PREDICTION	31
2.8.1 DETERMINATION OF DAMAGE AND PERFORMANCE INDICATORS BY SURVEY TECHNIQ	UES 34
3 INSPECTION	40
	<u> </u>
	40
3.1 GENERAL	40
3.1.1 DEFINITION OF INSPECTION	40
3.1.2 INSPECTION'S CLASSIFICATION	40
3.1.3 KOLE OF INSPECTION IN THE EVALUATION OF A STRUCTURE HEALTH STATUS	40
3.1.4 BEST PRACTICES	41
3.2 KOUTINE MAINTENANCE INSPECTIONS	42
3.2.1 OBJECTIVES	43
3.2.2 DESCRIPTION	43
3.2.3 FREQUENCY OF EXECUTION	43





4 MONITORING

3.2.4	DATA COLLECTION	44
3.2.5	INSPECTION OUTCOMES	44
3.3 (CONDITION RATING INSPECTION	44
3.3.1	OBJECTIVES	45
3.3.2	DESCRIPTION	45
3.3.3	FREQUENCY OF EXECUTION	46
3.3.4	DATA COLLECTION	46
3.3.5	INSPECTION OUTCOMES.	46
3.4 E	EXTRAORDINARY INSPECTIONS	47
3.4.1	OBJECTIVES	47
3.4.2	DESCRIPTION	47
3.4.3	FREQUENCY OF EXECUTION	48
3.4.4	DATA COLLECTION	48
3.4.5	INSPECTION OUTCOMES.	48

50
50

4.1 4.2 4 3	GENERAL OBJECTIVES AND REQUIREMENTS MONITORING CLASSIFICATION	50 52 53
л.5 Л.Л		54
45		55
451		55
452	ANALYSIS OF THE STRUCTURE AND DEFINITION OF SPECIFIC OBJECTIVES FOR THE SYSTE	=M
1.0.2		
4.5.3	MEASUREMENT STRATEGY AND DEFINITION OF THE SENSOR NETWORK	56
4.5.4	DATA ACQUISITION AND PROCESSING	58
4.5.5	INTERPRETATIVE MODELS AND CALIBRATION OF REFERENCE MODEL	58
4.5.6	DECISION-MAKING PROCEDURES	59
4.5.7	ACCURACY REQUIREMENTS	59
4.6	FINAL DESIGN	60
4.7	INSTALLATION AND MANAGEMENT	61
4.7.1	INSTALLATION	61
4.7.2	TESTING AND COMMISSIONING	62
4.7.3	MANAGEMENT	62
4.8	MONITORING SYSTEM MAINTENANCE	62
4.9	MONITORING DATA ACQUISITION AND PROCESSING	62
4.9.1	DATA ACQUISITION	62
4.9.2	DATA PROCESSING	64
4.10	DATA TREATMENT	65
4.11	BRIDGES MONITORING	65
4.11.	1 Деск	65
4.11.	2 PIERS/ABUTMENTS	67
4.11.	3 BEARINGS	68
4.11.	4 JOINTS	69
4.11.	5 PRESTRESSING CABLES	70
4.11.	6 STAY-CABLES/HANGERS	70
4.12	TUNNELS MONITORING	71
4.12.	1 MONITORING SYSTEM DESIGN	71
4.12.	2 DISPLACEMENT MEASUREMENTS	72
4.12.	3 STANDARD TUNNEL MONITORING SYSTEM SECTION	74

5 DAMAGE AND DEGRADATION PROCESSES

76



5.1 DAMAGE AND DEGRADATION PROCESSES IDENTIFICATION	76
5.1.2 DESIGN AND CONSTRUCTION ISSUES	70
5.2 DAMAGE INDICATORS	77
5.2 SUBVEYING TECHNOLOGIES	77
5.4 DATA ANALYSIS METHODS	78
J.F. DATA ANALISIS MILITIODS	70
6 DAMAGE CHARACTERIZATION PROCEDURE	79
6.1 GENERAL	79
6.2 PHASE 1: DAMAGE DETECTION AND LOCALIZATION	79
6.3 PHASE 2: DAMAGE PARAMETRIZATION AND ASSESSMENT	80
6.4 PHASE 3: DAMAGE MODELLING AND PREDICTION	84
6.4.1 METHODS FOR DAMAGE MODELLING	84
6.4.2 PREDICTION OF DAMAGE EVOLUTION	85
6.5 PHASE 4: DAMAGE MONITORING	86
7 DAMAGE CLASSIFICATION	88
7.1 DAMAGE CLASSIFICATION PROCEDURES	88
7.2 PROCEDURES FOR INCLUDING DAMAGE INDICATORS IN EVALUATING KPIS	88
7.3 PROPOSAL FOR THE DEVELOPMENT OF A DAMAGE CLASSIFICATION DATABASE	91
8 ACTIONS EVALUATION	93
8.1 GENERAL	93
8.2 ACTIONS ON STRUCTURES	93
8.2.1 Bridges	94
8.2.2 TUNNELS	120
8.3 ACTIONS EVALUATION	125
8.3.1 Bridges	125
8.3.2 TUNNELS	128
8.4 USE OF DATA FOR ACTIONS MODELLING [SAC]	130
8.4.1 BRIDGES	130
8.4.2 TUNNELS	135
9 MODEL UPDATING BASED ON DIAGNOSTIC LOAD TESTING	137
	407
J.I GENERAL	13/
9.2 MODEL UPDATING	138
3.3 INTELHOUS OF CALIBRATION OF MODELS: DETERMINISTIC AND PROBABILISTIC MODELS	141
	142
	140
	140 170
J.4 FROCEDURES FOR IDENTIFTING RETS FROM DIAGNOSTIC TESTING	140
10 CONCLUSIONS	<u>151</u>
	150
	I JZ





12 APPENDIXES



Associated with document Ref. Ares(2020)3731189 - 15/07/2020

<u>158</u>





Table of Figures

Figure 2.1 – Examples for damage processes	18
Figure 2.2 – Damage causes	24
Figure 2.3 – Main types of actions	24
Figure 2.4 – Damage causes	25
Figure 2.5 – Correlation between damage causes and effects	26
Figure 2.6 – Correlation between observations, DIs, PIs and data coming from document	ts 30
Figure 2.7 – Conceptual weaknesses in a bridge deck (Extract from [4])	31
Figure 2.8 – Conceptual weaknesses in a precast multicellular cross section (Extract from	า [4]) 31
Figure 2.9 – Type of information in dependence of the different levels of investigation (Adju from [8])	isted 32
Figure 2.10 – Surveying techniques classification	33
Figure 4.1 – Examples for reasons for the use of monitoring systems (Adjusted from [8])	50
Figure 4.2 – Monitoring classification	53
Figure 4.3 – Diagram of the monitoring system design process (Adjusted from [16])	55
Figure 4.4 – Deck instrumentation (Adjusted from [19]).	67
Figure 4.5 – Piers/abutment instrumentation (adjusted from [20]).	68
Figure 4.6 – Bearings instrumentation (Adjusted from [21]).	69
Figure 4.7– Joints instrumentation (adjusted from [22]).	70
Figure 4.8– Prestressing cables instrumentation (adjusted from [23]).	70
Figure 4.9 – Stay-cable/hangers instrumentation (adjusted from [24] and [25]).	71
Figure 4.10 – Example of a measure of convergence as a function of time	72
Figure 4.11 – Typical arrangement of strain gauges for measurements of mass displacem [26]	ients 73
Figure 4.12 – Ground movement measurement by an inclinometer [27]	73
Figure 4.13 – Example of application of pressure cells (Extract from [28])	74
Figure 4.14 – Tunnels monitoring system instrumentation	75
Figure 6.1 – Damage characterization procedure phases	79
Figure 7.1 – Relationship diagram for KPI - PI – DI	89
Figure 8.1 – Adjustment of the snow load on the ground in relation to changes of the reperiod (Extract from [64])	eturn 99
Figure 8.2 – Directions of wind actions on bridges (Extract from [65])	101
Figure 8.3 – Example of the numbering of the Lanes in the most general case (Extract (EN-1991-2, 2005))	from 103
Figure 8.4 – Application of load model 1 (Extract from [66])	104
Figure 8.5 – Application of tandem systems for local verifications (Extract from [66])	105
Figure 8.6 – Application of load model 2 (Extract from [66])	105





Figure 8.7 – Fatigue Load Model 3 [66]	108
Figure 8.8 – Diagrammatic representation of constituent components of a temperature (Extract from [68])	profile 110
Figure 8.9 – Probabilistic collision model (Extract from [63])	114
Figure 8.10 – Collision force on supporting substructures near traffic lanes for bridge supporting structures for buildings (Extract from [70])	es and 116
Figure 8.11 – Recommended value of the factor rF for vehicular collision forces on hor structural members above roadways, depending on the clearance height h (Extract from	izontal m [70]) 116
Figure 8.12 – Impact loads on members of the superstructure (Extract from [70])	117
Figure 8.13 – Indicative impact area for ship impact. (Extract from [70])	118
Figure 8.14 – Ground types (Extract from [71])	119
Figure 8.15 – Loads on the segmental ring (Extract from [72])	121
Figure 8.16 – Hydrostatic pressure (Extract from [73])	124
Figure 8.17 – Load scheme (Extract from [73])	124
Figure 8.18 – Use of data for permanent actions	131
Figure 8.19 – Use of data for snow actions-rods/rulers	131
Figure 8.20 – Use of data for snow actions-snow pillows/scales, passive gamma rad	iations 131
Figure 8.21 – Use of data for wind actions	132
Figure 8.22 – General scheme (Extract from [4])	133
Figure 8.23 – Use of data for thermal actions	134
Figure 8.24 – Strategies for accidental design situations (Extract from [70])	135
Figure 8.25 – Use of data for soil actions-pressure plates	135
Figure 8.26 – Use of data for soil actions-GPR, investigation campaigns bibliographica	al data 136
Figure 8.27 – Use of data for hydraulic actions-pressure plates	136
Figure 8.28 – Use of data for hydraulic actions-piezometers	136
Figure 8.29 – Use of data landslide actions	136
Figure 9.1 – Changes over time of resistance and loads (Extract from [85])	137
Figure 9.2 – Model updating methods	138
Figure 9.3 – Model updating framework	140
Figure 9.4 – RAMSSHE€P approach (Extract from [90])	149
Figure 9.5 – Performance Indicators (PIs) and related Key Performance Indicators from diagnostic load testing.	(KPIs) 150





Table of Tables

Table 2.1 – Damage processes for reinforced concrete and steel structures [4]	19
Table 2.2 – Damage processes [4]	19
Table 2.3 – Degradation mechanisms of concrete (a and steel (b bridges and associa classes of defects [4]	ated 20
Table 2.4 – Classification of the defects of concrete and steel bridges [4]	21
Table 2.5 – Degradation mechanisms of tunnels and associated classes of defects [4]	22
Table 2.6 – Classification of the defects of tunnels [4]	23
Table 2.7 – Examples for each type of action	24
Table 2.8 – Classification of surveying techniques included in [11]	34
Table 2.9 – PIs vs Surveying technologies (for both bridges and tunnels)	35
Table 2.10 – Damage process vs PIs (bridges)	36
Table 2.11 – Damage process vs DIs (bridges)	37
Table 2.12 – Damage process vs PIs (tunnels)	38
Table 2.13 – Damage process vs DIs (tunnels)	39
Table 3.1 – Inspection similar to routine maintenance inspections in different countries	43
Table 3.2 – Frequency of execution of the inspections similar to routine maintena inspections	nce 44
Table 3.3 – Inspections similar to condition rating inspections in different countries	45
Table 3.4 – Frequency of execution of the inspections similar to condition rating inspecti	ons 46
Table 4.1 – Deck instrumentation.	66
Table 4.2 – Piers/abutments instrumentation.	67
Table 4.3 – Bearings instrumentation	68
Table 4.4 – Joints instrumentation.	69
Table 4.5 – Prestressing cables instrumentation.	70
Table 4.6 – Stay-cable/hangers instrumentation.	71
Table 6.1 – Surveying technologies summary table	84
Table 7.1 – Example of Pair-wise comparison matrix for a given PI	91
Table 7.2 – Example of Criteria weights of DIs for a given PI	91
Table 8.1 – Examples of potentially relevant actions for bridges and tunnels (Extract from	[4]) 94
Table 8.2 – Weight density of construction materials (Extract from [61])	95
Table 8.3 – Mean value and coefficient of variation for weight density (Extract from [63])	96
Table 8.4 – Mean values and standard deviations for deviation of cross-section dimensi from their nominal values	ons 96
Table 8.5 – Calculation procedures for the determination of wind actions (Extract from [6	65]) 100





Table 8.6 – Subdivision of the carriageway (Extract from [66])	102
Table 8.7 – Load model 1 - characteristic values [66]	104
Table 8.8 – Set of "frequent" lorries [66]	107
Table 8.9 – Set of equivalent lorries [66]	109
Table 8.10 – Definition of wheels and axles [66]	109
Table 8.11 – Recommended values of linear temperature difference component types of bridge decks for road, foot and railway bridges (Extract from [68])	for different 112
Table 8.12 – Indicative equivalent static design forces due to vehicular impact or supporting structures over or adjacent to roadways (Extract from [70])	n members 115
Table 8.13 – Indicative equivalent static design forces due to impact on supe (Extract from [70])	erstructures 115
Table 8.14 – Classes of structures subject to impact from derailed railway traffic (E [70])	Extract from 117
Table 8.15 – Most common sensors for bridge action evaluation	125
Table 8.16 – Most common sensors for tunnel action evaluation	125



1 Problem statement

1.1 Introduction

In recent years, we have faced rapid growth of testing, inspection, and monitoring technologies in various sectors. In the domain of transport infrastructure, intensive research has been carried out to enable the use of these technologies to support asset management of bridges and tunnels.

Accurate information from monitoring of structures is crucial to take the right decisions on maintenance and safety; unfortunately, there are gaps in the existing European standards and the monitoring practice at national level: accepted and harmonised approaches to condition survey and diagnostics of structures are lacking until now. This hinders asset owners and public authorities in charge of maintenance of the transport infrastructure to apply the latest development in maintenance strategies.

Moreover, structural monitoring is not addressed in the current Eurocodes (CEN/TC 250), and the existing standards on monitoring are not consistently interpreted and implemented in different European countries due to a lack of coherent policies and the gaps in knowhow.

The current standards do not embed the full extent of knowledge on analysis of extensive measurement and monitoring data to provide input for optimal maintenance strategies and safe operation of the infrastructure and on the adoption of digital technologies beyond the conventional inspection methods. The high diversity of transport infrastructure assets and their environments add to the complexity for standardised monitoring.

Specifically, on Structural Health Monitoring (SHM) only a few technical guidelines and standards available, which are only valid at the national level and very largely diverse in their approaches. Some examples are the Italian standard "UNI/TR 11634:2016. Linee guida per il monitoraggio strutturale"; the Austrian standard "RVS13.03.01. Monitoring von Brücken und anderen Ingenieurbauwerken"; the SAMCO "Guideline for Structural Health Monitoring"; the Chinese standard "GB 50982-2014 Technical code for monitoring of building and bridge structures"; and the Canadian "Guidelines for structural health monitoring". The guidelines provide the classification of the monitoring systems, goals of the monitoring activities, treatment of the monitoring data and damage identification techniques. However, the guidelines for SHM are often used incorrectly.

Foroptimal safety, availability, and cost-effectiveness of transport infrastructure, IM-SAFE envisions a paradigm shift from the time-based/corrective maintenance towards risk-based/predictive maintenance through data-informed decision-making enabled by a new and harmonised European standard for monitoring, including a standardised digitalisation approach. The new standard should be supported and implemented coherently by the public authorities and the industrial stakeholders across Europe.

The gathered data will be used for assessing the actual safety and the risk levels of the structure as well as for predicting the future safety and risks. This is currently possible by using the latest digital innovations that integrate structural models, predictive degradation models and data analytics techniques. The combined human expert and artificial intelligence will result in the transformation of measured data into knowledge about the performance and safety of the structure.

In this context, the advantages of performance monitoring will be fully exploited and compared to the conventional approaches based on visual inspections and non-destructive testing (NDT).





The expected result of IM-SAFE is, among others, to realise this vision by filling-in the gaps in the current standards and closing the gap between the standard and the practice. IM-SAFE will lead to new European standards for monitoring and maintenance of the structures, and the rules in the structural design codes (Eurocodes). The essential aspects that will be included are:

- Guidance about the physical parameters to be monitored and the performance and predictive analysis
- Guidance on how the monitoring data can inform the safety assessment and maintenance approaches
- Requirements on the algorithms used for damage identification
- Requirements on reliability, robustness, installation, operation, maintenance of sensors/monitoring systems

1.2 Diagnostics of Structures based on survey data

Condition survey seeks to gain an understanding of the current condition of the structure with the purpose of diagnostics. Diagnostics of a structure is a process of reviewing the structure and/or network-specific data gathered from inspection, monitoring and testing that allows the identification of the current condition of the structure., as well as the prediction of damage and prognosis of future performance. Diagnostics of structure comprises damage detection, actions evaluation and identification of system limitations. Data collection can be performed either manually or (semi-)automatically using a wide variety of technologies and techniques. The method and frequency of data collection depend on the type and size of the infrastructure assets owned/operated by a public authority or a company.

Structural Health Monitoring (SHM) to support maintenance strategies has been proposed during the last decades: embedded or external SHM encompasses Non-Destructive Testing (NDT) in order to provide damage detection and condition assessment throughout sensors and the analysis of the current condition of the structure.

Shortcomings of the actual practice and the gaps between current standards and practice to be tackled by IM-SAFE are:

- Many existing data collection, monitoring and inspection protocols are insufficiently covered in standards and the fragmentation in responsibilities; for instance, procedures of bridge maintenance lie within national legislations, but no specific standard on bridge monitoring is in place beyond some common practices.
- There is still an urgent necessity for the harmonisation of the developed techniques for SHM-based condition assessment and risk evaluation. Moreover, risk-based inspection and infrastructure monitoring standards are not yet unified through the European Union. Although there are international standards, they are not directly applicable in the EU countries because they do not address European-specific issues. Therefore, the development of standards and guidelines that guide the SHM of transport infrastructure is crucial.
- Consolidated Key Performance Indicators (KPIs) and common agreements of the methods for quantification of the relevant KPIs are needed in order to validate the monitoring and inspection actions and results.

In contrast to the traditional maintenance at predefined intervals, predictive maintenance responds more effectively to the changes of condition of the asset. In the IM-SAFE approach such a strategy can be supported by real-time monitoring of the conditions of a transport infrastructure asset in combination with predictive structural and degradation models. Combining a model-driven approach (using data in a predefined structural model) and a data-





driven approach (deriving a model from gathered data) will ensure optimal maintenance and safety.

Aligned with these technologies, IM-SAFE evaluates the methods (such as Modal Analysis (MA), statistical analysis methods, Big Data Analytics and Artificial Intelligence.) for the analysis of data obtained from monitoring and NDT technologies. To this end, new algorithms and relevant utilisation of Artificial Intelligence / Machine Learning for diagnostics incl. damage detection and condition monitoring need to be explored and standardised. Techniques such as anomaly detection and forecasting can be applied to time series dataset, such as measures collected by sensors (accelerometer, inclinometer, etc.) installed on civil infrastructures. In the case of anomaly detection, a machine learning (or deep learning) AI model can be trained to learn the "normal" (intended as reference) state of an infrastructure and then be able to detect "anomalies" (intended as deviations from the reference state) that might not be easily observable from a direct analysis of the data. In some cases, if the engineer has observed and annotated specific anomalies in previous time series, the AI model can be trained to detect specifically those anomalies, turning the problem to a classification exercise. For a large number of assets, the model creation for each specific asset can be automated, thus enabling large scale monitoring.

This new approach has been examined through a preliminary **S**trengths-**W**eaknesses-**O**pportunities-**T**hreats (**SWOT**) analysis, which showed that:

- The greatest strength lies in the quantitative recording of inventory and status data, which allows better condition predictions and savings through the efficient use of means in the life cycle.
- o The biggest weakness lies in the correct gradual implementation with targeted investments in research, software and IT as well as training and integration into the company processes.
- o The greatest opportunity lies in building up efficient, transparent decision-making systems as a modern operator and thereby leveraging considerable efficiency potential.
- o The greatest threats are poor strategies and implementation, which means that despite the investment, neither the desired efficiency gains nor the necessary acceptance in the company can be achieved.

Within this context, the objectives of task T2.2 are:

- To evaluate the methodologies and instruments currently available for diagnostics of structures, incl. damage detection.
- To identify the required technical guidance to incorporate the available data collection technologies and data analysis methods in diagnostics of structures.

1.3 IM-SAFE Project scope

Project scope and application domain of this report is consistent with the NMBP-36-2020 Call, which focuses on bridges and, where relevant, on tunnels. It is noted that the general concepts to a large extent apply to other types of infrastructure and transport infrastructure networks at large. This report is relevant to the domain of diagnostics of structures with a focus on the methodologies and instruments for diagnostics of transport infrastructure. Where relevant and feasible, differentiation is made between bridges and tunnels. With regard to the type of infrastructure, information, analysis, and conclusions presented in this report apply to modern European transport infrastructure (i.e. transport infrastructure constructed in 1960's or later). With regard to construction materials, concrete structures (i.e. civil engineering objects in plain, reinforced and prestressed concrete) and steel structures (i.e. civil engineering objects in steel) are covered, recognizing that to a large extent the fundamental considerations and





general procedures may apply to civil engineering objects for other types of structural materials.

1.4 Objectives of the deliverable

This report forms the technical background for the formulation of the proposal for the mandate to CEN for a further amendment to the existing EU standards on diagnostics of structures based on survey data and for a new standard for monitoring strategies. The activities for Task 2.2 in the context of WP2, which are documented are as follows:

- the review of the damage detection indicators and procedures, including principles of:
 - Identification of Damage Indicators (DIs), related to the road infrastructure Key Performance Indicators (KPI), to be measured and evaluated for the purpose of damage classification.
 - Procedures for determining DIs based on condition survey data.
 - Procedures for damage classification, accounting for type, size and location of defects or other relevant issues depending on the type of structure, the actions on structure, and the risks that may potentially affect the structure in the future, such as the one following from changes in traffic loads or service life demand, and from resilience issues related to climate change and increased use (ref. to EU projects like SAFEWAY).
 - Procedures for including DIs in evaluating the KPIs.
- Actions evaluation procedures, including principles of:
 - Procedures for determining actions on structures based on condition survey data.
 - Procedures for including actions determined from condition survey in risk and safety analysis.
- Identification of system limitations, including:
 - Procedures for determining deviations from design specifications based on condition survey.
 - Procedures for assessing KPIs from diagnostic testing.

1.5 Report contents

Chapter 2 provides an overview of damage and degradation mechanisms, with a focus on damage causes and effects and the proposal of a damage detection procedure based on existing damage evaluation approaches. Principles on damage and performance indicators and the description of survey techniques and data analysis methods for damage identification, localization, quantification and prediction are also provided.

In Chapter 3 a classification of inspection methodologies is given. Each type of inspection is described with respect to its objectives, frequency of execution, data collection methods and outcomes.

Chapter 4 introduces the objectives and requirements of monitoring systems and their classification based on the period of execution and on the purpose. Monitoring system architecture and design process are described. Additionally, guidance on monitoring system installation and management and data acquisition, processing and treatment is given.

Chapter 5 is focused on damage and degradation processes. For each damage process the description, the identification of related DIs and PIs and of the survey techniques used for the detection is given. Survey techniques are classified based on a colours rating scale with respect to selected parameters (e.g., availability, cost, efficiency). Data analysis methods related to each damage process are also given.





Chapter 6 describes the damage detection procedure introduced in chapter 2, providing for each phase (damage identification and localisation, quantification, prognosis and monitoring) the description of the decision-making procedure.

In Chapter 7 a procedure for damage classification and for the development of a damage classification database is given.

Chapter 8 focuses on actions evaluation, including general information on each type of action, the surveying technologies used for their evaluation and principles on how to use data for actions modelling.

Lastly, in Chapter 9 a description of the decision-making process for model updating purposes, model calibration methods and the procedures for assessing KPIs from diagnostic testing is provided.

Summary and conclusions of the report are given in Chapter 10.





2 Overview of damage and degradation mechanisms

2.1 General

In the context of the IM-SAFE project, "**damage**" is defined as a disruptive change in the condition of a structure, structural components or structural members that can unfavourably affect its current of future structural performance. Aiming to classify and describe damage and degradation processes, the following distinction has been made ([1]).

- **Degradation**, defined as the worsening of the performance of material over time, as the result of (complex interaction between) degradation mechanisms, which are (chemical, physical, mechanical, biological or multi-type) root cause of damage.
- Deterioration, which is defined as the progressive reduction in the ability of a structure or its components to perform according to their intended functional specifications. Typically, deterioration of a structure or its components will be driven by degradation of materials.

Structures, indeed, are inevitably affected by damage processes, due to which deterioration processes may arise and, therefore, lead to durability, safety, and serviceability problems. of both single and/or multiple elements. Hence, it is crucial to optimize inspection and maintenance strategies based on reliable information on the processes.

According to [2] and [3] damage processes can be classified as follows:

- They may affect structures and structural components at the local or global scale.
- They may act singly or in combination.
- They may be gradual and observable or gradual and non-observable.
- They may have a chemical, physical, biological, or mechanical origin or even a combination.

Basically, a damage assessment should consider the nature, the intensity, the extent, and the location of each damage process. In this respect, since deterioration processes have different origins and acting timespan, a damage detection procedure is proposed in [2.5], in order to analyse data for defining indicators that are capable of evaluating the effects of acting identified deterioration processes. Some examples for damage processes are provided in Figure 2.1.









a) Corrosion



c) Fatigue

b) Alkali-aggregate reactions



d) Freeze-thaw

Figure 2.1 – Examples for damage processes

2.2 Damage and degradation mechanisms

This section overviews the principal damage and degradation mechanisms that affect structures, resuming the classification of damage processes and the distinction between concrete and steel structures made in [4].

Table 2.1 – Damage processes for reinforced concrete and steel structures

, for instance, shows a general overview of chemical, physical, and biological processes considered in the context of the IM-SAFE Project.

	Chemical	Physical/Mechanical	Biological/Organic
	Alkali-aggregate	Freeze-thaw	Living organisms' activity
	reaction (AAR)	Creep	Accumulation of
	Internal sulphate attack (ISA)	Shrinkage	dirt or rubbish
rete	External sulphate attack (ESA)	Thermal cracking	Oil and fat contamination
onc	and salt crystallisation	Abrasion/ Erosion	
ŏ	Carbonation	Fire	
	Chloride contamination	Overloading	
	Leaching	Fatigue	
	Acid attack		



Reinforcement steel uniform and pitting corrosion Fracture of prestressing steel (Steel dissolution due to) Stray current





Fatigue

Steel Structures	Chloride contamination Leaching Acid attack	Reinforcement steel uniform and pitting corrosion Fracture of prestressing steel (Steel dissolution due to) Stray current Fatigue	Living organisms' activity Accumulation of dirt or rubbish
		- sugue	

Table 2.1 – Damage processes for reinforced concrete and steel structures [4]

Table 2.2 shows the damage processes with respect to the materials and its impact on the structure.

	Proposed Damage Processes		Material			Impact			
N°			Steel	Masonry	Change in geometry	Change in integrity	Change in material properties	Change in actions	
1	abrasion	•	•	•	•	•			
2	aggradation (alluviation)	•	•	•				•	
3	erosion	•	•	•	•	•		•	
4	pitting corrosion	•	•		•	•	•		
5	changing geotechnical conditions	•	•	•	•	•		•	
6	aging of material	•	•	•	•	•	•		
7	alkali aggregate reaction	•			•	•	•		
8	chemical action	•	•	•	•	•	•		
9	corrosion related to prestressing steel	•	•		•	•	•		
10	corrosion related to reinforcement steel	•			•	•	•		
11	corrosion related to structural steel		•		•	•	•		
12	fatigue	•	•			•	•		
13	sulphate reaction	•			•	•	•		
14	corrosion related to equipment made of steel	•	•		•	•	•		
15	corrosion related to fixings, connectors	•	•		•	•	•		
16	overloading of an element	•	•	•	•	•		•	
17	biological growth	•	•	٠	•	•	•	•	
18	freeze-thaw	•		•	•	•	•		
19	high temperature		•		•	•	•		

Table 2.2 – Damage processes [4]





2.2.1 Bridges

There are three main groups of degradation mechanisms in connection with physical, chemical and biological phenomena. Table 2.3 shows these groups in detail with regard to the correlated damage classes for bridges.

DEGRADATION MECHANISMS	CLASS OF DEFECTS						
		Deformation	Destruction	Loss of material	Discontinuity	Contamination	Displacement
Physical	Accumulation of inorganic dirtiness ^{(a, (b)} Cyclic freeze–thaw action ^{(a, (b)} Erosion ^{(a, (b)} Crystallization ^(a) Extreme temperatures ^{(a, (b)}						
	Creep ^{(a, (b} Relaxation ^{(a, (b} Shrinkage ^{(a} Overloading ^{(a, (b} Fatigue ^{(a, (b}						•
	Geolechnical condition changes	•			-		
Chemical	Carbonation ^{(a} Corrosion ^{(a, (b} Aggressive compounds action ^{(a, (b} Chemical dissolving/leaching ^{(a, (b} Reactions between material						
Biological	Accumulation of organic dirtiness ^{(a, (b)} Activity of microbes ^{(a, (b)} Activity of plants ^{(a, (b)} Activity of animals ^{(a, (b)}						

Table 2.3 – Degradation mechanisms of concrete (a and steel (b bridges and associated classes of defects [4]

Defects associated with bridges can be subdivided into classes, types and categories. Table 2.4 shows a breakdown of defects into classes, types and categories for bridges, which can be used for a systematic arrangement of information about damage processes in bridges.

CLASS OF DEFECT	TYPE OF DEFECT	CATEGORY OF DEFECT			
Deformation	Incorrect geometry of	Incorrect shape of concrete or steel profiles ^{(a, (b}			
	constructed element	Invalid arrangement of reinforcement or of bolts, rivets, welds ^{(a, (b}			
		Invalid arrangement of prestressing tendons ^{(a}			
	Change of the geometry of the element axis	Excessive elastic deformation ^{(a, (b}			
		Permanent deformation ^{(a, (b}			
	Change of the geometry	Excessive elastic deformation ^{(a, (b}			
	along the element length	Permanent deformation ^{(a, (b}			
Destruction	Change of the chemical characteristics	Change of concrete characteristics ^{(a}			



Page | 20



Associated with document Ref. Ares(2020)3731189 - 15/07/2020

of material	Change of the physical characteristics	Change of reinforcing steel characteristics ^(a) Change of prestressing steel characteristics ^(a) Change of protective layer characteristics ^(a) , ^(b) Change of profile steel or steel slabs/walls characteristics ^(b) Change of bolts, rivets, welds characteristics ^(b) Change of concrete characteristics ^(a) Change of reinforcing steel characteristics ^(a) Change of prestressing steel characteristics ^(a) Change of protective layer characteristics ^(a) Change of profile steel or steel slabs/walls characteristics ^(b)
Loss of material	Loss of structural material	Loss of concrete ^{(a} Loss of reinforcing steel ^{(a} Loss of prestressing steel ^{(a} Loss of profile of steel or steel slabs/walls ^{(b} Loss of bolts, rivets, welds ^{(b}
	Loss of the material of the protective layer	Loss of material of concrete protection ^{(a} Loss of protection of reinforcing steel ^{(a} Loss of protection of prestressing steel ^{(a} Loss of protection of profile steel or steel slabs/walls ^{(b} Loss of protection of bolts, rivets, welds ^{(b}
Discontinuity	Crack	Crack of concrete ^{(a} Crack of reinforcing steel ^{(a} Crack of prestressing steel ^{(a} Crack of protective layer ^{(a, (b} Crack of profile steel or steel slabs/walls ^{(b} Crack of bolts, rivets, welds ^{(b}
	Fracture	Fracture of concrete ^{(a} Fracture of reinforcing steel ^{(a} Fracture of prestressing steel ^{(a} Fracture of protective layer ^{(a, (b} Fracture of profile of steel or steel slabs/walls ^{(b} Fracture of bolts, rivets, welds ^{(b}
Contamination	Inorganic	Aggressive ^{(a, (b} Neutral ^{(a, (b}
	Organic	Aggressive ^{(a, (b} Neutral ^{(a, (b}
Displacement	Incorrect linear displacement	Excessive movement ^{(a, (b} Restricted movement ^{(a, (b}
	Incorrect rotation	Excessive movement ^{(a, (b} Restricted movement ^{(a, (b}

Table 2.4 – Classification of the defects of concrete and steel bridges [4]





2.2.2 Tunnels

There are three main groups of degradation mechanisms in connection with physical, chemical and biological phenomena that can also be applied to tunnels. Table 2.5 shows these groups in detail with regard to the correlated damage classes for tunnels.

	CLASS OF DEFECTS						
DEGRADATION MECHANISMS		Deformation	Destruction	Loss of material	Discontinuity	Contamination	Displacement
Physical	Accumulation of inorganic dirtiness						
	Cyclic freeze-thaw action		•	•			
	Erosion						
	Creep						
	Relaxation						
	Shrinkage				•		
	Overloading			•	•		•
	Fatigue		•				
	Geotechnical condition changes				•		
	Temperature						
	Wet areas						
Chemical	Carbonation		•				
	Corrosion		•	•			
	Aggressive compounds action		•				
	Chemical dissolving/leaching		•			•	
	Reactions between material components		-	-			
Biological	Accumulation of organic dirtiness					•	
	Activity of microbes					•	
	Activity of plants					•	
	Activity of animals						

Table 2.5 – Degradation mechanisms of tunnels and associated classes of defects [4]

Note:
-basic degradation mechanism,
-additional degradation mechanism

Defects associated with tunnel structures can be, as for bridges, subdivided into classes, types and categories, which can be as mentioned before for bridges used for a systematic arrangement of information about damage processes (Table 2.6).

CLASS OF DEFECT	TYPE OF DEFECT	CATEGORY OF DEFECT		
Deformation	Incorrect geometry of	Incorrect shape of concrete component		
	constructed element	Invalid arrangement of reinforcement		
		Permanent deformation**		
	Change of the geometry along the element length	Permanent deformation		
Destruction	Change of the chemical	Change of concrete characteristics		
of material	characteristics	Change of reinforcing steel characteristics		





Associated with document Ref. Ares(2020)3731189 - 15/07/2020

	Change of the physical	Change of concrete characteristics
	characteristics	Change of reinforcing steel characteristics
		Ice formation due to water
		damages on joint tape
		damages on sealing
Loss of material	Loss of structural material	Loss of concrete***
		Loss of reinforcing steel
Discontinuity	Crack	Crack of concrete****
		Crack of reinforcing steel
		Differentiate cracks, if necessary, as different effects/causes are associated with them
	Fracture	Fracture of concrete****
		Fracture of reinforcing steel
		Take into account the void in the inner shell
		Less thick inner shell
		insufficient concrete cover
		spalling/detaching
		wet area (e.g. through cracks, joint tapes in structures that retain pressurized water, access due to damage to the plastic sealing membrane in structures that relieve pressurized water)
		drainage damages
Contamination	Inorganic	Aggressive
		Neutral
		Efflorescence/washout
	Organic	Aggressive
		Neutral
Displacement	Incorrect linear displacement	Excessive movement
		Restricted movement
	Incorrect rotation	Excessive movement
		Restricted movement

Table 2.6 - Classification of the defects of tunnels [4]

2.3 Damage causes and effects

2.3.1 Damage causes

Damage of structures may arise for several reasons, which can be grouped in the following clusters, represented in Figure 2.2.

- Actions:
 - Accidental actions;
 - Environmental actions;
 - Operating conditions.
- Deterioration processes;
- Construction and design errors:
 - Construction errors;
 - Design errors.
- Change of use.







Figure 2.2 – Damage causes

2.3.1.1 Actions

The three main types of actions are represented in Figure 2.3:



Figure 2.3 – Main types of actions

Some	examples	for eac	ch type	are	given	in	Table 2	2.7.
					3			

ACCIDENTAL	 Accidental external or internal gas explosions Accidental impact by various forms of vehicle,
ACTIONS	including road vehicles, aeroplanes, trains, etc.
ENVIRONMENTAL ACTIONS	Snow actionsWind actionsThermal actions
OPERATING	 Overloading due to misuse Overloading caused by exceptionally strong winds or
CONDITIONS	heavy snow Fatigue

Table 2.7 – Examples for each type of action

2.3.1.2 Degradation/deterioration processes

As outlined in paragraph 2.2, structures are affected by degradation and damage processes, which can lead to durability, safety, and serviceability issues. Degradation may occur due to





chemical, physical, mechanical, and biological processes, which are summarized in 2.2.1 and 2.2.2 for bridges and tunnels, respectively.

2.3.1.3 Construction and design errors

Errors can occur during the design and/or construction stages. Moreover, it should be noted that design errors are not always detected before construction has begun, and this leads to significant consequences. According to [5], most of these types of errors are due to lack of understanding of basic engineering methods, inadequate development of details, or sometimes to last-minute changes without proper assessment of the consequences of these changes. Construction errors may be related to inadequate understanding of the design concept or to last-minute changes in the construction sequence.

2.3.1.4 Change of use

Every structure is designed for a specific use and must meet specific requirements for serviceability, structural safety, durability, sustainability, and reliability during the entire required service life. If that use changes, the structure may not meet the new requirements.

This may concern in particular bridges, which, for instance, when subjected to a change of the road category, may not bear the increase of variable loads (e.g., traffic loads) and therefore damage may occur.

2.3.2 Elements affected by damage

In section 2.3.1 damage causes are categorized. Consequently, a categorization of the levels on which damage may act is given here (Figure 2.4):

- Material;
- Section;
- Structural component;
- Structural system.



Figure 2.4 – Damage causes

Each of the causes identified above may have different consequences with respect to their level on which the damage acts:

- Actions (accidental, environmental, operating) can affect both material, section structural components and system.





- Degradation/deterioration processes (chemical, physical, biological) can affect both material, section and structural component.
- Construction and design errors can affect both material, section, structural components and system.
- Change of use can affect both material, section structural components and system.

The relationship between each damage cause and the relative effects is summarized in Figure 2.5.

	CALLSE			EFFECT						
	MATERIAL	SECTION	STRUCTURAL COMPONENT	STRUCTURAL SYSTEM						
ACTIONS	ACCIDENTAL ACTIONS ENVIRONMENTAL ACTIONS OPERATING CONDITIONS	×	×	×	×					
DEGRADATION/ DETERIORATION PROCESSES	CHEMICAL PHYSICAL BIOLOGICAL/ORGANIC	×	×	×						
CONSTRUCTION & DESIGN ERRORS	CONSTRUCTION ERRORS DESIGN ERRORS	×	×	×	×					
CHANGE OF USE	•	×	×	×	×					

Figure 2.5 – Correlation between damage causes and effects

2.4 Basis for damage evaluation

Indicators allowing for assessment of structural integrity, state of deterioration or condition assessment can be derived in various ways from experimental data: significant changes in the data relating to the condition of the structure may be indicative of a developing failure. However, such deductions can only be established through an adequate process of data assessment and evaluation, which enables an appropriate context-based engineering interpretation to be made of the indications obtained.

Damage evaluation, therefore, is based on a condition assessment, which relies on the potential to capture deviation from a reference condition, usually represented by a specified set of limit states, which separate desired states of the structure from adverse states. Beyond serviceability and limit states criteria, in fact, it is possible to define limit states based on the condition on the structure, such as the partial damage limit states [6]. The main system properties or a model that reproduces the system response are determined based on a **system identification** process, which, according to [3], often refers to the estimation of physical model parameters, such that the existing numerical model best reflects the structural behaviour observed experimentally.

The Damage Indicators (DIs) may be used to capture deviations from the reference condition, considering directly the monitored variables (e.g. strain) measured by means of dedicated sensors or tests, or more indirect properties, for instance continuous response functions or modal parameters.

Depending on selected Dis, further damage measurements may be performed with visual and/or in-depth inspections. The choice of the indicators and their combined evaluation aims to provide a univocal, exhaustive, and as much as possible unambiguous interpretation of the actual structural behaviour.

Damage identification methods can be classified in the following categories:





- Data-driven approaches;

- Model-based approaches.

Data-driven and model-based methods for damage detection, localization, characterization and prediction are illustrated in 2.4.1.1 and 2.4.1.2.

The identification of damage generally is described by the indication of a four Level process:

- 1. Damage identification and localization;
- 2. Damage quantification;
- 3. Damage modelling and prediction;
- 4. Damage monitoring.

Damage identification usually requires the analysis of responses recorded at several locations, i.e., a higher spatial resolution in the description of the structural condition but at a certain extent can still be performed with a data driven approach provided spatially distributed damage features, for example modal shapes, are used.

The evaluation of the type and severity of damage, typically requires a model-based approach to map the monitored response to different damage types and scenarios.

2.4.1 Damage evaluation approaches

The observation of the structure over time is obtained by measurements, which should be analysed to determine the current state of the structure and to reflect the capability of a structure to continue to perform its intended function after being exposed to aging and the accumulation of damage. Since damage significantly alters the properties of the structure, information gathered from measurements and SHM monitoring systems are key factors in damage evaluation. Model-based methods, as well, can be used for real-time damage identification, updating significant parameters of a numerical or finite element model.

Hence, damage evaluation can be performed with both data-driven approaches and modelbased approaches, which are described in the following paragraphs.

2.4.1.1 Data-driven approaches

Data-driven methods extract damage features relying solely on the recorded response and rely on picking up shifts of or differentiation in damage sensitive features, with respect to those corresponding to regular operating conditions. These methods are hence attractive for adoption within real-time damage identification but the absence of a numerical model of the system often hampers the estimation of damage severity.

On the purely data-driven front, the advent of advanced and increasingly affordable monitoring technologies has allowed for continuous monitoring of structures based on use of heterogeneous sensors, which are able to deliver a "Big Data" stream of monitored information. Naturally, the treatment of this data requires suited processing tools, which are typically tied to data-driven time-series analysis, statistical methods and Machine Learning (ML) schemes.

2.4.1.2 Model-based approaches

Model-based methods most typically rely on the use of the damage features to update significant parameters of a numerical, often finite element (FE), model. They are usually less attractive for real-time damage identification, due to the computational cost of the updating process, but the availability of a FE model allows for higher-end damage characterization and, if a degradation model is available or can be defined based on monitored data, they can further provide salient information regarding the remaining service life of the structure.





In model-based methods, damage is detected through the updating of the parameters of a numerical and/or probabilistic model, in which the parameters related to resistance and actions can be modelled with probability density functions.

The numerical model, calibrated and updated using sensors measurement, can be used to explore several critical scenarios that are representative of the limit conditions for which the thresholds must be defined. In the practical applications, thresholds. should be consistent with the ultimate or serviceability requirements and, additionally, consider the continuously increasing loss of function due to the damage response characteristics of the structures. In order to maintain a prescribed performance level, it is possible to distinguish between:

- Attention thresholds
- Alarm thresholds

Additional threshold levels characterized by different severity degree may be considered based on specific needs (e.g., road operators and asset owners operational requirements).

When a model is inserted in the analysis loop, it is possible to more straightforwardly achieve higher end SHM tasks, such as localization and quantification, albeit at the cost of higher computational effort.

For further information on model updating techniques, see Chapter 8 of [4].

2.5 Damage detection procedure

The purpose of the damage detection is to determine whether the structure under examination is affected by a damage process, and which are the causes for the occurrence of damage.

Hence, based on the existing literature ([7], [3]), the following damage detection procedure is proposed and envisages a four Level process:

- 1. Damage identification and localization;
- 2. Damage quantification;
- 3. Damage modelling and prediction;
- 4. Damage monitoring.

2.5.1 Damage identification and localization

The first step for the damage detection procedure consists in the **identification** of the type of process(es) responsible for the damage, which can have a chemical, physical, biological, or mechanical origin or even act in a combined way. It also consists in the **damage localization**, which is defined as the process, deterministic or probabilistic, of ascertaining where the damage to structure is located.

2.5.2 Damage quantification

The second step for the damage detection procedure consists in the **damage characterization**, which is defined as the process, deterministic or probabilistic, of determining the time of occurrence, , the size and other features of the damage, such as the origin of the damage and the current progress in the damage phenomenon (e.g., carbonation front depth, crack width, etc.).

2.5.3 Damage modelling and prediction

The third of the procedure consists of damage modelling and prediction, which aims to control the evolution of damage over time.

2.5.4 Damage monitoring

The last step of the procedure consists of the damage monitoring.





2.6 Damage and Performance Indicators

Damage indicators (Dis) and performance indicators (Pis) are used for the implementation of management strategies of infrastructure networks and infrastructure objects, which must be kept at a desired performance level.

These indicators can be qualitative or quantitative based, and they can be obtained during principal inspections, through a visual examination, a non-destructive test or a temporary or permanent monitoring system: these values are the basis for the assessment of bridges and tunnels state condition.

As outlined in [4], damage processes, if detected, must be documented as observations by the bridge inspectors. Inspectors are also responsible for observing and assessing bridges/structures to determine those damage processes that are most likely to affect the structure or its components: the inspector must make a meaningful diagnosis with the help of the observations and knowledge of damage processes.

Relevant observations taking place during an investigation can be made through the perception of the human senses or through data measured by instruments. Determining the absence or presence of a property belongs to the group of qualitative observations and measuring or counting the observed phenomenon belongs to the group of quantitative observations. The fitness of purpose of a bridge or a structure can be measured by a performance indicator (PI).

Successive inspections allow a distinction whether it is (a) a pure observation (e.g. stable crack) or (b) a possible PI (e.g. growing crack). Thus, there is the need of differentiation between observations and Pis.

For further information about Performance Indicators and their categorization, these concepts are thoroughly described in [4].

Hence, in the context of the IM-SAFE Project, with regard to the data used in a performance assessment, the following distinction has been made:

- Data coming from documents
- Observation: a datum from a primary source, which may be acquired by human senses or by measuring/recording of some properties via adequate instruments. Observations can be qualitative (i.e., only the absence or presence of a property is noted), or quantitative if a numerical value is coupled to the observed phenomenon by counting or measuring.
- Performance Indicator: an observation or a parameter derived from observations that quantitatively describes property of the structure and/or of the aspect of its performance and are used to qualify fitness of the structure for its purpose during service life.
- Damage indicator: an observation or a parameter derived from observations that serves for quantitative or qualitative damage detection, damage localization and/or damage characterization.
- Monitoring parameters

The categories of data described above enable the characterization of the performance of the structures.

Once it has been assessed that a damage process, whether active or inactive, is occurring, then it could be identified with a damage indicator (DI), which may coincide with a performance indicator and/or an observation or not.





Observations, Pis and Dis can be obtained directly from the raw data or as a result of a data processing process. In Figure 2.6 the relationship between the different types of data is represented: it is possible to see that observations, PIs, DIs, and data coming from documents are closely related, since both observations and data coming from documents sets overlap part of the PIs and Dis ones. The data processing boundary line separates the area on the chart where raw data can be used from the area where post processing is required.

Parameters to be monitored can be selected by the data set of observation, PIs and DIs.



Figure 2.6 - Correlation between observations, DIs, PIs and data coming from documents

It should be noted also that all observations collected through inspection and maintenance activities must be included in reports, which then become also part of the data coming from documents set.

2.7 System limitations

Diagnostics of structures should consider also risk arising from system limitations, since it can increase due to degradation and damage processes. The concept of "system limitations" is not related to loads acting on structures, and neither are to deterioration mechanisms. It is directly related, instead, to the intrinsic properties of the structures and therefore to their design and construction and to the vulnerability, which is defined as the degree to which a system, or part of it, may react adversely during the occurrence of a hazardous event. This concept implies a measure of risk associated with the physical, social and economic aspects and implications resulting from the system's (e.g. bridges or tunnels ability to cope with the resulting event).

Hence, each structure is characterized by vulnerable zones, which can be detected with several methods [4]. A typical example is a Gerber hinge in a girder or frame bridges. They consist in a Gerber deck, suspended span or reduced end beam, supported on nibs of abutments or adjacent beams. Their position influences the magnitude and the distribution of stresses and the sensitivity to actions such as differential settlements or thermal variations.







Figure 2.7 – Conceptual weaknesses in a bridge deck (Extract from [4])

Another example of conceptual weakness may be associated to cross-sections, such the precast multicellular one:



Figure 2.8 – Conceptual weaknesses in a precast multicellular cross section (Extract from [4])

The limitation lies in the fact that these types of elements, generally, can hardly be inspected without a traffic disruption, so, if not properly designed and/or protected during construction and operation, costly interventions are needed.

Locations of vulnerable zones in girder and frame bridges are presented in [4].

2.8 Survey techniques and data analysis methods for damage identification, localization, quantification, and prediction

The assessment of a deteriorating structure requires the use of inspection and/or testing techniques to define the mechanism(s) and severity of any structurally significant degradation, as well as to be able to make a prognosis about future condition. Defects associated with tunnel and bridge structures can be subdivided into classes, types and categories defined in [4].

Data about the structure can be gathered in several ways by the following types of activities:

- Inspections
- Survey
- On-site testing and measurements
- Material sampling
- Laboratory testing
- Monitoring

Each type of information that can be obtained from the activities above represents a specific level of detail: the relationship between the expense, in terms of extent and number of measurements, relative to the insight/information gained is represented in Figure 2.9.

The execution of good quality tests is crucial for a correct decision-making in the perspective of condition control and optimum maintenance.



Page | 31





(extent and number of measurements)

Figure 2.9 – Type of information in dependence of the different levels of investigation (Adjusted from [8])

According to [9], locations of the inspection, survey, testing, investigation, monitoring, or other information gathering or condition monitoring activities should be carefully selected so that the desired information about the current condition and deterioration of materials and/or structural performance can be obtained, considering factors such as:

- The likely mechanism(s) and rate of deterioration
- The environmental conditions
- The conservation strategy and tactics, together with the inspection, testing, investigation and monitoring regimes defined at the time of design or re-design.

Three different inspection levels can be distinguished:

- **I. Preliminary assessment (Visual Inspection)**: consists of a preliminary assessment of the structure based on information obtained through inspections, study of the available historical project documentation and a rough check on the structural safety.
- **II. Detailed investigation**: consists of detailed inspections, tests and monitoring. The detailed inspection of the structure, which aims to dispel or confirm any doubts as to whether the structure is safe.
- **III. Assessment and prediction by advanced analysis**: includes structural monitoring and modelling. This inspection level might be used for problems with substantial consequences, for which an advanced analysis may be needed.

For further information on inspection levels and their use in performance assessment, see Chapter 9 of [10].



Page | 32



Each inspection level is based on the use of surveying techniques, which can be classified based on several criteria.

Firstly, based on the impact of testing on structures, it is possible to distinguish:

- Destructive techniques
- Non-destructive techniques
- Semi-destructive techniques

Based on the environment in which tests are performed, it is possible to distinguish:

- In-situ surveys
- Laboratory investigations
- Remote analysis

Based on the object under investigation, instead, the following categorization is made:

- Local analysis
- Global analysis

Lastly, concerning the type of information, data can be qualitative or quantitative.

In the context of IM-SAFE project, surveying technologies are categorized as summarized in Figure 2.10.



Figure 2.10 – Surveying techniques classification

Hence, surveying techniques are divided in the following main categories based on technology involved.:

- **Sensors** involve destructive and non-destructive techniques, which include either IoT and sensor systems/remote sensing or physical/chemical methods, respectively.
- **Platforms** include satellite, RPAs/UAV and mobile mapping systems.





The classification of surveying techniques described in [11] based on the principles described above is summarized in Table 2.8.

CLASSIFICATION SYSTEM																			
		TECHNOLOGY		ІМРАСТ			TYPE O	F DATA	DA ACQUI	TA SITION	METHOD OF TESTING		TYPE OF METHOD		ENVIROMENT OF TESTING			ANA	LYSIS
		SENSORS	PLATFORMS	DT	NDT	SDT	QUALITATIVE	QUANTITATIVE	IoT AND SENSORS SYSTEMS	REMOTE SENSING	ACTIVE	PASSIVE	PHISICAL METHODS	CHEMICAL METHODS	IN SITU	IN LAB	REMOTE ANALYSIS	GLOBAL	LOCAL
	LIDAR	0			•		•	•		0	•		0		•			•	•
	SATELLITE		•		•		•	•		0	•	0	0				0	0	
	GPR	0			•		•	•		0	0		0		0			•	0
	RPAs-UAV		0		•		•	•		0	0		0		0		0	0	0
	ACCELEROMETERS	•			•			•	•				•		•		•	•	
S	CLINOMETERS	•			•			•	•				•		•		•	•	
OGIE	CRACKMETERS	0			•			•	•				0		0		0		•
IONH	MECHANICAL TESTS	0		0			•	•		0			0		0	0			0
IG TEC	ENDOSCOPY	0			•	•	•			0		0	0		0				•
VEYIN	BOROSCOPY	0			•	•	•			0		0	•		0				0
SUR	GUIDED WAVES TECHNIQUE	•			•		•	•		•		0	•		•		•	•	•
	ACOUSTIC EMISSION	0			•		•	•		0	•		•		0		0	•	0
	FIBRE OPTIC SENSORS	•			•		•	•	•		•	0	•				•	•	•
	WATER RESISTANCE AND PENETRATION	•		•			•	•		0		0		0		0			•
	RADIOLOGICAL METHODS	•		•			•	•		•		•	•			•			•
	CHEMICAL METHODS	•		•			•	•		•		0		•	•	•			•
	MAGNETIC AND ELECTRICAL METHODS	0			•		•	•		0	•	•	•	•	0	•			•

Table 2.8 – Classification of surveying techniques included in [11]

2.8.1 Determination of Damage and Performance Indicators by survey techniques

The section aims to correlate the following information:

- Performance Indicators (PIs),
- Damage Indicators (Dis)
- Damage processes
- Surveying techniques
- Analysis phase.

For this purpose, the following tables have been built, summarizing information included in Annex A:





PIs vs SURVEYING TECHNOLOGIES, which also includes a colour scale regarding the analysis phase in which the specific technology is being used.

> DAMAGE PROCESS vs DIs

> DAMAGE PROCESS vs Pls

The combined use of the tables above allows to define which are the best surveying techniques to be used in order to identify each damage process and which are the related Damage and Performance Indicators.

		SURVEYING TECHNOLOGIES																	
	 LEGEND INSPECTION DETAILED INSPECTION MONITORING 	LIDAR	SATELLITE	GPR	RPAS-UAV	ACCELEROMETERS	CLINOMETERS	CRACKMETERS	FIBRE OPTIC SENSORS	MECHANICAL TESTS	ENDOSCOPY	BOROSCOPY	GUIDED WAVES TECHNIQUE	ACOUSTIC EMISSION	WATER RESISTANCE AND PENETRATION	RADIOLOGICAL METHODS	CHIMICAL METHODS	MAGNETIC AND ELECTRICAL METHODS	
	Rupture	••	••	••	••				•	• •			•	••		•		•	
	Holes	••	••	••	••			•	•	•	•	•	0	0		•		•	
	DEFORMATION	••	••	••	••		•		•	• •	•	•	•	•		•		•	
	WIRE BREAK	••		••	••				•	•	•	0	•	•		•		•	
	LOSS OF SECTION	••		••				•	•	0	•	•	•	••	•	•	•	•	
	DETERIORATED MORTAR JOINTS				•	•			•					0		•			
	FREQUENCY					••							•	•					
ORS	VIBRATIONS/ OSCILLATIONS					•							•	•					
DICAT	OBSTRUCTION/IMPENDING	•	•						•		•	0	•	0					
EIND	DISPLACEMENT	•	•		•		•		•	•	•	0	•	•	•	•	•	•	
IANC	CRACKS	•	•	•	•			•	•		••	••	•	•	•	•	•	•	
ORN	Scaling	•		•												•			
PERF	REINFORCEMENT BAR FAILURE/BENDING	•			•					•			•	•	•	•		•	
	STIRRUP RUPTURE			•	•					•			•	•	•	•		•	
	CRUSHING	•	•							•	•	0							
	DEBONDING	0	00	•					•	•	• •	• •	•	•	•	•	•	•	
	DELAMINATION	•		•				•	•	•	•	•	•	•	•	•	•	•	
	PRESTRESSING CABLE FAILURE					•			•										
	SPALLING	0									•	•	0	•	•		•		
	TENSIONING FORCE DEFICIENCY					••			•	•			•	•	•	•		•	

Table 2.9 – Pls vs Surveying technologies (for both bridges and tunnels)





2.8.1.1 Bridges

			Cracks	CRUSHING	Rupture	DELAMINATION	Scaling	Spalling	Holes	DEBONDING	DESTRUCTION/IMPENDING	DISPLACEMENT	Deformation	WIRE BREAK	Prestressing cable	Reinforcement bar failure/bending	STIRRUP RUPTURE	Tensioning force deficiency	Loss of section	DETERIORATED MORTAR JOINTS	Frequency	VIBRATIONS/OSCILLATIONS
		ABRASION			•				0				0	•					0	0	•	•
		AGGRADATION (ALLUVIATION)									0	•	0									
		EROSION	•		•		0		•			•	0	•		•	•		•	•	•	•
SSES	PHYSICAL	CHANGING GEOTECHNICAL PROPERTIES	0	•	•				0			•	0	•	0	•	0	•			•	•
		AGING OF MATERIAL	0							0		•	0					•	0	0	•	0
		FATIGUE	0		•				0				0	•	0	•	•			•	•	•
		IMPACT DUE TO AN ACCIDENT			•						0	•										
ROCE		OVERLOADING OF AN ELEMENT	0	•	•							0	0	•	0	•	•	•		0		
ge pi		FREEZE THAW	0			0	•	•	0	0			0						•	•		
AMA		HIGH TEMPERATURE				0						•	0					•		•	•	•
Ď		ALKALI AGGREGATE REACTION	0			0						•	0			•	•	•			•	•
	11CAL	SULPHATE REACTION	0			0	0	•	0			•	0			•	•	0			•	•
	CHEW	CHEMICAL ATTACK				0	0						0	•	0	•	•		•	0		
		CORROSION	0				0							0	0	•	0		0		•	
	BIOLOGICAL	BIOLOGICAL GROWTH	0	•	0				0	0	0	0	0									
	DESIGN AND CONSTRUCTION ISSUES	DESIGN AND CONSTRUCTION ISSUES	•				•							•		•	•		•		•	

Table 2.10 – Damage process vs PIs (bridges)




	DAMAGE PROCESSES																
DAMAGE PROCESSES RELATED TO DESIGN AND CONSTRUCTION ISSUES	BILOGICAL GROWTH	PITTING CORROSION	CHEMICAL ATTACK	SULPHATE REACTION	ALKALI AGGREGATE REACTION	HIGH TEMPERATURE	FREEZE THAW	OVERLOADING OF AN ELEMENT	IMPACT DUE TO AN ACCIDENT	FATIGUE	AGING OF MATERIAL	CHANGING GEOTECHNICAL CONDITIONS	EROSION	AGGRADATION (ALLUVIATION)	ABRASION		
															0	Abrasion resistance	
															0	SURFACE HARDNESS	
						•				•					0	Compressive strength	
													•	•		CONCRETE SPALLING	
										0		•				DISPLACEMENT RATE	
0	•									•		•				VERTICAL STRESSES	
											0					VELOCITY	
											•					PATH OF DEFORMATION	
									•							INDUCED STRESSES DUE TO AN IMPACT	
									•							INDUCED DISPALCEMENTS	
									•							VIBRATIONS	
								•								AVERAGE CRACK DENSITY	
								•								Longitudinal crack density	
								•								TRANSVERSE CRACK DENSITY	
								•								CRACK OPENING	
						•	•									MODULUD OF ELASTICITY	
							•									FUNDAMENTAL TRANSVERSE FREQUENCY	DAN
							•									WATER SATURATION	AAG
						•										TENSILE STRENGTH	EIN
					•											Phase composition	
					•											GEL PRESENCE	TOP
					0											CONTENT OF ALKALI	S.
					0											Moisture	
					•											Presence of deformed Areas	
					•											INTERNAL EXPANSION	
					0											MECHANICAL PROPERTIES OF CONCRETE	
			•	•												INTERNAL HEAT DISTRIBUTION	
•		•	•	•												CHLORIDE IONS CONTENT	
			•													Waveform amplitude	
			•													PH OVER CROSS SECTION	
•		•	•													DIELECTRIC PROPERTIES	
•		•	•													DIELECTRIC CONSTANT CHANGES	
0		•														PRESENCE OF DEFECTS	
0	•															DISPLACEMENT RATIOS	
	•															Bulging	
•																DEFORMATION	
0																Horizontal stress	

Table 2.11 – Damage process vs DIs (bridges)



Page | 37



2.8.1.2 Tunnels

			PERFORMANCE INDICATORS																		
			CRACKS	CRUSHING	Rupture	DELAMINATION	Scaling	SPALLING	Mechanical instability	DEBONDING	Obstruction/Impending	DISPLACEMENT	Deformation	Wire break	Prestressing cable	Reinforcement bar failure/bending	STIRRUP RUPTURE	Tensioning force deficiency	Loss of section	CHANGES OF CONCRETE PH	INDUCED STRESSES
		CONTINUOUS VERTICAL ROCK MOVEMENT	0	0	0																
		LOCAL ROCK MOVEMENT (PUNCHING)	0	•	•							0	•								
		HIGHER HORIZONTAL ACTIONS	0	•	•											•			•		
	AL	BENDING STRESS	0	•	•																
	HYSIC	DEBONDING	0							0						•					
S	Р	PARTIAL SPALLING OF CONCRETE COVER	0					•		0						•			0		
CESSE		OVERLOADING (ROCK MOVEMENT) OF PRESTRESSING	0	•	•			•			0	0									
: PRO		DEFORMATION OF THE GROUND			0						0	0									
IAGE		WATER IMPACT			•				•											0	•
DAN	JES	MISSING REINFORCEMENT	0	•	•											•			•		
	JCTION ISSI	DEFORMATION DUE TO SHRINKAGE, TEMPERATURE WITHIN THE SHELL BLOCKS	0									0	•								
	NSTRL	DIFFERENT CASTING TIMES	0									0	•								
	ND CO	DIFFERENT CONCRETE QUALITIES	0			•						0	•								
	SIGN A	DELAMINATION OF CONCRETE LAYERS	0			0						0	•								
	DES	ANCHOR FAILURE	0	•	•						•	•									

Table 2.12 – Damage process vs PIs (tunnels)





	DAMAGE PROCESSES															
ANCHOR FAILURE	DELAMINATION OF CONCRETE LAYERS	DIFFERENT CONCRETE QUALITIES	DIFFERENT CASTING TIMES	DEFORMATION DUE TO SHRINKAGE	MISSING REINFORCEMENT	WATER IMPACT	DEFORMATION OF THE GROUND	OVERLOADING (ROCK MOVEMENT) OF PRESTRESSING	PARTIAL SPALLING OF CONCRETE COVER	DEBONDING	BENDING STRESS	HIGHER HORIZONTAL ACTIONS	LOCAL ROCK MOVEMENT (PUNCHING)	CONTINUOUS VERTICAL ROCK MOVEMENT		
				•		•	•							•	INDUCED STRESSES	
													•		VERTICAL FAULT DISLOCATION	
												•	•		VERTICAL STRESS	
												•	•		Horizontal stress	
													•		Deformation rate	
												•			Displacement ratio	
											•				STRESS DISTRIBUTION PROFILES	
		•	•							0					Strenght mesurement	
									0						Existence of depressions	
									0						ULTRASONIC PULSE VELOCITY	
								•							Elongation at maximum load	
								•							Modulus of elasticity	
								•							Shear modulus	
							•								NDUCED DEFORMATIONS	AM/
						•									Bulging	AGE
						•									Mechanical instability	INDI
						•									VHANGES OF CONCRETE PH	CAT
					•										Magnetic flux density	ORS
					•										Field fluctuations	
				•											Shrinkage rate	
		•													CHEMICAL COMPOSITION	
		•													Compression rate	
		0													Hardnbess	
	•														DELAYED FINISHING	
	•														Surface crushing	
	•														SUPERFICIAL CRUSHING	
•															Anchor weaknesses	
•															Chloride ions content	
•															Presence of defects	

Table 2.13 – Damage process vs DIs (tunnels)





3 Inspection

3.1 General

3.1.1 Definition of inspection

On-site examination within the scope of quality control and damage and/or condition assessment, aiming to assess the present condition of a structure.

3.1.2 Inspection's classification

Within the inspection in structures field, three main levels can be distinguished:

The first level corresponds the routine maintenance inspection. It usually consists in a visual inspection to provide qualitative (or semiqualitative) and rather subjective description about the structural elements and/or the structure itself.

The second level is the condition rating inspections, which is based on perform more detailed evaluations of the problematic components and the defects that were estimated in the previous level in order to quantify the effects of the damage on the structural performance

The third level entails extraordinary inspections to provide improved knowledge of the condition, structural capacity, in-service performance, or any characteristic beyond the scope of other types of inspection. In this stage, some tools like instrumentation and/or detailed modelling are usually employed. This level includes different techniques such as structural engineering tests, Asbestos Containing Material (ACM) identification and verification, underwater inspection, fracture critical/redundancy and sub-standard load rating among others.

3.1.3 Role of inspection in the evaluation of a structure health status

Bridges and tunnels are ones of the most critical structures in the transport network. Thus, their collapse usually entails critical economic consequences and human life losses. As is stated in [12] only in the EEUU, the National Bridge Inventory revealed that, currently, there are 691.060 bridges and the detailed information on the number of U.S bridges that have failed or were in a severe condition is not readily available elsewhere. In the same research it is mentioned that during the period between 1989 and 2000, a total of 503 bridge collapses were reported in the United States. In [13] is commented that the statistics on bridge failures from 2009 to 2019 in China shows that most of these failures are related to anthropic factors, and that a lack of real-time monitoring, risk assessment and other managements issues are potential factors that can cause collapses. This study also conclude that the management issues related to construction, design, maintenance and supervision are the key of these collapses being a 69.9% of the collapsed structures. As can be denoted, nowadays, with the great transport network with the current number of critical infrastructures such as tunnels the maintenance is a priority. Within the maintenance operations in structures, the most frequent task are the inspections, that are performed systematically following the requirements of the manual, guidance or standard of the responsible agency. Routine inspection processes can detect anomalies, damage or critical components and require further investigation if it is necessarv.

The different inspection manual, guidance or standards were developed in recent year fitting to current needs and leaning from the experience of previous collapses as the case of New York's Schoharie creek bridge in 1987 that provoked a greater attention in underwater inspection and the corresponding agency (FHWA) responded with "Scour at bridges", a new guidance for developing and implementing a scour evaluation program.





Nowadays with the current experience and the high develop of the technology, the different responsible maintenance agencies elaborated several standards, guidance, or manuals for each type of structure, and the most common potential problems. As it was highlighted in [14], only in the European Union, several different manual or guidance's are available for each agency or each country. In these documents the classifications damage, the forms or the damage indicators are different, thus, currently arise the need of a common procedure to follow the different agencies of each country.

3.1.4 Best practices

Several manuals or guidelines can be found to perform a suitable inspection in function of the country or the type of structure. Among others may be highlighted classified by country:

EEUU:

- Manual for Bridge element inspection, AASHTO, EEUU.
- Metrics for the Oversight of the National Bridge Inspection Program, Federal Highway administration, EEUU.
- Commonly Recognized (CoRe) Structural Elements (2001).
- Manual for Condition Evaluation of Bridges, 2nd ed. (2000).
- Movable Bridge Inspection, Evaluation, and Maintenance Manual (1998).
- Standard Specifications for Structural Supports for Highway Signs, Luminaires, and Traffic Signals, 4th ed. (2001), 272 pp.
- Bridge Inspector's Reference Manual, FHWA NHI 03-001(2002).
- Bridge Scour and Stream Instability Countermeasures Experience, Selection, and Design Guidance, 2nd ed., NHI-01-003 (2001).
- Culvert Inspection Manual, FHWA-IP-86-2 (1986).
- Highway and Rail Transit Tunnel Inspection Manual, FHWA-IF-05-002 (2005).
- Inspection of Fracture Critical Bridge Members, FHWA-IP-86-26 (1986).
- Recording and Coding Guide for the Structural Inventory and Appraisal of the Nation's Bridges, FHWA-PD-96-001 (1995).
- Underwater Inspection of Bridges, FHWA-DP-80-1 (1989).
- USDA Timber Bridges Design, Construction, Inspection, and Maintenance (1992), Forest Service.
- Tunnel Inspection Handbook, massDOT (2018).
- Tunnel Operations, Maintenance, Inspection, and Evaluation (TOMIE) Manual, FHWA-HIF-15-005 (2015).

Australia:

• Structures Inspection Manual, Department of Transport and main roads.

France:

• Road tunnel civil engineering inspection guide. Book 1: from disorder to analysis; from analysis to rating, Centre d'Étrudes des Tunnels (2015).

United Kingdom:

• Guide to surveys and inspections of buildings and associated structures, IStructE, London.





- Requirements for Inspection and Management of Bridges, BD 62/94 and BD 63/94.
- CD 352-Design of road tunnels, Standards for Highways (2020).

Spain:

- Guía para la realización de inspecciones principales de obras de paso en la red de carreteras del estado, Ministerio de fomento.
- Inspección básica de puentes de ferrocarril, Adif.
- Orden circular 27/2008 sobre metodología de inspección de túneles, Ministerio de Fomento (2008).

Denmark:

• Inspection of Bridges (1994), Danish National Road Directorate.

Finland:

- Guidelines and Policy for Bridge MR&R Operation
- Guidelines for Bridge Inspection
- Bridge Inspection Manual
- Bridge Repair Manual (SILKO–Guidelines)

Germany:

- Highway Structures Testing and Inspection, DIN 1076 (1999), Deutsche Norm.
- Preservation and Maintenance (n.d.), Construction and Housing, German Federal Department of Transportation.
- Guideline for the Structural Design and Equipment of Bridges for Monitoring, Inspection and Maintenance (1997), German Federal Department of Transportation.
- Recording and Assessment of Damages, Guideline RI-EBW-PRÜF, 1998.
- ASB Structure Inventory, (coding manual for SIB–Bauwerke) (1998).

Norway:

• Handbook for Bridge Inspections (2001), Norwegian Public Roads Administration.

Canada:

- BIM Inspection Manual, Version 3 (2005), Alberta Infrastructure and Transportation.
- BIM Inspection Manual—Level 2, Version 1 (2004), Alberta Transportation.
- Structure Inspection Manual (2000), Ontario Ministry of Transportation.

Although each country has different guidelines for defining the inspections, most of them are based on the same fundamentals. Throughout this chapter, the three previously defined inspection levels will be described and related to each of the inspections carried out in some of these countries.

3.2 Routine maintenance inspections

The routine maintenance inspection has variations depending on the country and the used guideline or standard. Nevertheless, all of them share the same main objective, checking the serviceability and functionality of the structure through a visual inspection. Table 3.1 shows some of the inspections in different countries that can be assimilated to the routine maintenance inspections.





Country	Inspections
Australia	Routine maintenance inspection.
Denmark	Daily inspection, routine inspection and reports from users.
Finland	Annual inspection.
France	Routine visit and annual inspection.
Germany	Superficial inspection and minor test.
Norway	General inspection.
Sweden	Regular inspection, superficial inspection and general inspection.
United Kingdom	Superficial inspection and general inspection.
United States	Routine inspection.

Table 3.1 – Inspection similar to routine maintenance inspections in different countries

3.2.1 Objectives

The purpose of the Level 1 inspection is to verify the serviceability of the structure. Besides, in the routine maintenance inspection the inspector must search any new sign of damage, defects and unusual behaviour and follow up on the known ones.

3.2.2 Description

All the inspections in Table 3.1 try to achieve the objective of a routine maintenance one through visual inspections. Nevertheless, depending on the country, this inspection is carried out by different inspectors, and it is even divided into several inspections. In Australia, Finlandia, Norway and the United States the routine inspection is performed in a single stage that comprehends the observations required for determining the functionality of the structure, the identification of any visible changes in the initial conditions and the follow up of the previously recorded defects. In addition, these inspections may be performed in conjunction with the routine maintenance. On the other hand, in countries such as Denmark, France, Germany, Sweden and the United Kingdom some prior inspections are also performed. These daily or superficial inspections or routine visits are carried out by the road maintenance crews and the road agents (during their patrols) or the contractor to detect new conditions of the structure and ask for routine inspections if it is necessary. Besides, in Denmark, the users have the possibility of generating reports for warning about any impact damage, vandalism, erosion, etc.

Prior to the inspection, the accredited inspector must check that the structural inventory, the safety equipment and the required permits are in place. The structural inventory is a report made during the construction of the structure that shows its original condition. In fact, this inventory may be changed to include new defects found during previous inspections. Hence, the inspector has an updated report to compare the actual condition of the structure with the original or previous inspection structural health.

During the routine maintenance inspection, components of the structure such as the deck, the footways, the signposts, the substructure, the walls, etc. have to be studied to find and follow up visible defects like cracking in the material, displacements, corrosion, delamination, depression of some component, settlements, concentration of vegetation or debris, drainage, functionality of the light and security inspection, etc. In addition, some pictures can be taken to help with their localisation and identification.

It must be noted that, during the inspection, the confined spaces of the structure, inaccessible areas or components below the water level and ground do not need to be evaluated.

3.2.3 Frequency of execution

The frequency of execution depends on the guidelines and the standards of each country so as it can be observed in Table 3.2.





Country	Inspection	Frequency of execution
Australia	Routine maintenance inspection	One per year
Denmork	Daily inspection	One per day
Denmark	Routine inspection	One per year
Finland	Annual inspection	One per year
Franco	Routine visit	Frequently
France	Annual inspection	One per year
Cormony	Superficial inspection	Four per year
Germany	Minor test	Three years after major tests
Norway	General inspection	One every year or two
	Regular inspection	Frequently
Sweden	Superficial inspection	One per year
	General inspection	One every three years
United Kingdom	Superficial inspection	Frequently
	General inspection	One every two years
United States	Routine inspection	One every year or two

Table 3.2 – Frequency of execution of the inspections similar to routine maintenance inspections

Furthermore, the countries usually established the requirement of performing a Level 1 inspection after a major natural event like floods or earthquakes to verify the serviceability of the structure. However, these inspections are usually called damage inspections in the guidelines.

3.2.4 Data collection

The information obtained in the routine maintenance inspection shall be recorded in a Level 1 Inspection Report, where the accredited inspector must write all the new defects identified during the inspection and the condition of the damage that were already recorded in the structural inventory. Usually, these reports are a checklist of defects that the inspector must complete. Furthermore, the report has enough space for adding the location and any comment about the identified defects. Finally, the inspector also has to write down its data, the data about the structure (e.g., name, location, material, type, etc.), the date of the inspection and the designation of any other inspection, if necessary. Hence, it is also a useful guide for assisting the inspectors. On the other hand, all the drawn sketches and taken photographs must be included in a Photographs and Sketches Record.

3.2.5 Inspection outcomes

The outcomes of a Level 1 inspection are the reports commented before. These reports collect the position and nature of the visible defects in accessible areas of the structure. In addition, the Level 1 Inspection Report allows inspectors to include comments about not identifiable defects and the reason for their omission. Finally, in the conclusions of the report, the inspectors have to discuss the need to monitor the structure and give recommendations about the necessity of maintenance actions, minor repairs and higher-level inspections.

3.3 Condition rating inspection

In the condition rating inspections, the current structural health of the construction and all its elements are studied. Nevertheless, similar to the Level 1 inspection, each country has different inspections to reach this goal. These inspections can be observed in Table 3.3.





Country	Inspections
Australia	Condition rating inspection.
Denmark	Principal inspection.
Finland	General inspection and basic inspection.
France	IQOA evaluation and detailed inspection.
Germany	Major test.
Norway	Major inspection.
Sweden	Major inspection.
United Kingdom	Principal inspection.
United States	Hands-on inspection and fracture-critical member inspection.

Table 3.3 – Inspections similar to condition rating inspections in different countries

3.3.1 Objectives

The purpose of the Level 2 inspection is to rate the current load carrying capacity and the condition of a structure and all its components. Besides, in the condition rating inspection, the accredited inspector must look for any defect in the structure, analyse its origin and measure its magnitude. Hence, the condition rating inspection is used for identifying and quantify the defects of a structure, determine its residual life, assess its current load carrying capacity and define the required maintenance actions.

3.3.2 Description

The condition rating inspection is an arm-length through visual examination of the structure, using some destructive and non-destructive testing techniques to obtain any necessary measurement or material sampling. Nevertheless, depending on the country, this inspection may be divided into several types.

In countries like Australia, Denmark, Germany, Norway, Sweden and the United Kingdom the evaluation of all the structure and its components is performed in a singular inspection. However, in countries such as France it is carried out in two different inspections, a detailed arms-length inspection for noting all the defects existing in the structure and an IQOA (Image de la qualité des Ouvrages d'Art) evaluation of these defects for the assessing of the structure. In the United States, in addition to the thorough visual inspection, an inspection of the most unfavourable components of the structure is also carried out. On the other hand, in Finland an additional general inspection (called basic inspection) of 125 reference bridges is performed to extract the information that is the basis for the creation and updating of bridge deterioration models. Finally, it must be noted that in countries like Germany, Norway, Sweden or the United States, the Level 2 inspections include the evaluation of structural elements below the water level.

Prior to the inspection, the accredited inspector shall ensure that all the necessary documentation, inspection equipment, safety equipment, required permits and access equipment is prepared and reviewed.

During the condition rating inspection, all the defects and damage in the structure are mainly identified by an arm-length thorough visual inspection. In addition, the accredited inspector must also determine the origin of these defects, their magnitude, their rate of variation and the effect that they have in the structure and its behaviour. Some examples of typical defects and damage identified during these inspections are scouring, cracking, impact damages, visible settlement or rotation of substructure elements, displaced bearings, corrosion, cracking in weld beads, rot, etc. Subsequently, the inspector has to grade the condition of each structural component based on this information. Furthermore, the inspector shall take several photographs, with digital timestamp, of all the structure and each one of the identified defects to help in their evaluation and localization. On the other hand, the possible modifications in the structural elements must be identified, listed and evaluated. All this information has to be used







for the inspector to evaluate the structural health of the construction. This assessment must be based on the experience and the criteria of the accredited inspector. The inspector can use any non-destructive or destructive testing techniques (e.g., timber drilling, resistograph testing, ultrasonic testing, ground penetrating radar, nuclear densitometry, etc.) and closure the necessary lanes to ease the evaluation process and to acquire all the required data of the inspection. Besides, all this information can be used for developing structural models that allow the performance of several structural analyses. Finally, if the evaluation of some specific behaviour is necessary, the inspector can require a Level 3 special inspection.

3.3.3 Frequency of execution

The frequency of execution depends on the guidelines and the standards of each country so as it can be observed in Table 3.4.

Country	Inspection	Frequency of execution		
Australia	Condition rating inspection	Between one and five years		
Denmark	Principal inspection	One every six years minimum		
Finland	General inspection	Between five and eight years		
Finianu	Basic inspection	One every five years		
Franco	IQOA evaluation	One every three years		
FIGILE	Detailed inspection	Between one and nine years		
Germany	Major test	One every six years		
Norway	Major inspection	Between five and ten years		
Sweden	Major inspection	One every six years		
United Kingdom	Principal inspection	One every six years		
	Hands on inspection	One every three years		
United States		minimum		
	Fracture-critical member inspection	One every two years		

Table 3.4 – Frequency of execution of the inspections similar to condition rating inspections

It must be noted that the frequency of execution of some inspections in Australia, France and Norway also depend on the typology and the condition of the structure.

3.3.4 Data collection

The information extracted from the condition rating maintenance shall be recorded in several reports. The Level 2 Inspection Report saves the inspection component inventory, the assessment of each of these components, their exposure, the origin and effect of all the identified defects and the overall condition of the construction. Besides, the taken photographs must be also introduced in the Photographs and Sketches Record. On the other hand, the accredited inspector must also generate a Defective Components Report to write down all those components that require monitoring or Level 3 inspections, and the Standard Procedure Exceptions Report, which shows the components that cannot be evaluated and the reason for their omission in the inspection. Finally, special reports have to be created if additional testing surveys were carried out.

3.3.5 Inspection outcomes.

The outcomes of a Level 2 inspection are the reports listed above. Hence, the condition rating inspection main outcome is the condition grade of all the components that belongs to the studied structure. This grade takes into account the structural health of the element, the defects presented on it, as well as their magnitude and origin, and the risks of the atmosphere where the component is located. Besides, the inspection reveals the general condition of the construction, recommending the required maintenance and restoration works, as well as their costs, for keeping its correct behaviour. Finally, in the reports, the accredited inspector also discusses the need to monitor the structure or some of its elements and to request a special inspection.





3.4 Extraordinary inspections

Level 3 inspections are intended to provide improved knowledge of the condition, load carrying capacity, in-service performance and other characteristics that are beyond the scope of Level 1 and Level 2 inspections. These inspections are not carried out systematically. These arise as consequence of detected damage during inspections at previous levels. In this type of inspections, a visual inspection is carried out together with different test or complementary measurements. This level requires a previous inspection plan in which are detailed the elements object of study and the employed tools or test to be carried out, this inspection plan is usually elaborated for an accredited inspector. The level of accreditation depends on the manual, standard or guidance of the responsible agency. This inspector will decide in function of the structure and the results of the previous level inspections which tests will be performed, specifying the execution procedure if there are no any corresponding standards.

3.4.1 Objectives

The main objectives changes in function of the type of special inspection that will be performed. Briefly the general main objective is to know in greater depth the current state of the structure or of some elements reported in the previous inspection levels. The conditions of the inspection are established in the project-specific brief.

In the case of the detailed structural engineering inspection the particular objectives are identify and quantify the current deterioration process, determine the structural condition in terms of safety, reliability and the behaviour of a structure through physical testing and structural analysis, and finally to develop appropriate management and repair strategies.

In the Asbestos Containing Material (ACM) identification and verification inspection the first purpose is to identify the potential permanent/sacrificial inclusion of asbestos on structures in order to ensure that the department's asbestos register is up to date with the Work Health and Safety counterpart. The Program Manager will make arrangements for a visual inspection. If an ACM is identified the following objective is to confirm the presence of asbestos in suspected ACM where the material may be disturbed through any proposed activity on the structure. The inspection involves hands-on practices to gain access to the areas of concern and may involve the breaking back of limited areas of concrete to facilitate removal of samples for testing by a NATA-accredited laboratory under ISO 17020.

In underwater inspection the objective is to determine the current state of the underwater components and to provide a benchmark for future inspections. In case of the fracture critical/lack of redundancy inspection the objective is to inspect and to identify more deeply the fracture of critical members in the bridge with no load path redundancy. To correctly develop this inspection, diving equipment and specially trained and accredited personnel is required.

Finally, in other extraordinary inspections, the objectives are detailed in the defined inspection plan.

3.4.2 Description

Each extraordinary inspection presents its own procedure that is defined in the corresponding standards, manual or guidance of the agency responsible for the maintenance of the structure.

The detailed structural engineering inspection entails a visual examination of all accessible components of the structure. Where is necessary, it could be complemented by examinations, testing or analysis specified in the project-specific brief such as geotechnical investigations, static or dynamic load testing of the structure or non-destructive measuring technologies. In addition, a load capacity assessment may be included in the brief to determine the repeated live load capacity for the remaining service life of the structure. This inspection comprises different task such as the review of any previous inspection and testing reports, traffic counts,







studies, environment factors, carry out measurements and testing campaigns to supplement the visual inspection, determine the material properties and structural behaviour. Besides, it usually includes other tasks like identifying components which are limiting the performance of the structure or are likely to deteriorate, probable causes and projected rate of deterioration, durability, residual life of the structure, influential factors in the dynamic load allowance, and hydraulic performance among others.

Focusing on ACM inspection, this is a visual inspection to confirm the presence of potential ACM. If ACM is identified through a visual inspection the asbestos register has to be updated. Once the ACM is identified starts the process of ACM verification. If some of the components present ACM, the register must be updated indicating the elements and the ACM condition. In addition, it must warn Workplace Health and Safety and advise the responsible of the inspection to implement appropriate control measures.

An underwater inspection usually includes some tasks like the mapping of local scour, metal corrosion, reinforced concrete cracking and spalling, prestressed concrete splitting, pile loss and residual section, debris mapping, extensive photographic record or extraction of underwater samples among others. In addition, tidal and splash zone areas should also be inspected, and scour soundings undertaken.

Other extraordinary inspections (e.g., fracture critical member identification, confined space inspection, etc.) are carried out following the inspection plan requirements which are individually elaborated depending on the agency responsible for the maintenance of the structure and its nature and typology.

3.4.3 Frequency of execution

Unlike Level 1 and Level 2 inspections, performed at predetermined frequencies, a Level 3 inspection is carried out on an as-needed basis. A detailed engineering inspection will be carried out in some particular circumstances such as assessing the condition of a structure prior to carrying out programmed works like rehabilitation, strengthening or widening, as the result of recommendations in a Level 2 inspection, to provide a load rating for the structure among others.

In the particular case of the ACM inspection, it will be undertaken once only, at the earliest available opportunity, as part of the Level 1 or Level 2 inspection program. Concerning other extraordinary test, the frequency of execution is in function of the corresponding standard or manual of the responsible of the structure or as a result of the systematically inspections.

3.4.4 Data collection

Data recording requirements will be in accordance with those specified by the project-specific brief. The data recording will be, at least, similar to those required for a Level 2 inspection. Therefore, through visual inspections and measuring technologies, the inspector must extract information about the location, magnitude and effect on the structure of all the defects that must be evaluated during the specific inspection.

In the case of the underwater inspection, the responsible for the maintenance of the structure drafts a specific brief to define the scope of the work. The brief should include the elements that will be inspected and how the inspection will be performed if there are some specified requirements, specified in some guideline, standard or inspection pro forma sketches, the brief must specify the level of the reporting required. The accredited inspector shall review the data collected by the diver and ensure that the required level of detail has been recorded for reporting purposes.

3.4.5 Inspection outcomes.

A draft report is usually elaborated including the compilation of standard inspection forms and supplemented by a written report that is required by the plan inspection brief. All the





information from the tests carried out shall be included in these reports. Furthermore, the accredited inspector must write down the conclusions about the inspection and the request of any maintenance action, monitoring, etc. if necessary.

In the case of the detailed structural engineering inspection, a written report that includes the specific requirements outlined in the project brief shall be submitted. These requirements should include the diving surveys and materials testing, the rating of all primary defects, the identification of the deterioration mechanism and determination of the overall condition of the structure, the results of any load capacity assessment and, if it is required in the brief, a bridge equivalence rating. If the condition rating of the components or the overall rating condition of the structure differs from the Level 2 report, then a new Level 2 inspection report shall be prepared revising condition ratings of the inspected components.

Concerning the outcomes of the ACM inspection, three main tasks will be carried out: update the register of ACM, advise Workplace Health and Safety that a check has been conducted and potential asbestos containing material has been identified, and place signs on the structure warning of the risk in it. If the presence of ACM is confirmed, the register will be updated indicating the elements and the ACM condition. Besides, Workplace Health and Safety and the responsible of the inspection must be advised of the findings and the need to implement appropriate control measures. In other tests, the outcomes shall be the reports of the actions determined in the results, following the requirements of the certified inspector, the manual guidance or the corresponding standard if it is available.





4 Monitoring

4.1 General

In recent decades there has been an increasing diffusion of structural monitoring systems in combination with the most widespread and common structural management procedures: structural monitoring is the result of a set of technologies aimed at determining the state of a structural system and its evolution over time, to detect and quantify phenomena of degradation or damage and to allow the evaluation of the integrity of the system itself and its ability to remain in operational condition with adequate levels of safety for a given period of time. [8] identifies the following possible reasons for the implementation of a monitoring system, summarized in Figure 4.1.

MONITORING FOR STRUCTURE MANAGEMENT										
UNDER NORMAL CONDITIONS	UNDER SPECIAL CIRCUMSTANCES	DAMAGED OR DETERIORATING STRUCTURES								
 Structures susceptible to long-term movements Durability monitoring Fatigue assessment Comparison with design 	 Modification or demolition of existing structures Structures affected by external works Special investigations or assessments 	 Structures subject to long term movements Structures affected by degradation of materials Control tool for extending service life 								
RESEARCH MONITORING										
 Evaluation of novel forms of construction/materials Cost/benefits to be derived from monitoring Feedback into the design process/standards Improved understanding of in-service behaviour of structures 										

Figure 4.1 – Examples for reasons for the use of monitoring systems (Adjusted from [8])

Hence, condition monitoring activities, combined with inspections and testing, are key tools for the through-life management of the structure: actions to be taken concerning condition control, conservation and maintenance works should be planned based on the information gathered through the activities above, which, according to [15], must be carried out from an early stage in the service life of the structure. Planned activities are included in the Conservation Plan, which states the types of inspection, testing and condition monitoring that have to take place, what components of the structure are to be inspected/monitored, what the frequency of the inspections should be, etc.

In this respect, the IM-SAFE project proposes in [10] a framework for the data-informed safety assessment highly interrelated with the through-life management of the transport infrastructure which describes the different stages of the assessment of new and existing structures, focusing on various levels (network, system, and component) and taking into account the available data collected from inspection and monitoring activities during the lifespan and maintenance process of the structure. Inspections methods are described in chapter 3, while chapter 4 focuses on monitoring.

The use of monitoring systems has the following objectives:







- a better correlation between loads/actions acting on the structure, the consequent state of deformation and the technical predictions assumed as a basis for the design, thus a more reliable knowledge of the behaviour of the structure;
- the identification of more precise modelling, more efficient dimensioning criteria, and an improved safety assessment;
- an early detection of any anomalies in the structural response and thus the basis for possible reinforcement interventions and/or limitations of use, especially caused by the decay of structural resources due to behaviour under cyclical actions repeated over time or to occasional actions, such as those due to earthquakes or environmental and anthropic agents;
- the definition of strategies aimed at extending the expected life of the structure;
- an improved asset management;
- a collection of statistical data that could have an impact on normative provisions, also with regard to the effects of climate variations.

Hence, structural monitoring provides the opportunity to observe the behaviour of the structure over time and the load history to which it is subjected: the need for data acquisition about its static and dynamic behaviour is increasingly pressing for existing structures, especially for bridges and tunnels, which are critical components of the road infrastructure.

This information is obtained using a network of sensors suitably placed on the structure, data acquisition systems, units for storing and analysing measurements, systems for transmitting data to processing units, including remote ones, and software for the analysis and the interpretation of the data.

However, while structural monitoring techniques are relatively widespread and well established in the aeronautical and aerospace fields, the same cannot be stated for the civil one. There are many reasons for this, for instance, according to [16]:

- Each structure differs in terms of its properties and challenges: this means that the development and implementation of automated monitoring processes is less easy and therefore the intervention of human knowledge is always required even at intermediate process levels. This implies in most cases a specific know-how.
- Lack of knowledge about cost/benefit analysis for the use of a SHM system.
- The technologies involved in instrumental monitoring are profoundly interdisciplinary and therefore their diffusion depends on the sensitivity of the technical-scientific context to knowledge which differs from the cultural matrix of origin, and on the attitude to adopt the innovative contents.
- The service life of civil structures is much longer than the industrial systems, which means that degradation phenomena occur much more quickly, so it takes longer to demonstrate the effectiveness of monitoring systems.

Nevertheless, the need to monitor the behaviour of modern large-scale and technologically very complex structures and the awareness of the limits of traditional approaches to the management of structures, which in several countries were manifested in serious accidents, have motivated the technical and scientific community to research and develop SHM systems suitable for the needs of the civil infrastructure field.

The most typical applications are:

- Structural monitoring
- Seismic monitoring
- Geotechnical monitoring





- Hydraulic monitoring
- Monitoring of natural and artificial slopes

The principles described in the following paragraphs are mainly based on the provisions given in [16].

4.2 Objectives and requirements

Structural monitoring systems aim to implement an in-service control process, which consists of identifying the value assumed by specific parameters that characterise the behaviour of the structure and determining the changes in the values of these parameters that occur due to deterioration processes. According to [16], the following fundamental functions of any monitoring scheme, simple or complex, are to be performed:

- A data collection and management system: information obtained by structural diagnostics (inspections, samples, load tests, etc.) and by a network of sensors directly installed on the structure is processed in the system.
- A set of data processing procedures: assessment of the condition of the structure and its developmental trend.
- **Decision-making procedures**: guidance on the choice of subsequent actions based on the indications provided by monitoring.
- **Numerical model of the structure**: this model is validated through an initial calibration, which is developed with a level of detail consistent with the complexity and relevance of the structure.

This process requires a combination of very different but closely interacting activities, both of an experimental nature (structural diagnostics and data collection with a range of sensors) and of an engineering nature (numerical modelling and system identification procedures).

The general objectives of a monitoring system can be summarised as follows:

- Monitoring the condition of the structure with respect to potential limit states: the system is designed to control the value of some relevant parameters appropriately chosen in order to identify in a reliable way the approach or the exceedance of a limit state of relevant damage or, in the most critical cases, of possible collapse.
- **Identification of an occurring degradation process**: the system may detect the occurrence of abnormal behaviour of some characteristic responses of the structure to external actions, not in line with predictions or past measurements indicated either by an anomalous trend in measured data, , or by the exceedance of a critical absolute value.

The choice of the monitoring system and its degree of complexity is the result of a case-bycase analysis and aimed at obtaining the best use of the economic resources available and the professional skills involved. The main factors influencing this choice are related to the structure, the environment/loading conditions, and the monitoring system features/potentialities:

- **The relevance of the structure**: to be assessed in terms of economic value and function carried out.
- **Environmental conditions**: structural monitoring should be given higher priority in the case of a very aggressive environment in which the structure is located, in consideration of a higher possibility of damage.
- The complexity and degree of innovation of the structure: structures using new materials or innovative construction techniques have a particular interest in the use of monitoring systems.





- **The degree of reliability of monitoring and cost-benefit analysis**: structural monitoring is a natural development of the control process.

Monitoring, in any case, cannot meet all the performance requirements of an efficient control and must be supported in its conclusions by dedicated investigations: hence the need to consider the actual benefits of monitoring against the costs incurred, which obviously increase with the size and complexity of the system.

Ultimately, the decision on the commissioning of the monitoring system and its characteristics is based on an in-depth cost-benefit analysis, on the basis of which it is possible to identify, in realistic terms, the role to be assigned to the monitoring system and to design the sensor network and, more generally, the whole system in order to be proportionate to the expected benefits

Monitoring data should be properly analysed, reported, and evaluated in a series of periodic reports and in a concluding report containing condition assessments and recommendations for follow-up actions. Source data should also be stored off-line for later use. They may be considered suitable for immediate acquisition, analysis, and evaluation systems, capable of providing early warnings about reaching predefined alarm values.

4.3 Monitoring classification

The classification of monitoring system is provided in the Figure 4.2.



Figure 4.2 – Monitoring classification

Based on the period of execution, the monitoring approaches can be divided into the following categories:

- **Short-term monitoring**: the installation of a monitoring system for a limited period and, if necessary, the repetition of the installation at more or less regular intervals over time. According to [17], this approach is recommended in the following cases:
 - o *Extraordinary maintenance or upgrading* the installation of instrumental systems before, during and after the intervention to evaluate its effectiveness is recommended.
 - o Study of the behaviour of repetitive structural types.
 - o *Boundary transitory situations*, e.g., analysis of slope behaviour for preventive stabilisation interventions.
 - o Analysis of anomalous degradation/damage phenomena, for which it is required to investigate causes and evolutionary nature.

The installation of systems of this type is particularly suitable in cases where the aim is to study a known phenomenon observed during inspections. This approach includes occasional or periodic dynamic monitoring.

• **Long-term monitoring**: the implementation of a permanent monitoring strategy, in which the hardware/software system is designed to be operational for long periods up







to the service life of a structure. According to [17], this approach is recommended in the following cases for bridges:

- o Cable-stayed or suspended bridges and large span bridges (> 200 m).
- o Bridges with spans of more than 50 m in pre-stressed concrete built more than 40 years ago.
- o Bridges with inspection difficulties (box girders and non-inspectionable piers) in prestressed concrete or steel.
- o Bridges with innovative structural solutions.
- o Bridges of historical relevance.
- o *Bridges in critical environments*, characterised by high traffic loads (e.g., frequent transit of exceptional transports), with fatigue problems, in areas with high seismic risk or with critical boundary situations, such as the high risk of flooding and landslides or bridges where accidental phenomena, such as impacts, are likely to be of great importance.

Based on the purpose, the monitoring approaches can be divided into the following categories:

- Actions monitoring, which allows the evaluation of the actual loads acting on the structures.
- **Performance monitoring**, which allows an assessment of whether a structural system or component meets the performance requirements under a known or any load:
 - o System performance monitoring
 - o Component performance monitoring
- **Health monitoring**: provides real time information for the assessment of the safety and serviceability of a structure or structural components. It is founded on the condition that a sufficient number of measurable health indicators exist and can supply relevant information on the state of the structure.

4.4 Monitoring system architecture

The system architecture represents the topological structure by which the different components of a monitoring system are connected.

A monitoring system is characterised by sensing components (sensors), a system architecture connecting the different sensors, a data collection unit (gateway) and a data repository (cloud).

In details, the elements that compose the acquisition pipeline connecting sensor and storage system typically includes the following components:

- Sensor, in which the transduction from a physical quantity of interest (e.g., acceleration, displacement) into an analogue signal, typically electrical or optical, occurs.
- Analogue communication system, which connects the sensor to the analogue/digital converter.
- **Analogue/Digital Converter**, which is the device that transforms the analogue signal into a digital signal.
- **Digital communication system**, which connects the converter to the gateway.
- **Gateway**, device to which all digital signals of the monitoring network converge.
- **Cloud**, which represents the repository and the post-processing unit.

Based on the topological distribution of these components, it is possible to distinguish between:





- centralised system, in which the analogue/digital conversion takes place in a single device to which all the sensors are connected.
- digital sensor network, in which the analogue/digital conversion takes place at each sensor and typically sensor and converter are integrated in the same device (called digital sensor). Based on the communication mode of digital systems, it is possible to further distinguish between:
 - o cabled systems.
 - o wireless systems.
- mixed system, in which the conversion takes place at several devices distributed over the structure, each of which carries multiple analogue sensors.

The architecture of the monitoring system must be properly designed: the choice of system architecture and digital communication mode should consider the following requirements.

- Reliability
- System robustness
- Easiness of assembly
- Transmission speed
- Transmission distance
- Scalability and expandability
- Encumbrance and visual impact
- Availability of power supply

4.5 Preliminary monitoring system design

The flowchart in Figure 4.3 represents the sequence of activities that are carried out in the preliminary monitoring system design phase, which has the task of providing the detailed specifications for the development of the final design.



Figure 4.3 – Diagram of the monitoring system design process (Adjusted from [16])

4.5.1 Definition of general objectives

The client and the system designer should jointly analyse the existing technical and economic issues in order to define the desired objectives, which are often the result of a compromise between the desired performance and the financial resources available.





However, it is common that the request for the implementation of a monitoring system is made without a clear definition of the objectives to be achieved by the client.

They are usually expressed in generic terms such as:

- Assessment of the condition of the structure in order to intervene in case of significant damage.
- Detection of any deterioration processes occurring.

4.5.2 Analysis of the structure and definition of specific objectives for the system

A preliminary analysis should be carried out, the objective of which is to achieve a detailed knowledge of the structure and the evolution of its behaviour. This is hardly ever achievable, except by using systems that are so sophisticated and complex as to be economically unfeasible. The most logical way to proceed is to identify a reduced number of significant quantities that provide the essential information about the structure.

The preliminary analysis allows the identification of quantities related to the type and structural organization, the materials used, the presence of essential components for a satisfactory performance and, therefore, critical components: the specific mechanisms of collapse or damage of the structure are thus identified, depending on the expected loads and environmental conditions, and the quantities which control these mechanisms are identified. The information needed for the preliminary analysis is to be found in the available data, such as those provided by structural diagnostics (all type of inspections, samples, load tests, interventions, etc.)

These quantities can be:

- **State variables**, such as characteristics of the structural response or characteristics of the actions on the structure.
- System parameters.
- Quantities defining structural behaviour, such as transfer functions or modal parameters.

There are only a few cases where these quantities are directly measurable (state variables). They are generally obtained by re-processing one or more quantities measured by the system.

The quantities thus defined therefore become the specific objectives of the monitoring and provide the input data for the interpretative models which are then used, together with other data that may come from experimental diagnostics or other information, to meet the client's requirements for damage recognition, the estimation or updating of an index of structural safety or an index of the residual life of a structure.

It is important, hence, that the choice of these quantities is made with extreme care, in order to avoid including in the monitoring system the detection of quantities that are irrelevant in the interpretation models or not to include quantities that could be significant.

4.5.3 Measurement strategy and definition of the sensor network

Once the quantities to be monitored and the required accuracy have been identified, the system designer identifies the most appropriate measurement strategy, developing a detailed layout of the sensor network defining the type of quantity measured, the number, and position of each sensor.

4.5.3.1 Types of quantities

The most common quantities of interest in monitoring are:

• Mechanical quantities characteristic for:





- o Structural behaviour
 - Cinematic quantities
 - Dynamic quantities
 - Deformations
- o Actions
 - Forces
 - Impulses
 - Environmental actions
- Thermodynamic quantities:
 - o Temperature
 - o Irradiation
- Chemical quantities:
 - o Ph
 - o Humidity
 - o Agent concentration
- Electromagnetic quantities:
 - o Electrical and magnetic potentials

It should be preferred to monitor physical quantities that are directly related to the behavioural parameters, in order to limit uncertainties relating to the interpretation model.

4.5.3.2 Number and position of measurement points

In general, the number and the position of the measurement points depend on the structural type, the monitoring objectives, the required accuracy, and the type of instrumentation expected to be used.

In case non-point instrumentation is used, the measurement can be referred to any point in the field of observation. In all other cases (instrumentation applied to the surface of the structure or embedded in it), the number and position of the sensors should be such as to provide all the necessary information for the use of the chosen interpretative model.

In some cases, instrumental monitoring requires the measurement or identification of a field, e.g., the deformation or modal shape of a structure: in this case the choice of the number and position of the measurement points is always a compromise between the need to observe the field exhaustively, limiting the uncertainties due to the incompleteness of the observed field, and the cost of the sensors.

Concerning both damage and structural response, the following methods can be used. [18], in this respect, identifies the selection of an appropriate sensor arrangement as an application-specific task, where the procedure for sensor placement critically depends on the method that is pursued:

- **Local methods** -are used for the evaluation of damage and structural response on a local scale: they are based on the principles of placing sensors where damage is expected, assuming that it directly affects the measured response quantity, so the sensor's proximity to the anticipated damage locations is pivotal.
- **Global methods** are used for the evaluation of structural response on a global scale: they assume that damage alters the mass, stiffness, or energy-dissipating properties of the structure, which in turn alter the measured dynamic response of the global system
- If numerical models are available, the sensor placement can be optimized based on mathematical **performance criteria**.





Concerning actions, sensors layout must be properly chosen based on the type of action to be monitored and on the instrumentation to be used. See chapter 8 for further information.

Due to the nature of the sensor placement problem, the number and position of the sensors can be determined using dedicated optimization algorithms.

4.5.3.3 Measurement system specification

The measurement of a quantity involves the use of a measurement chain, ideally consisting of the actual sensor, the electronics conditioning the signal output from the sensor, the connection cable to the analogue/digital conversion system.

It is recommended that the preliminary design should be limited to clearly defining the metrological specifications required based on accuracy considerations, while leaving the developer of the executive project the task of choosing the most convenient sensor and hardware solution in terms of network logistical configuration, cost, and reliability, while still respecting the metrological specifications required.

The metrological specifications include, where applicable, at least the following instrument properties:

- Precision
- Repeatability
- Stability
- Sensitivity to environmental conditions
- Measuring range
- Linearity
- Signal/noise ratio
- Transfer function
- Sensitivity

The specifications may also include the following features:

- Calibration and adjustment
- Durability or service life
- Maintainability and replaceability
- Level of protection
- Size and weight
- Electrical consumption
- Type of power supply

4.5.4 Data acquisition and processing

Data acquisition and processing are discussed in 4.9 and 4.10, respectively.

4.5.5 Interpretative models and calibration of reference model

The identification of the condition of the structure and the presence of damage is in principle based on the minimization of an error function constructed from real results which photograph the real state of the structure at a certain moment in its life and the results of a numerical model which is built (calibrated) in such a way to be able to describe the real behaviour of the structure with an assigned accuracy.

The same goals can also be reached through a data-driven approach, determining state variables, from which behaviour parameters are extracted, and/or detecting anomalies on the basis of analyses of measurements, which are pre-treated and generally turned into a time series of observations.





Models are characterised by the following properties:

- period of time needed to establish a stable reference model;
- minimum length of observation time for damage of a given magnitude to produce anomalies in model parameters;
- minimum detectable damage severity;
- reliability of the information provided.

The construction and calibration of the model can be a very complex and costly operation, especially in the case of existing structures, for which the information on the materials, the history of the loads acting on them, the history of the interventions and modifications to which they have been subjected are often not known or only partly known, and only the use of indepth experimental surveys to reconstruct the structural organization, the materials used and their characteristics and the possible state of deterioration allow a reliable numerical model to be developed.

For further information about model calibration methods, see [4].

4.5.6 Decision-making procedures

The monitoring system may be required to manage thresholds, generally established on the basis of reference limit states, to which the implementation of countermeasures is associated: targeted inspections and surveys in minor cases, execution of provisional measures and decommissioning of the structure in more severe cases.

For further information on thresholds, see [10].

4.5.7 Accuracy requirements

The effectiveness of structural monitoring depends on the reliability of the identification of the condition and behaviour of a structure. Hence, the level of accuracy required to identify system parameters and state variables should be specified at the design stage. The way in which the required accuracy is specified depends on the type of variable (Section 4.5.2) and the form in which the uncertainty is described (deterministic or probabilistic).

4.5.7.1 Definition of required accuracy

The specification of the required accuracy depends on the objective and on the variables to be obtained.

If monitoring leads to a classification of the structural state the case of a classification the uncertainty is the probability of misclassification, defined as the relative frequency, normally expressed as a percentage, with which the system identifies an incorrect state.

The maximum acceptable uncertainty of an individual parameter represented by its numerical value can be described by the maximum error of estimate, or tolerance, of the parameter, defined as the maximum deviation between the estimate of the parameter obtained through monitoring and its exact value. Different tolerance values for positive or negative deviations may be specified where necessary. When the acceptable error depends on the amplitude of the parameter, it may be appropriate to describe the tolerance in terms of a percentage error.

Alternatively, it is possible to describe the tolerated uncertainty probabilistically through the acceptable values of variance or standard deviation of the parameter. When the standard deviation depends on the amplitude of the parameter, it may be appropriate to describe the tolerance in terms of a coefficient of variation.

The acceptable error of a function can be described through the confidence band. Also, in this case different tolerance values for positive or negative deviations can be specified.





4.5.7.2 Sources of error and evaluation of uncertainty

There are sources of errors of a variety of nature and importance throughout the process leading up to measurement and damage assessment.

In principle the following types of errors can be identified:

- Errors related to the measurement phase and the actual acquisition that determine the **metrological uncertainty** of the data.
- Errors related to data processing and interpretation for damage assessment and numerical updating leading to **model uncertainty**.

The assessment of <u>metrological uncertainty</u> considers all error components that may occur in the measurement chain, including:

- The transducer's instrumental error.
- The noise of the cabling system.
- The noise of the signal conditioning system.
- The analogue/digital conversion error.

Metrological uncertainties related to each source of error can be assessed according to existing regulations or, in lack of them, by statistical analysis of observations and/or a priori estimates based on probabilistic models. It is in anyway necessary to guarantee the correct functioning of the measuring chain through a periodic calibration to be carried out based on the design specifications of the system and/or in the case of exceptional events and extraordinary maintenance.

The following forms of error are considered in the assessment of model uncertainty:

- Estimation errors due to incompleteness of the field.
- Errors due to the uncertainty of the quantities assumed to be deterministic in the model.
- Numerical errors.
- A priori uncertainty in the auxiliary parameters of the model.
- Uncertainties due to model approximation or truncation.
- Other epistemic uncertainties.

The single uncertainties can be expressed in deterministic form through the maximum expected error, which can be expressed in absolute value or in percentage terms. Alternatively, it is possible to describe the tolerated uncertainty probabilistically through the expected values of variance or standard deviation of the parameter.

It should be noted that while the metrological uncertainty is easy to assess a priori once the various components of the system have been identified and the technology has evolved, it can be maintained at the level required for monitoring through the appropriate choice of hardware and the adoption of suitable measures in the installation of the system. The assessment of model uncertainty, on the other hand, still represents a very difficult problem to solve.

4.6 Final design

Once the monitoring strategy has been developed through the preliminary system design, the designer provides a final design of the monitoring system, specifying the system architecture (Section 4.4), hardware (sensors, acquisition units and communication devices), and software (application that allows the acquisition of measurements according to project specifications and their storage in digital format) components.

*	*	*
*		*
÷.		
×	*	×



The executive design is always accompanied by an installation manual and the specification of the management protocol:

- The installation manual should be complete with explanatory drawings and diagrams, giving instructions for both hardware and software components. The manual also contains methods for on-site quality control of the installation and functioning of the system.
- The management protocol must be appropriately described in an operating and maintenance manual for the monitoring system.

In conclusion, the design of a monitoring system is represented by a set of documents and graphic tables:

- A report containing the objectives of the monitoring, also including a description of the monitored structure with all the information of interest for the monitoring.
- A document containing the formal description of the monitoring system.
- Graphic reports showing the position of the system components, any interaction of these components with existing plants and the position of any signs of structural deterioration.
- Detailed graphic reports providing details for the execution of the system installation.
- A document containing the specifications of the system's software components.
- The installation manual.
- The user and maintenance manual.

4.7 Installation and management

4.7.1 Installation

The installation of the system, for all its components, should be carried out and supervised by competent personnel. The installation of the sensors, to be carried out based on proven valid standards and/or detailed instructions provided by the executive designer and/or producer, is supported by pre- and post-installation acceptance checks, for which appropriate documentation should be provided by the technical supervisor. The sensors must be adequately protected against mechanical damage and the aggression of environmental phenomena. The accessibility of the sensors installed on the structure must be ensured as far as possible.

In the case of sensors pre-installed in materials or components of the structure, their correct functioning should be verified and documented before connection to interrogation lines is made.

The sensors subject to calibration are fine-tuned at the end of the installation according to specifications and the results of the procedure are recorded accordingly. In particular, when possible, an on-site calibration of the entire measurement chain is performed.

The installer should provide the technical supervisor and/or system operator with all documentation relating to the installed sensors, including manufacturer's certificates, verification during installation and certification of calibrations and adjustments performed.

The cabling and other data transmission systems should be properly fixed to the structure and not be subject to vibration, shock, or damage of any kind. The continuity of the lines and the quality of the transmitted signals should be verified and documented for each individual section as well as for all transmission lines. Particular care must be taken with accidental discharge protection subsystems.





4.7.2 Testing and commissioning

After the installation has been completed, functional tests are carried out on all system components. The technical supervisor should provide the operator with appropriate documentation of all operations carried out in the presence of the parties involved.

4.7.3 Management

The system operator is responsible for ensuring that the system works properly by supervising the data acquisition, storage, and analysis operations if they are carried out automatically, or for supervising the execution of measurements if they are carried out manually or with assistance.

The operator shall also promptly inform the client of the occurrence of any early warning and alarm signals of malfunctioning of the system or parts of it. The management of the system generally includes the provision of periodic reports on the measurements and functioning of the system. The preparation of data analysis and interpretation reports and the final assessment of the condition of the structure may be a task assigned to the operator, or may be assigned to other specialists, typically the designer or structural consultant, whether or not they are part of the client's organisation.

4.8 Monitoring system maintenance

Routine maintenance of the system includes periodic calibration and adjustment operations, to be programmed according to sensor specifications, replacement of damaged sensors, maintenance of power supply subsystems, cleaning of electrical and optical contacts, implementation of software updates. In case of sensor replacement, zero readings must be updated and, if necessary, also the reference numerical models.

To simplify **routine maintenance** operations, the sensors should be placed with some redundancy, at least at the most critical measuring points, and the durability guaranteed by the supplier of the most critical sensors in terms of significance of readings and difficulty of replacement should be adequate.

Extraordinary maintenance is required when, due to unexpected events or exceeding of the service life, significant parts of the system have to be replaced. Such situations generally concern electronic components of the system which have become obsolete or damaged for various reasons and can no longer be repaired.

4.9 Monitoring data acquisition and processing

4.9.1 Data acquisition

The design of the acquisition system should be developed based on the number and location of the measurement points as well as the size of the structure to be monitored, the organization of the system operator and economic considerations regarding the costs of installation and management.

The preliminary design sets out the system performance requirements and provides the system architecture. System performance requirements include:

- Number of channels (or inputs) and types of sensors that can be connected.
- Maximum sampling frequency.
- Possibility of continuous, programmed, manually controlled, automatic acquisition on trigger or pre-trigger.
- Presence of signal pre-analysis electronics.
- Database dimensions.





- Size.
- Power supply system.
- Energy consumption.

For each type of sensor, acquisition methods should be defined based on the nature of the quantity measured and its significance in the content of subsequent processing:

- Sampling frequency.
- Periodicity of acquisition.
- Signal synchronization requirements.

4.9.1.1 Sampling frequency

The minimum sampling frequency is chosen according to the type of physical phenomenon observed and its variability over time, which can be represented by its maximum frequency. In particular, the Shannon-Nyquist theorem specifies that the sampling frequency cannot be less than half the maximum frequency of the phenomenon:

The down-sampling of the analogue signal determines, due to the aliasing phenomenon, an alteration of the frequency content. When the maximum frequency is much higher than the frequency of interest, the sampling frequency can be reduced, provided that low-pass filters are introduced before the conversion from analogue to digital.

Although in general the quality of the digital signal increases with the sampling frequency, the maximum sampling frequency is nevertheless limited in order to consider the limitations of the data transmission and storage system and in particular:

- Memory capacity of the converter buffer.
- Available transmission bandwidth in the local and remote network.
- System storage capacity.
- Data processing time.
- Speed of access to the database.
- Reporting time to data display.

4.9.1.2 Periodicity of acquisition

In the case of periodic acquisition, the acquisition rate, and the length of each record (or the number of samples) should be specified. It is also necessary to specify whether these parameters can be modified by the user during system management. The length of the record is chosen considering the required accuracy of the post-processed data too. For instance, if the acquired signal is in the frequency domain, the frequency resolution of the transformed signal depends on the total length of the record:

In the case of event-driven acquisition (triggering), the algorithm enabling event recognition should also be specified. The simplest trigger condition is the exceeding of a threshold, which may be related, for example, to:

- The maximum, average or RMS (Root Mean Square) amplitude of the signal.
- The variation of the measurement from the previously acquired sample.
- The amplitude (maximum, average or RMS) of one of the spectral properties of the acquired signal.





It should also be specified whether the trigger condition applies to the individual sensor or to a combination of sensors. More refined trigger conditions can be implemented to recognize or classify particular events (e.g., earthquake).

Particular attention should be paid to the robustness of the algorithm, in order to avoid false positives, i.e., the acquisition of unwanted signals, and false negatives, i.e. the non-acquisition of significant signals.

4.9.2 Data processing

The raw data coming out of the acquisition system are:

- Single numerical values in the case of static quantities (quantities which vary very slowly over time).
- Temporal sequences of data acquired at a rate appropriately chosen for the needs of the monitoring system in the case of dynamic quantities.

Raw data are hardly significant in itself and, therefore, are subjected to a process that transforms them into valid information to be used for the interpretative models subsequently employed to recognize the behaviour of the structure and to update the interpretative model.

In principle, three successive processing phases can be identified:

- Data pre-processing.
- Evaluation of the significant quantities to be monitored.
- Updating of the interpretative models.

4.9.2.1 Data pre-processing

Data pre-processing aims to improve the quality of the measured data without, however, transforming the quantity under consideration. The main steps consist of:

- Elimination of noise effects, elimination of spikes (anomalous readings), zero drifts, etc.
- Control of the correctness of the acquisition and a verification that a full scale has been chosen that does not lead to instrument saturation or to values that are too low, close to the resolution of the analogue/digital converter.
- Transformation into engineering values.
- Data validation using structure symmetries or with correlations in case of redundant values.

4.9.2.2 Evaluation of the significant quantities to be monitored

The evaluation uses the data recorded and validated after pre-processing to calculate the quantities identified as system-specific objectives (behaviour parameters or state variables). In this case, the operations, of various degrees of complexity, generally involve a real transformation of the measured quantity into other related quantities. Typical processes are, for example:

- Integration of accelerometric values to obtain the displacement.
- The determination of transfer functions.
- The separation of effects into quantities which depend on several causes.
- The determination of modal parameters.





4.9.2.3 Updating of the interpretative models

The updating includes all the operations required by the damage identification processes carried out through the use of interpretative models that lead to updating the parameters of the numerical model of the structure.

4.10 Data treatment

The data acquired by the instrumentation, once stored and referred to homogeneous periods of acquisition, are processed to obtain time series that can be analysed with the interpretative models and with the recognition processes of the state of the structural system.

In particular, the algorithms used in this phase of the process aim to eliminate noise affecting signal quality, eliminate drifts and non-significant frequencies, recognize and eliminate spurious readings, integrate missing data and finally separate the effects of non-significant environmental components in order to perform the necessary compensations.

A large number of procedures, generally statistical, and numerical filters are available for this purpose and can be conveniently applied.

Other important algorithms for data processing are those aimed at the compression (data reduction) of time series. In the case of long-term monitoring, in fact, the amount of data quickly becomes very large, requiring special informatic solutions for storage and preservation (e.g., clouds), and specific algorithms to optimize the occupation of space and retrieve data efficiently and quickly.

For further information, see D4.1.

4.11 Bridges monitoring

To analyse the behaviour over time of a bridge during the operational life and in case of occurring events that can affect the stability of the bridge itself, it is advisable to use a monitoring system which consists of several instrumentation types. The aim of this section is to provide general guidance on the instrumentation that can be used in a monitoring system. The following elements are analysed, providing for each one an example of a possible monitoring system able to describe its static/dynamic behaviour and, eventually, to detect the appearance and/or evolution of degradation/damage processes:

- Deck
- Piers/abutments
- Bearings
- Joints
- Prestressing cables
- Stay-cables/hangers

It should be noted, though, that the information contained herein represent only an example of the type of instrumentation that could be installed, given that other technologies could be used, either new or existing.

4.11.1 Deck

The information regarding the instrumentation to be used for a bridge deck monitoring system, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.1.

1		
*		*
÷.		ž
*		



ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
		MODAL PARAMETERS	HIGH
	ACCELEROMETER	VIBRATIONAL LEVELS	HIGH
	CLINOMETER	DEFORMATION (DISPLACEMENTS/ROTATION)	MEDIUM
DECK	CRACKMETER	CRACK WIDTH	MEDIUM/LOW
	THERMOCOUPLE	TEMPERATURE	LOW
	HUMIDITY SENSOR	HUMIDITY	LOW
	CORROSION SENSOR	RESISTIVITY INDICATOR, POTENTIAL DIFFERENCE, PH	LOW

Table 4.1 – Deck instrumentation.

- Accelerometers are devices that measure the vibration or acceleration of the element on which they are fixed. They measure linear acceleration and allow the analysis of vibrations and structures dynamic behaviour. Aiming to obtain a correct dynamic representation, the essential measuring points are quarter and middle span. Additionally, in case of girders, it is recommended to monitor at least the lateral beams, which are the ones more likely to be vulnerable to damage. Besides, in order to correctly derive modal shapes, global vibrational levels and to identify asymmetries and torsional components in vibration modes, it is advisable to install at least two accelerometers for monitored cross section, so that the deck transversal behaviour may be comprehensively described. However, accelerometers positioning must be properly chosen, based on the specific structure needs.
- **Clinometers** analyse structures static behaviour. They allow the detection of structures elements deformations, highlighting the appearance of eventual damage processes. The number of measuring points must be sufficient to allow a correct reconstruction of the deformed shape of the structure. In case of girders, it is recommended to monitor at least the lateral beams, which are the ones more likely to be vulnerable to damage. A minimum number of five clinometers per beam is generally recommended.
- **Crackmeters** are devices that allow to monitor the evolution of cracks width. They are placed on existing cracks to assess their evolution over time or where it is likely that a crack may be appearing.
- **Thermocouples** measure the temperature, thus assessing the environmental conditions in which the structure is located. It is recommended to have at least one probe at the intrados and one at the extrados in order to be able to detect both uniform and variable temperature variations along the height of the section. Temperature sensors location should also be chosen according to the exposure of the structure (sun/shade) to fully characterise the structure behaviour. It is advisable to have at least one thermocouple per monitored structure, to assess the environmental conditions and to eventually correlate changes in monitoring parameters values to thermic phenomena that affect the structure static and/or dynamic behaviour.
- **Humidity sensors** measure humidity and, thus, assess the environmental conditions. Typically, one sensor per monitored structure is sufficient, assuming the punctual measurement can be generalized to the entire structure.
- **Corrosion sensor** can be of various types depending on the desired measurement. Since a local measurement is given, sensors should be placed at points where corrosion has already occurred. This kind of instrumentation can also be implemented for a general measure of the environmental conditions (e.g. pH measure) to assess whether the environment where the structure is located is unfavourable or favourable for potential corrosion phenomena.







Associated with document Ref. Ares(2020)3731189 - 15/07/2020



Figure 4.4 – Deck instrumentation (Adjusted from [19]).

4.11.2 Piers/abutments

The information regarding the instrumentation to be used in a pier/abutment monitoring system, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.2.

ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
	ACCELEDOMETER	MODAL PARAMETERS	HIGH
	ACCELEROIVIETER	VIBRATIONAL LEVELS	HIGH
PIERS/ABUTMENT	CLINOMETER	DEFORMATION (DISPLACEMENTS/ROTATION)	MEDIUM
	CRACKMETER	CRACK WIDTH	MEDIUM/LOW
	CORROSION SENSOR	RESISTIVITY INDICATOR, POTENTIAL DIFFERENCE, PH	LOW

Table 4.2 – Piers/abutments instrumentation.

- Accelerometers are devices that measure the vibration or acceleration of the element on which they are fixed. They measure linear acceleration and allow the analysis of vibrations and structures dynamic behaviour. Aiming to obtain a correct dynamic identification and to measure accelerations, the essential measuring points are the top and the bottom of the pier. In this way, modal shapes that involve or are governed by piers can be derived, and the accelerations transferred from to bottom of the structure to the deck in case of extreme events, such as earthquakes, can be evaluated. Additionally, in case of frame piers, it is recommended to monitor all piers or at least the outer alignments. However, accelerometers positioning must be properly chosen, based on the specific structure needs.
- **Clinometers** analyse structures static behaviour. They allow the detection of structures elements deformations, highlighting the appearance of eventual damage process. It is recommended to monitor all the structure piers and, in case of frame piers, to install clinometers at least on the outer ones (on the two opposite alignments) in order to detect eventual differential settlements. It is advisable to install at least two clinometers per pier (top and bottom).
- **Crackmeters** are devices that allow to monitor the evolution of cracks width. They are placed on existing cracks to assess their evolution over time or where it is likely that a crack may be appearing.





• **Corrosion sensor** can be of various types depending on the desired measurement Since a local measurement is given, sensors should be placed at points where corrosion has already occurred.



Figure 4.5 – Piers/abutment instrumentation (adjusted from [20]).

4.11.3 Bearings

The information regarding the instrumentation to be used for a bearing monitoring system, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.3.

ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
BEARINGS	DISPLACEMENT TRANSDUCER	DISPLACEMENT	MEDIUM*
	CLINOMETER/ ROTATION TRANSDUCER	DEFORMATION (DISPLACEMENT/ROTATION)	MEDIUM
	LOAD CELL	BEARING REACTION	LOW

* HIGH in case of sudden events, e.g., earthquakes..

Table 4.3 – Bearings instrumentation

The objective of bearing monitoring is to measure their displacement and to detect any anomalies in their behaviour, as well as to assess, in the case of isolated systems, the relative displacement between deck and piers. Each bearing should have its own monitoring system. In order to measure bearing displacements/rotations in space in the three dimensions, it is advisable to install displacement transducers, clinometers or rotation transducers. In addition, if the support reaction wants to be derived and monitored, a load cell can also be installed.







DISPLACEMENT TRANSDUCERS, CLINOMETER, ROTATION TRANSDUCER, LOAD CELL

Figure 4.6 – Bearings instrumentation (Adjusted from [21]).

4.11.4 Joints

The information regarding the instrumentation to be used for a joint monitoring system, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.4.

ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
JOINTS	ACCELEROMETER	VIBRATIONAL LEVELS	HIGH
	DISPLACEMENT TRANSDUCER	DISPLACEMENT	MEDIUM/LOW

Table 4.4 – Joints instrumentation.

Joints are the connecting elements between two spans, and, as such, are particularly susceptible to deterioration and require continuous maintenance. In order to monitor any maintenance issues, accelerometers should be installed to identify joints vibrational levels so as to associate any abnormal impulses with localised damage. In addition, it is advisable to install displacement transducers to assess the relative displacement between spans across the joint. If possible, the displacement transducer should be installed at both endpoints of the investigated span.







Figure 4.7– Joints instrumentation (adjusted from [22]).

4.11.5 Prestressing cables

The information regarding the instrumentation to be used for a prestressing cable monitoring, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.5.

ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
PRESTRESSING CABLES	ACCELEROMETER	VIBRATIONAL LEVELS	HIGH
		MODAL PARAMETERS	HIGH

Table 4.5 – Prestressing cables instrumentation.

Accelerometers are used to assess any abnormal vibrations and, above all, to identify any changes in modal parameters that could be attributed to a variation of stress in the cable (loss of prestressing force and/or reduction of the cross-section area, e.g. due to corrosion). It is recommended to install at least one accelerometer per segment between two constraint points (header/damper), and to add measuring points in the weak zones and as many as deemed appropriate to collect information along the entire longitudinal development of the cable.



Figure 4.8– Prestressing cables instrumentation (adjusted from [23]).

4.11.6 Stay-cables/hangers

The information regarding the instrumentation to be used to monitor stay-cables and hangers, the corresponding monitoring parameters and the suggested sensors frequency of acquisition is summarized in Table 4.6.





ELEMENT	INSTRUMENTATION	MONITORING PARAMETER	FREQUENCY OF ACQUISITION
STAY- CABLES/HANGERS	ACCELEROMETER	MODAL PARAMETERS	HIGH
		VIBRATIONAL LEVELS	HIGH
	STRAIN GAUGE	DEFORMATION/STRESS	LOW*

*HIGH in case fatigue behaviour is investigated

Table 4.6 – Stay-cable/hangers instrumentation.

Accelerometers are used to assess any abnormal vibrations and, above all, to identify any changes in modal parameters that could be attributed to a variation of stress in the cable. Moreover, strain gauges can be installed in areas where damage has occurred to measure and monitor the phenomenon evolution.



ACCELEROMETERS, STRAIN GAUGES

Figure 4.9 – Stay-cable/hangers instrumentation (adjusted from [24] and [25]).

4.12 Tunnels monitoring

Tunnel monitoring is essential for short- and long-term stability monitoring of the tunnel itself, as well as the surrounding soil and adjacent buildings.

Monitoring can be carried out both during the excavation of a tunnel and for an existing gallery.

All the stresses that are caused by the excavation of a tunnel are due to multiple factors, among them the most important relate to the geological-stratigraphic characteristics in which the structure is located, the geotechnical and mechanical parameters of the materials, the excavation methodologies and, lastly, the geometric and elastic characteristics of the linings.

The analysis of the information above leads to a correct decision-making for the design of monitoring systems.

To analyse the behaviour over time of a tunnel during the operational life and in case of occurring events that can affect the stability of the tunnel itself, it is necessary to use a monitoring system which consists of several instrumentation types.

4.12.1 Monitoring system design

The flowchart presented in Section 4.5 can be applied also in the case of tunnels. The fundamental information to be defined is summarized as follows:

- The quantities to be measured (displacements, stresses, interstitial pressures, free surface position, etc.) as a function of the monitoring objectives;







- The location of the measurement points (on the surface, in the cable, fixed stations, measurement stations, etc.);
- The instrumentation used to perform the measurements;
- The temporal frequency of the measurements;
- The criteria for recording and processing the measurements and the transmission of data to the stakeholders;
- The threshold values of the measures and related provisions to be adopted. The criterion adopted to define these values must be based on the results of the design calculations and on the experience acquired in similar structures.

4.12.2 Displacement measurements

Displacement monitoring requires ad hoc instrumentation, which depends on the type of displacement to be measured and should be placed in an adequate number of cross-sections deemed significant.

The main displacement measurements are described in the following chapters.

4.12.2.1 Convergence measures

"Convergence" is defined as the decrease in the distance between two points of the tunnel. It is calculated by measuring the distance between target points fixed to the excavation walls or to the linings. Through the convergence measures it is possible to check the response of the support system, comparing their values with thresholds and identifying any warning signs of instability (Figure 4.10).



Figure 4.10 – Example of a measure of convergence as a function of time

4.12.2.2 Measurements of mass displacements

It is also possible to detect rock mass displacements.

Their measurement allows to evaluate the deformations, mainly radial, around the cavity and consequently to evaluate the extent of the mass area affected by the excavation.






Figure 4.11 – Typical arrangement of strain gauges for measurements of mass displacements [26]

4.12.2.3 Measurements of surface displacements

In the case of "superficial" tunnels, subsidence, and horizontal displacements of both the surface and the rock mass can be measured from the surface.

The objective of surface monitoring is to control the movements of the rock mass, especially in the presence of potentially unstable slopes or escarpments.

The tools used to measure these surface displacements usually include:

- hydraulic levelling chains for the continuous measurement of height variations;
- Topographic levelling for the survey of the level of the spreading of strong points placed on the surface;
- Single-base and multi-base strain gauges to measure the vertical displacements along the tunnel axis and along the vertical ones close to the cable;
- Inclinometers to measure horizontal displacements on the sides of the tunnel;
- Wall clinometers, often installed on pre-existing structures to measure its rotations.



Figure 4.12 – Ground movement measurement by an inclinometer [27]

4.12.2.4 Stress measurements

Tunnels monitoring systems also implies the evaluation of the stresses in the linings and their interaction efforts between the ground/rock mass and the structures. Through the stress measurements it is possible to monitor the stress condition to which the structural elements are subjected.





The instrumentation used for the measurements usually consists of:

- Pressure cells for the measurement of radial stresses at the mass-lining interface and measurements of radial and circumferential stresses within final concrete linings;
- Load cells often installed at the foot of the ribs and at the piers-shell connection to measure the compression stress in the ribs;
- gauge bars for local deformation measurements. From these measurements, normal voltage values parallel to the axis of the bar are obtained. They can be applied on the ribs or inserted inside the final covering.



Figure 4.13 – Example of application of pressure cells (Extract from [28])

4.12.3 Standard tunnel monitoring system section

An example for a standard tunnel monitoring system section is provided in Figure 4.14, which includes the number and the positions for the devices related to convergence and loads measurements and the assessment of the state of deterioration. Convergence measurements may be performed through the use of optical sights or laser scanners. It is also possible to evaluate the load to which the tunnel is subjected:

- Stress control: may be achieved with pressure cells or flat jacks;
- Deformation measurements: deformation may be measured by means of MEMS inclinometers and accelerometers or by means of optical fibre technology.

Lastly, it is possible to check the state of deterioration of the tunnel using Georadar.







Figure 4.14 – Tunnels monitoring system instrumentation







5 Damage and degradation processes

Engineering structures are a key component of the infrastructure system and their long-term performance in service plays a critical role on the structural health management. Civil infrastructures are exposed to various external loads such as earthquakes, gusts, environmental actions, traffic, and wave loads during their lifetime, which can lead to damages and degradation processes which need to be taken into account to avoid structural problems and to increase the structural safety and serviceability [29].

One of the most important aspects during the service life of structures is ensuring their safety in time and the early detection of eventual damages and degradations. The principal problem is that structures may get deteriorated and degraded with time in an unexpected way (some damage processes are time-variant dependent, most of them are unavoidable and occur in many cases, even may not be recognizable by visual inspection). This leads to structural failures causing costly repair and heavy loss of human lives. Identifying the principal sources involved in damage and degradation processes is fundamental for understanding them [29]. This chapter presents an overview of the information available in **Error! Reference source n ot found.** regarding the existing damage and degradation processes for bridges and tunnels.

5.1 Damage and degradation processes identification

Damage and degradation processes are classified in two main groups regarding their source. The first group of damages is caused by environmental or external factors (physical, chemical, and biological processes). The second group contains information of damages caused by design and construction issues. in addition, two structural materials are considered to characterize damage and degradation processes: concrete and steel. **Error! Reference s ource not found.** presents in a general way definitions for each damage and degradation process within the scope of the project. All descriptions are focused on explaining how the damage processes are developed in the structure, describing their significant causes and consequences.

5.1.1 Environmental exposure or external factors

Engineering structures are exposed to different environmental conditions (e.g., temperature, wind, precipitation, freeze, traffic), which in combination with other factors lead to damage and degradation processes. These processes can significantly affect material properties. Structure exposure to environmental conditions generally governs the material durability requirements, materials selection, material design, and construction of the structure. Therefore, structures exposure has a high influence on their materials durability, maintenance works, and their costs.

A distinction between physical, chemical, and biological processes is made to understand the effects of environmental exposure and external actions.

5.1.1.1 Physical processes

Damages that lead to a physical disruption in the structure or its components are considered physical damage processes. These processes are usually guided by external actions and their influences, affecting the current or future functionality of the structure. In, **Error! Reference s ource not found.** the descriptions of all physical processes considered can be found, while **Error! Reference source not found.** and **Error! Reference source not found.** show the pr oposed DIs and available technologies for each.

5.1.1.2 Chemical processes

The chemical processes considered involve the dissolution of substances or chemical reactions between components of the materials. Reaction products might cause problems due to dissolution or expansion, leading to a chemical issue. These processes are usually





prolonged in time, and highly dependent on the material micro-structure such as organic, carbonate, pH value, and sulphate and sulphide ion contents.

5.1.1.3 Biological processes

Biological aggressors such as microorganisms, algae, fungi, and various bacterias can generate biological processes in structures, affecting their service life performance. While **Error! Reference source not found.** describes principal pathologies due to biological p rocesses, focusing on biological growth in substructures, **Error! Reference source not found.** links these pathologies with the proposed damage indicators. Finally, **Error! Reference source not found.** shows the available technology to identify and monitor these biological processes.

5.1.2 Design and construction issues

All structures are prone to be damaged by the construction or rehabilitation of ancillary equipment and services, such as installing additional safety fences and repairing utility apparatus, respectively; these types of processes are covered as design and construction issues. Processes that are developed due to inadequate practices, inadequate design, or lousy construction methods are directly related to poor construction practices, which usually lead to durability issues. **Error! Reference source not found.** defines all design and c onstruction issues in structures.

5.2 Damage indicators

Error! Reference source not found. links damage and degradation process with damage i ndicators (DIs). Damage indicators are carefully assigned to each damage process to understand how the structure is affected and how the combination of them can worsen the structure condition. An overview of damages and degradation mechanisms can be found in Chapter 2.

5.3 Surveying technologies

Error! Reference source not found. contains surveying technologies addressed in the p roject in the Document [11] and related to the detection of specific damage processes mentioned in the Document D2.2, Chapter 4.1. The surveying technologies uniquely considered are a part of broad spectrum category available on the market – however in the document there is mentioned selected group, based on the future potential perspectives and best practices in EU countries.

Tables presented in the Document D2.2 **Error! Reference source not found.** attribute to s pecific damage mechanism a set of surveying technologies that can be proposed for detection and/or quantification of the damage related to bridges and/or tunnels.

Selection of the parameters has been taken into account when comparing surveying technologies – detection effectiveness, accuracy, availability and versatility. Rating scale has been prepared subjectively – should be treated as an overview and preselection suggestions, since the damage mechanisms can be very complex and the choice has to be adjusted to the specific case or more than one surveying methods selected.

Having the description of each surveying methods in the Document [11], and technology explained it is possible for one to consider different factors important in decision on suitability and reliability of the method as a whole and resultant of variables. Apart from the technology itself, it is also quite important to acknowledge whether the method which is the most suitable for the case is also available in the area/country of interest. Depending on the priorities it is obvious to assess, which method will meet the requirements and also will be economically reasonable.





The section is organised as follows – for each damage process there is a table of techniques dedicated for its detection or symptoms of it according to damage indicators associated with the process. Each surveying technology is rated In the colour scale starting from red colour representing the low level of adjustment, through low-medium/moderate in orange, to moderate-high and high level of adjustment. The rating has been prepared based on the literature review performed in [11] as well as by studying the real cases documentation and the experience of the consortium members. Essential information for each case is to guide the recipient into practical aspects of the decision making process for the large group of surveying methods presented in [11]. with respect to damage process, performance and damage indicators.

The tables demonstrate essential surveying technologies for the damage process and the distinction between them, as it is can be seen – for some of the damage processes the selection is narrowed, which on one hand make the choice easier, but at the same time – it can be also a disadvantage. In other cases there can be quite a variety of techniques, but with similar characteristics – then the choice depends mostly on individual preferences and the budget involved.

5.4 Data analysis methods

Error! Reference source not found. lists all relevant surveying technology and for each t echnology it indicates the applicable and recommended data analysis methods, describing the best practices, state of the art, recent approaches, and further references.





6 Damage characterization procedure

6.1 General

Throughout this deliverable, along with [11], the different phases experienced in the identification of a damage in an infrastructure have been explained. It comprehended from its detection to the assessment via inspections (Chapter 3) and surveying technologies ([11]), to the monitoring of the structure (Chapter 4). The purpose of this chapter is to summarize and organize this knowledge following a generalized workflow divided in the different phases of the damage characterization procedure. As such, Chapter 6.2 covers damage detection and localization, while Chapter 6.3 deals with its parametrization and assessment. The next two phases, modelling and prediction and monitoring, are explained in Chapters 6.4 and 6.5 respectively. These phases are tightly linked together, as the data generated from monitoring can be used to update models, and the predictions can be checked through monitoring. Nevertheless, modelling and prediction are possible the moment you first identify the damage, it is placed before monitoring. These four phases are represented by the flowchart seen in Figure 6.1.



Figure 6.1 – Damage characterization procedure phases

6.2 Phase 1: Damage detection and localization

In order to identify, monitor and model a damage, it is necessary to assert the likelihood that there is a damage in the structure. It must be detected and located with the required level of refinement. This phase is undertaken by the different inspections that can be performed in a structure. In some cases, the damage might be first detected as a result of continuous monitoring, which are explained in Chapter 4 and Chapter 6.5. However, the structure will undergo an inspection regardless, in order to assert the monitoring detection and to determine whether the damage exists in the structure or not.

Since Chapter Inspection provided a detailed view on the type of inspections, along with their motivation and outputs, this chapter will summarize it to better illustrate the damage detection procedure.

There are three levels of inspections: (i) routine maintenance inspections; (ii) conditional maintenance inspections; (iii) extraordinary inspections. Apart from the routine maintenance inspections, the other levels use the reports of lower-level inspections as starting data to better understand the state of the structure. Therefore, the inspection level that first detects the damage is usually the routine maintenance inspection.

This type of inspection aims to verify the serviceability of the structure and is approximately done once per year. It is performed by visual inspection of the structural components in order to detect the possible damages. In order to plan the inspection, the inspector has to check the available documentation and resources at his disposal. In the first place, he should review the structural records, which contain: characteristics of the structure, hazards, the condition of the structure at the time of the last inspection, any worsening of defects over time, significant





maintenance or modifications since the last inspection, analysis and decision regarding hidden or enclosed structural features that should be accessed. Further steps include risk assessment and methods selection before inspection [30]. Besides the structural records, historical data about fire events, evidence of vandalism, burnt vehicles, other incidents, storage of combustible materials in vicinity of the structure should also be studied. Finally, the location itself must be reviewed. Before data collection, the area around the bridge and nearby access terrain to the designated spaces must be checked, including the clearing of vegetation, and studying the lightening conditions. Also, the relevant areas of study and what information should be collected about the bridge or tunnel are to be set in this stage. It should also include the basic geographical information of the target and surroundings (for instance if satellite imaging will be used), traffic frequency, nearby roads, trees, buildings shall be considered prior to inspection [29]. Once the inspection takes place, its results are reflected in a report that can include both pictures and the inspector comments, to further detail the state of the damage or characterize the possible omissions of the process.

6.3 Phase 2: Damage parametrization and assessment

Once the damage has been located, it is necessary to determine the value of its parameters, including uncertainty, and impact on the overall structure. To do so, the adequate surveying technology or test must be selected, which in turns requires to estimate the physical parameter or event that is to be detected, such as the width of a crack or the erosion of the concrete. The location of the damage must also be addressed, as it might not be superficial, removing certain options.

One of the main challenges when estimating the extent of a damage, is the relationship between the data acquired and the damage. This means how the data taken can serve as a way to estimate its location, extension and impact on structural performance. This challenge is tightly related to the techniques or technologies used to gather data. It also depends not only on the technology itself, but on the damage that is being estimated. The same technique might not be related in the same way to two different kinds of damages.

The different surveying technologies, along with the parameters that they measure, setup, environmental dependency, and their advantages and disadvantages, have been discussed in [11]. Therefore, similarly as in Chapter 6.2, this chapter will represent a support for the information contained in the mentioned deliverable. To do so, Table 6.1 has been created to serve as an index of the different surveying technologies reports contained in [11], together with some others that are out of the scope of IM-SAFE. The purpose of this table is to ease the search for a fitting surveying technology by highlighting the different events/properties that they measure, along with their application, in a single table.





Surveying Technologies	Measured event / property	Application
LiDAR	Geometric information + Intensity	Asset monitoring
Satellite	Changes over time	Asset monitoring
GPR	 Masonry arch bridges Unknown geometries remaining in the interior of the bridge Evidence of restorations and/or reconstructions Existence of cavities and fractures/cracking in masonry Moisture in masonry Moisture in masonry Inventory of bridge foundations Filling distribution in masonry Thickness of ashlars (pavement, ring arch, spandrel walls, etc.) Concrete bridges Estimation of concrete cover depth. Mapping reinforcing bars (deck and beams). Location of cable ducts and other utilities such as deck joints. Damage detection on concrete (corrosion, cracking, etc.). Moisture detection and water content estimation. Tunnels Thickness of concrete segment/lining. Thickness of the backfill grouting layer. Damages in concrete lining and grouting layer. Damages (e.g., cracks/fissures and voids) behind tunnel linings. Moisture/water content. Depth and location of reinforcement (rebar). Inspection of immersion joints. 	Masonry and concrete bridges. Tunnels
Magnetic and electrical methods	material. Detection of corrosion in post-tensioned concrete elements. Qualitative analysis, such as localization of the rebars in structure, determination of their dimensions and diameter of cover Characterization of durability of concrete elements	Post-tensioned concrete elements
Water resistance test	and durability of the surface protections. Determine the space of pores and absorption rate in the concrete. Determination of the effectiveness of hydrophobic surface agents used in securing building materials from the influence of water	Concrete
Acoustic Emission Techniques (AE)	 Detection of dynamic processes in materials, Detection of leaks, Detection of flaws, Tracking of degradation processes in concrete, Detection of damage mechanisms related to corrosion. Level of intensity of cracking processes, Integrity testing of metallic structures, Integrity testing of concrete structures. 	Evaluate possible catastrophic failures or the level of damage. Link the degree of damage in the structure with the operating conditions of the facility.





Boroscopy and Endoscopy	Detect and photograph abnormal sections on the structural elements of the bridge or tunnel with cracks or deformations or affected by corrosion or chemical attack.	Diagnostics of civil engineer structures
Fibre Optic Sensors (FOS)	 Strain Deformation Temperature Vibration Pressure Acceleration Inclination 	Long-term monitoring and remote control of the condition of facilities
Guided Waves Propagation (GW) techniques	 Discontinuity from the wave signal diffracted by the crack. Detection of delamination and debonding. Surface cracks depth Homogeneity of concrete Quality variation of concrete Detection of voids, imperfections Determination of the age of concrete. 	Detection of the damage in structural health monitoring of the reinforced concrete
Mechanical tests on cored samples.	Characterization of concrete properties	Reliable assessment of the safety of bridges or tunnels
Chemical Methods	Carbonation front depth	degradation of the structure
Quantitative chemical methods.	 Corrosion risk of rebars resulting from the influence of chlorides ions. Capacity of concrete to resist chloride ions penetration. Detection of corrosion parameters of reinforcement Expansion, cracking, strength loss and disintegration due to sulphate ions. Quantification of harmful ions. 	Quantitatively assessment of the rate of degradation
Radiological and Nuclear Methods	•Cracks dimensions •Early signs of corrosion •Microcracking progress •Quantification of water movement	Assessment of reinforcement characteristics and distribution in reinforced concrete structures.
Surface measurements	Compressive strength and hardness of concrete elements	Monitoring concrete structural elements of bridges
Water penetration test/ Permeability test	Resistance or durability of concrete under hydrostatic pressure	Bridges or tunnels diagnostics
Weight in Motion systems (WIM- Systems)	 Weight of the vehicles (estimation) Axle group loads, axle loads, wheel loads of the passing vehicles Tire impact forces Strain forces Velocity of the vehicles 	Prevention of the overload of the structures
Micro Electro- Mechanical Systems (MEMS) - Accelerometers.	Linear acceleration	Asset monitoring. Analysis of vibrations and structures dynamic behaviour.







Micro Electro- Mechanical Systems (MEMS) - Clinometers.	Inclination with respect to the horizontal axis	Asset monitoring. Analyse structures static behaviour.
Crackmeters	Crack width	Monitoring of existing/new cracks on bridges or tunnels
Slope clinometers	Slope displacements in the subsurface	Evaluation of eventual relative ground movements near natural slopes, embankments and retaining walls. Monitoring of landslides evolution.
Piezometers	Static water level or hydrostatic pressure in the subsurface	Foundations of tunnels or other structures
Flat Jacks	Stress	In situ stress measurements on tunnel linings and bridge piers.
Displacement Transducers	Displacements	Bridge performance evaluation (e.g., bearing displacements).
Linear Polarization Resistance and AC impedance measurements	Corrosion rate	Laboratory method of determining steel loss during the corrosion process
Radiographic On- site Testing	Location of reinforcement, occurrence of honeycombing	In situ detection
Neutron Activation Analysis	Residual radioactivity	Laboratory method for concrete samples
Ponding test	One-dimensional chloride ingress profile	Laboratory method for concrete samples
Electrical Impedance Spectroscopy	Carbonation depth progress	Laboratory method for studies of corrosion kinetics, morphology of the corrosion
Alternating Current Field Measurements	Crack length, depth sizing of surface cracks	In situ Inspection of fillet welds in highway bridges
Infrastructure Corrosion Assessment Magnetic Method (iCAMM™)	Reinforcement condition monitoring	Condition of bridge culverts
Scanning Electron Microscope (SEM)	Composition of concrete, changes in relationship between constituents as a result of aging or damage process	Laboratory method for microscopic investigation of hardened concrete
Mercury Intrusion porosimetry	Distribution of pore sizes in cement-based materials	Laboratory method for concrete samples





Active Thermal Imaging/infrared thermography (IRT)	Detecting subsurface deteriorations/ Area and depth of subsurface delamination	Structural Health Monitoring
Hyper Spectral Imaging/ UV/VIS NIR	Water-to-cement-ratio, different curing times	Laboratory method for concrete curing assessment
Laser-Induced Breakdown Spectroscopy	Chloride detection	On-site and laboratory method for concrete structures

Table 6.1 – Surveying technologies summary table

6.4 Phase 3: Damage modelling and prediction

As stated at the beginning of Chapter 6.1, damage modelling and prediction is tightly related to its monitoring, which is described in Chapter 6.5. The reason is that data monitoring constantly generates vast amounts of information in heterogeneous formats as a result of SHM systems, inspections and surveying technologies. This data is stored and managed accordingly to their type and purpose, creating a historical record as the structure advances through its service life. However, the first iteration of the damage model, as well as some early predictions, can be constructed with the first identification of the damage.

With this information, the damage geometry and characteristics can be modelled using appropriate techniques, such as finite elements, and are to be expressed using the appropriate format, such as the Industry Foundation Classes (IFC) or LASer (LAS). Also, the historical data can be transformed into predictive models that can indicate the expected behaviour of the damage. This, in return, facilitates maintenance planning as it can be based on grounded predictions. The purpose of this chapter is to describe the different ways to model a damage, both geometrically and physically, as well as to present the techniques used to predict the evolution of the damage.

6.4.1 Methods for damage modelling

The accurate and reliable structural models must consider the damages that have been estimated and measured in previous phases, along with their uncertainties. Therefore, in this chapter, two different methodologies of damage modelling will be explained.

Geometrical Damage Modelling: In this methodology, the damage is physically represented in the structural model. This modelling can be tackled according to different dimensions depending on the type of information desired. For instance, 3D models can be used to represent the geometry of a solid damage, as well as its position in the structure. These damage models are to be expressed in an appropriate format and are often modelled using the IFC data schema in a BIM model, since these models allow the introduction of semantic information that can further characterize the damage [31]. For instance, [32] modelled a beam crack in IFC 4 using different available software. However, 3D models are computationally expensive, and are used only in special cases where a high level of detail is required. As such, 2D and 1D model simplifications are employed to reduce the computational burden. In 2D models, which use surface bodies, the damages can be expressed as a reduction of the thickness of the body (e.g., if there is some loss of material) or by modifying the stiffness of the contact between bodies (e.g., a damaged connection that turns from a fixed to a pinned connection). For example, in [33], the authors randomly modified the thickness of the elements of a mesh of a surface body to study the variation of its ultimate strength. Concerning 1D models, as they are limited by the deeper level of simplification, usually express the damage either as a variation of the stiffnesses in contacts, or by introducing some releases (e.g. plastic hinges in cracked beams). In [34] the authors update a structural model of a bridge considering





the nature of its connections by modelling them in different ways (pinned connection, rigid connection and semi-rigid connection using rotational springs).

Physical Damage Modelling: In this approach, the appearance of a damage is bound to change the physical properties of the material. Hence, the damage is not considered by a physical modelling but as an event that modify some of the physical properties in the material. For instance, [35] analysed samples of sea corroded steel and extracted the stress-strain curve of the material to study its variation in presence of corrosion. [36] induced shear deformations in a steel beam to observe the microstructural changes of the material and how they affect the material mechanical properties. [37] & [38] tested and analysed a damaged beam in order to tune their analytical model so that the digital behaviour matches the real word behaviour.

It should be noted that a structural model can use both methods for modelling the damage of the body it represents. Therefore, it is possible to develop a 3D structural model of a tunnel that has the cracks represented geometrically, as well as variations in its material due to the changes in the microstructure that the damage induces in it.

6.4.2 Prediction of damage evolution

The creation of models that estimates the effect of damages on structures is a very useful tool for the prediction of their evolution. In fact, there are multiple methods that have been developed to study and predict them These comprehend from simple regression models to more sophisticated methods such as machine learning or deep learning techniques.

Once the damage of the structure has been modelled, datasets can be created for the training of the algorithms dedicated to the prediction of the evolution of the damage. These datasets must be created by sampling the vector space of the model that represents the damage of the structure, which was modelled using all the information collected in previous sections over the years. Once enough data is obtained, the datasets have to be used to train and validate the commented algorithms, which can be carried out for starting the prediction of the evolution of the studied damage. It should be noted that each of the algorithms commented below are based in particular methodologies or structures:

Linear and non-linear regression: This type of regression describes the relationship between dependent variables and one or more independent variables in which the dependent variables are considered the response or prediction variable. Depending on the nature of our dataset and the behaviour of our variables different methodologies can be used such as simple linear regression, stepwise, multiple, regularization or even mixed regressions based on linear and non-linear effects, more information about the different methodologies can be found in [39] and [40].

Support Vector Machines: This methodology is considered a nonparametric technique that attempts to learn a functional relationship between some pairs of inputs and outputs given an experimental design or dataset. It is considered an extension of the non-linear regression based on the principle of structural risk minimization as is described in [41] and [42].

Polynomial chaos Expansions (PCE): This is a powerful surrogate modelling technique that aims as providing a functional approximation of a computational model through its spectral representation on a suitable built basis of polynomial functions [43], different applications and a more detailed explanation can be found in [44] and [45].

Gaussian process regression (GPR): Also called kriging, is a stochastic algorithm that perform a statistical interpolation method that capitalizes on a gaussian process to interpolate a wide range of complex functions [46]. With the introduction of the GP-regression [47] it is able to support noisy data and has become a widespread use in machine learning. An applied case in reliability analysis will be [48].

Other Machine learning or deep learning methodologies: This field is one of the fields that has experienced in recent years in which different types of architectures and methodologies Page | 85





have been implemented for the prediction of all types of applications. Depending on the damage type, its nature, how it develops and changes, the prediction of its evolution can be carried out by different architectures. The different architectures can be classified in two groups, supervised and unsupervised learning. The following table performed in [49] compares some of the most used methodologies y deep learning. More methodologies and information can be obtained in [50] and [51].

Furthermore, if the damage evolution is properly estimated, it is possible to calculate the threshold parameter values that will ensure the safety of the structure. For achieving this purpose, a calibrated numerical model of the structure needs to be generated. The calibration can be achieved by using several model updating techniques to minimize an error function that compares the numerical and experimental structural responses. Starting from the calibrated numerical model, its parameters can be constantly modified to represent the evolution of the damage and to predict the consequences it will have on the structural health response. Therefore, when the failure of the structure is reached, the value of the corresponding parameter can be identified and defined as a threshold value. These values are of real interest for planning the damage monitoring of a structure.

6.5 Phase 4: Damage monitoring

As explained in Chapter Monitoring, the monitoring of a structure aims to determine the state of a structural system and its evolution over time. This also includes the identification and parametrization of phenomena such as degradation and damages, along with their uncertainties. The monitoring intention is that of evaluating the integrity of the system and its ability to perform its function in adequate conditions for a given time. There are different types of monitoring approaches, depending on the intention of the monitoring itself. However, health monitoring is used to obtain real time information that can inform the estimation of the safety and serviceability of a structure.

Structural Health Monitoring (SHM) is the process of implementing a damage identification strategy. This process involves the observation of the structure over time by obtaining periodically spaced measurements. Then, these measurements should be transformed into damage-sensitive features that are to be analysed to estimate the current state of the structure. For long term monitoring, the output of a SHM system reflects the capability of a structure to continue to perform its intended function after being exposed to aging and the accumulation of damage so far. In the case of extreme events, the system can be used as a rapid screening mechanism that indicates the severity of damage and the condition of the structure. The shortcoming of SHM systems is that they are based around the notion that the damage will significantly alter the properties of the structure. However, damages are typically local phenomena's that might not affect the global response of the structure [52]. As a result, the role of inspections or targeted surveys that provide quality and accurate information about a damage and its effect on its local environment is highlighted.

Chapter 6.2, along with Chapter Inspection, noted that routine inspections are usually the tool to detect a new damage, while the conditional and extraordinary inspections usually build up from the knowledge provided by them. These means that higher level inspections are usually linked to the monitoring of a certain damage, since they result in the accumulation of information about the damage. In the same manner, surveying technologies that can be utilized continuously through the lifecycle of the asset provide means to generate quality information that can be used for the estimation of the current state of the damage. Furthermore, the data obtained with surveying technologies that are performed at periodical intervals can be compared to one another in order to assess not only the current state of the damage, but its evolution.

As different SHM systems, inspections and surveying technologies can be used to monitor a certain damage and its influence in the structure, the issue of data interoperability arises. Both SHM systems and surveying technologies usually result in digitalized data, but each one of

Page | 86





them can provide the data in one or more formats that cannot be directly linked. On the other hand, inspections are usually recorded in a report, which can take the form of a normal document or table, sometimes in physical format. Therefore, digitalization and data interoperability are important enabling technologies and lack of solutions can hinder efficient/real time modelling of damage and prediction of condition development. These issues have been tackled by WP3 and WP4 of this project, so the next chapter will focus solely on the modelling and predictive aspects.



7 Damage classification

7.1 Damage classification procedures

The development of a damage classification may be a crucial instrument for the assessment of new and existing bridges, as well as for the evaluation of maintenance strategies. Clustering and homogenization of the input data provided by inspection, testing and monitoring is indeed a great deal of effort for road operators and infrastructure managers, which have to manage a huge amount of information in order to keep assets at a desired performance level.

Hence, procedures for damage classification are needed, accounting for type, size and location of defects or other relevant issues depending on the type of structure, the actions on structure, and the risks that may potentially affect the structure in the future, such as the one following from changes in traffic loads or service life demand, and from resilience issues related to climate change and increased use. In case of bridges and tunnels, specific performance indicators (PIs) and damage indicators (Dis) can be included in the database, in order to describe the health status of the assets and accounting for damage in performance assessment and maintenance strategies. These indicators can be qualitative or quantitative based, and they can be obtained during principal inspections, through a visual examination, a non-destructive test or a temporary or permanent monitoring system.

Accordingly, a damage classification procedure consists of the following steps:

- **1. Damage detection**: damages affecting the structure under investigation are detected through inspection, testing and monitoring.
- **2. Damage characterization**: Once damage has been detected, the following information is needed and must be inserted into the database.
 - a. Level: network / system / component level at which the damage is detected.
 - b. **Location**: identification of the elements of the structure on which damage is located.
 - c. **Type**: identification of the type of damage occurring on structures
 - d. **Causes**: damage may be due to the overloading of the structure, to the aging of materials and to several damage processes. Damage causes are outlined in [2.3.1].
 - e. **Quantification**: identification of the qualitative/quantitative parameters related to the detected damage.
 - f. **Extent**: characterization of the extent of damage, which is the basis for intervention and maintenance prioritization and planning.
- 3. **Information updating**: database information must be updated over time with additional information collected through inspection plan, maintenance interventions and monitoring systems.

7.2 Procedures for including Damage Indicators in evaluating KPIs

Chapter 5 of [4] described Performance Indicators (PI) at the component, system and network levels. Furthermore, in this deliverable (D2.2), Chapter 2 provided an overview of damages and degradation mechanisms, including their basis for evaluation and their possible causes and effects. More specifically, Chapter 2.6 described Damage Indicators (DI) and PI and briefly introduced their relationship. This chapter aims to further characterize this relationship, considering Key Performance Indicators (KPIs) as well.

As mentioned in Chapter 2.6, a Damage Indicator is an observation, or a parameter derived from observations, that serves for quantitative or qualitative damage detection, damage







localization, damage parametrization or damage assessment. On the other hand, a Performance Indicator is defined as an observation, or a parameter derived from observations, that quantitatively describes property of the structure and/or of the aspect of its performance and are used to qualify fitness of the structure for its purpose during service life. While similar, their domains are vastly different. A DI simply addresses the severity of a single damage, while a PI tackles the asset in generalized manner. However, this also means that they are related. The appearance of a certain damage in a structure, while it might be considered non relevant in structural calculations in the case of minor damages, it is still part of the structure, and will therefore affect the PIs.

The purpose of this chapter is to further characterize the relationship between DIs and PIs. This will be tackled in a generalized manner, as the PIs and DIs of different projects and assets are bound to be different. Nevertheless, this abstraction serves to illustrate the relationship between both concepts, and how they affect one another. Figure 7.1 presents a diagram that expresses the relation between the different concepts.



Figure 7.1 – Relationship diagram for KPI - PI – DI

It is better to express their relationship in a top-down approach, starting from the Key Performance Indicators (KPIs). A KPI is a subset of the PIs that are critical for to determine the state of the structure. They carry the most significant information about its current performance and are used to quantify its fitness for purpose. Due to them being a subset of the PIs, the relationship between the three concepts, KPIs, PIs and DIs, is reduced to PIs and Dis..

As such, the first step is to define a hierarchical structure that represents the influence relation between different PIs and the DIs. Generally, a PI might be related to one or more DIs that are performance relevant in its domain. However, as seen in the diagram of Figure 7.1, a certain parameter can be a KPI, a PI and a DI at the same time. As an example, a PI that reflects the behaviour of a bridge would be its frequency. In this case, this PI is influenced by corrosion...) and damages (material discontinuities, ambient conditions (wind, temperature...). However, since it can be directly measured in the structure, no direct relationship to a DI is needed or, if its determined that the frequency measurement is detecting a damage, it can be set as both PI and DI. Furthermore, if the frequency carries sufficiently significant information to determine the fitness for purpose of the structure, it might be declared a KPI as well, falling in the central region of the diagram of Figure 7.1.

In the case of a PI that targets the structural integrity of the truss of a bridge, it should be directly related to the DIs that describe the presence and severity of cracks, or the plastic deformation and displacement of the elements.

In the technical domain, the selected performance measures are to follow certain properties [53]:

- Appropriateness: It should adequately reflect one or more goal or objective.
- **Comprehensible and defensible:** It should be clear, simple, and concise.





- Comprehensive: It should cover all the possible consequences.
- **Dimensionality:** It should be comparable across time and geographic regions, and it should have the required level of dimension associated with the decision-making problem.
- Measurability: It should be measured objectively.
- **Predictable:** It should be possible to use it as grounds for prediction of future levels.
- **Realistic and operational:** It should be practical, meaning that it should be measurable without excess effort or time.
- Unambiguous: It should clearly define its metric and its relationship with the consequences.

While the performance indicators are to be studied for each case, common structural performance indicators are shown below [54]:

- **Structural reliability:** Quantification of the probability of failure. It considers load models and resistance, effects that might occur over time, and the possibility of extreme hazards.
- **Cumulative probability of failure:** Quantification of the probability that the time of failure is less than a generic time interval. It is calculated from the probability density function of the time of failure.
- **Survivor function:** Complement of the cumulative probability of failure. Therefore, it reflects the probability of not failing before a generic time interval.
- **Hazard rate function:** Measure of the instantaneous failure rate. It defines the conditional probability that it will fail in a future time interval, given that it survived until the present time.
- **Structural redundancy:** Estimation of the ability of the structural system to continue carrying load after the failure of one of its components.
- **Structural robustness:** Quantification of the ability of the structural system to suffer damage because of an extreme action.
- **Structural vulnerability:** Reflects the susceptibility to some external natural or manmade action.
- **Structural risk:** Quantification of the combined effects of actions, probability of failures, and related consequences in a given context.
- **Structural resilience:** Estimates the ability of recovery against the occurrence of a hazardous event.

Once the structure that defines which DIs influences which PIs/KPIs is set, the final step is to quantify said influence. Both the structure relations and the quantification are to be evaluated in detail for each case, as the context and resources of each project are different from one another. For instance, an analytic hierarchy process can be used to quantify the importance of a certain set of DIs in a given PI [55]. The starting point of this methodology is the hierarchical/dependence structure between DIs and PIs, as mentioned previously. By knowing which DIs influence a certain PI, it is possible to compare them pair-wise to determine a rank of influence. This is done using a pair-wise comparison matrix, as is shown in Table 7.1 and Table 7.2, which determines the relative importance of different DI pairs to the PI of study. The steps to follow are:

1. To set the importance (i) of the DI in the rows with respect of the DIs in the columns. This means that the diagonal of the matrix is always filled with ones. It also means that if DI_1 is given a i_{12} over DI_2 , DI_2 has a $1/i_{12}$ over DI_1 .







- 2. To sum the weights column-wise (Σ) .
- 3. To divide the cell values between the sum of its own column (n).
- 4. Average the cells row-wise to obtain the criteria weights (ω). These weights determine the rank of importance of the DIs that influence the PI under study.

DIs related to PI _n	DI_1	DI_2	DI_3
DI1	1	i ₁₂	i ₁₃
DI ₂	1/i ₁₂	1	i ₂₃
DI ₃	1/i ₁₃	1/i ₂₃	1
Sum	Σ1	Σ2	Σ3

Table 7.1 – Example of Pair-wise comparison matrix for a given PI

DIs related to Pin	DI1	DI2	DI3	Criteria Weights
DI1	n ₁₁	n ₁₂	n ₁₃	ω_1
DI ₂	n ₂₁	n ₂₂	n ₂₃	ω2
DI ₃	n ₃₁	n ₃₂	n ₃₃	ω₃

Table 7.2 – Example of Criteria weights of DIs for a given PI

7.3 Proposal for the development of a Damage Classification Database

Throughout this deliverable, different types of data acquisition systems or methods have been mentioned, such as inspections or surveying technologies. In Chapter Monitoring and Chapter 6.5, the concept of monitoring has been described and characterized in the context of structural damage. This forms part of the Structural Health Monitoring of an infrastructure and highlights the importance of recording data through time. Damage records are an important part of the structural health of a structure, as it represents the evolution of an asset through its lifecycle. If a specific zone of the structure presents continuous damages over time, despite being maintained, it is necessary to evaluate the possibility of taking actions to ensure a permanent solution. This only represents the value of historical records, but other factors as the time needed for measuring the damage, as well as the equipment used, are important elements in decision making and cost calculations. Most importantly, these records can also be used to assess the implications of the damage for reliability and risk, including the safety of the users.

Therefore, this chapter aims to present a series of fields or attributes that should accompany a damage measurement in a database. Bear in mind that the implementation of these fields is not discussed in this chapter, as databases are part of a vast field of knowledge in computer science [56]. These fields are as follows:

- **Id:** Unique id of the item.
- **Damage parameter:** Damage feature being measured (e.g., crack width).
- Damage Indicator: Damage indicator related to the item.
- Date: Date of measurement.
- Location: Location of the measurement.
- Duration: Length of the entire measurement process.
- **Surveying Technology:** Surveying technology used to measure the damage feature.
- **Measurement:** Value of the measure.



- Units: Units of the measurement.
- Equipment: Equipment/Sensor used in the survey.
- Accuracy: Measurement accuracy. It is related to the equipment.
- Environmental conditions: Indication of the environment conditions at the time of measure (e.g., heavy rain or fog). It could be optional for technologies that are not environment-dependent.
- **Responsible:** Indication of the responsible personnel or inspector.

Nevertheless, these items present a simplified set of required parameters needed to infer knowledge from continuous monitoring of a damage. The SHM of an infrastructure often comprehends the use of a vast number of sensors and performing regular inspections. These heterogeneous sources of data are to be studied and joined together to further enrich the knowledge of the asset. Therefore, interoperability and digitalization solutions are to be taken into account, such as the use of Internet Of Things [57] and standardized data formats based around BIM [58], [59].





8 Actions evaluation

8.1 General

The damages that might appear in a given structure are directly linked to the actions that structure is bearing. Therefore, properly measuring or estimating these actions is a key factor throughout the lifecycle of the asset. In some cases, these actions cannot be directly measured in a practical way, such as the weight of the structure. However, they can be estimated through other means. For instance, the weight of the structure might be estimated from its volume and the density of a series of samples of the materials. The purpose of this chapter is to describe load models provided by standards for actions on both bridges and tunnels, including permanent actions (e.g., weight), variable actions (e.g., wind) and accidental actions (e.g., landslides) and to provide information on how to use in-situ data. In particular, Chapter 8.2 will provide an overview of Eurocode load models and describe how to consider in structural assessment the load effects that these actions generate upon the structure, describing both deterministic and probabilistic load models for each category. Chapter 8.3 will describe the methods for the quantification of actions, direct or indirect, such as using specific sensors. Finally, Chapter 8.4 will describe the use of the obtained data in previous chapters to create representative and accurate load models.

8.2 Actions on structures

Actions are a particular source of potential harm and present hazards to structures. Different actions can act on a structure, directly (direct actions), indirectly (indirect actions) or as environmental actions.

Based on their variation in time, actions on structures are classified as:

- permanent actions, which are present and constant during the entire duration of the reference period and fixed in space (e.g. self-weight, permanent equipment, floor or road finishing, etc.);
- variable actions, that are normally not present during the entire reference period and variable in time and space, e.g. imposed, traffic loads or climatic (wind, snow, thermal, etc.) loads;
- accidental actions, that have a low probability of occurrence during the reference period, but can have an important influence on structural reliability due to their magnitude (e.g. impact, explosion, fire, impact, flood, avalanches, landslides, etc);
- seismic actions.

In Table 8.1, examples of potentially relevant actions for bridges and tunnels based on their source and variation in time are given.

Actions	Category
Self-weight	direct & permanent actions
Weight of installations or road finishing etc.	direct & permanent actions
Imposed deformations due to shrinkage	indirect & permanent actions
Imposed deformations due to differential settlements	indirect & permanent actions
Other imposed deformations	indirect & permanent actions
Imposed accelerations	indirect & seismic actions
Traffic loads	direct & variable actions
Wind load	environmental & variable actions





environmental & variable actions
direct & accidental actions
direct & accidental actions
environmental & accidental actions

Table 8.1 – Examples of potentially relevant actions for bridges and tunnels (Extract from [4])

Concerning actions on bridges and tunnels, two important phenomena should be mentioned: climate change and mobility change.

Climate changes may have a significant effect on climate loads on structures (wind, snow, temperature, precipitation) used for either design of a new structure or assessment of an existing one. The characteristic values of climatic loads used in the assessment model can significantly vary due to the climate change influence. These characteristic values, according to [60], should be multiplied for scaling factors, which are greater than or equal to 1, to consider climate change effects based on the observed data series.

For a further description of the way to evaluate Factors of Changes (FC) relying on climate projections, see [4][3.3.1.3].

Moreover, road traffic in Europe has increased significantly over the last decades, both in terms of traffic volume and intensity, with consequent effect on load actions.

The mobility change may have a considerable impact on the traffic load models magnitude, configuration and corresponding relevant models included in the standardization codes, as well as on dangerous events that may occur, for instance on the risk of fire and explosion.

8.2.1 Bridges

•

Actions on bridges can be categorized as follows:

- Permanent actions
 - o Structural
 - o Non-structural
- Variable actions
 - o Snow
 - o Wind
 - o Traffic
 - o Thermal
- Accidental actions

The description of the actions listed above is given in the following paragraphs.

8.2.1.1 Permanent actions

This category includes:

- Structural actions
- Non-structural actions

Self-weight of structural elements belongs to this category, which, according to [61], should be classified as a permanent fixed action and the sum of self-weight of structural and non-structural elements should be taken into account as a single action.

* *	, *	**	
		_ *	
×	۰,		



Appendix A provides the average values of weights per unit volume. The determination of the characteristic values of the self-weight, dimensions and weights per unit volume shall be carried out in accordance with [62], clause [4.1.2].

Materials	Density
	γ [kN/m ³]
pavement of road bridges	
gussasphalt and asphaltic concrete	24,0 to 25,0
mastic asphalt	18,0 to 22,0
hot rolled asphalt	23,0
infills for bridges sand (dry) ballast, gravel (loose) hardcore crushed slag packed stone rubble puddle clay	15,0 to 16,0 ¹⁾ 15,0 to 16,0 ¹⁾ 18,5 to 19,5 13,5 to 14,5 ¹⁾ 20,5 to 21,5 18,5 to 19,5
pavement of rail bridges	
concrete protective layer	25,0
normal ballast (e.g. granite, gneiss, etc.)	20,0
basaltic ballast	26

Table 8.2 – Weight density of construction materials (Extract from [61])

Permanent actions normally have a small uncertainty related to the magnitude in comparison with other kind of loads. Uncertainties can be generally related to:

- Variability within a structural part
- Variability between different structural parts of the same structure
- Variability between various structures

In particular, according to [61], the characteristic values of the weights per unit volume of structural parts should be updated based on the available data.

8.2.1.1.1 Permanent structural actions

Permanent structural actions are determined by the evaluation of self-weight of the structural elements, which are beams, slab and supporting elements such as cable anchors. The self-weight G is determined by the following equation relation:

$$G = \int \gamma dV \qquad [8-1]$$

Vol

Where V is the volume described by the boundary of the structural part, γ is the weight density of the material. When the material can be assumed homogeneous, the previous equation can be written as:

$$G = \gamma_{av} V \qquad [8-2]$$

Where γ_{av} is the average value of weight density.

Permanent actions usually have a very small and slow variation in time around their mean or reach monotonically a limiting value.







From a probabilistic point of view, it is possible to take into account the variation of the selfweight by assuming upper and lower characteristic values (see [4.1.2] of [62]). The evaluation is based on the hypothesis that weight density and dimensions of a structural part are assumed to have Gaussian distributions [63].

Material	Mean value	Coefficient
	[kN/m ³]	of variation
Steel	77	< 0.01
Concrete		
Ordinary concrete ²⁾	24	0.04
High strength concrete	24-26 ⁴⁾	0.03
Lightweight aggregate concrete	4)	0.04-0.08
Cellular concrete	4)	0.05-0.10
Heavy concrete for special purposes	4)	0.01-0.02
Masonry	-	≈ 0.05
Timber ³⁾		
Spruce, fir (Picea)	4.4	0.10
Pine (Pinus)	5.1	0.10
Larch (Larix)	6.6	0.10
Beech (Fagus)	6.8	0.10
Oak (Quercus)	6.5	0.10

Examples for mean values μ_{γ} and coefficients of variation V_{γ} are provided in the Table 8.3.

The values here reported refer to the population of data taken from various sources, but they may be updated based on the available site-specific data (Section 8.4).

Concerning volume, it is usually possible to assume that mean values of dimensions are equal to the nominal values given on drawings, but its mean value V of the structural parts might be calculated directly from the mean values of the dimensions, as well as standard deviation from the values for the standard deviation for the dimensions.

Mean values and standard deviations for deviations of cross-section dimensions from their nominal values are given in Table 8.4.

Structure or structural member	Mean value	Standard deviation
Rolled steel		
steel profiles, area A	0.01 A _{nom}	0.04 A _{nom}
steel plates, thickness t	0.01 t _{nom}	0.02 t _{nom}
Concrete members ²⁾		
a _{nom} ≤ 1000 mm	0.003 a _{nom}	4 + 0.006 a _{nom}
$a_{nom} \ge 1000 \text{ mm}$	3 mm	10 mm
Masonry members		
unplastered	0.02 a _{nom}	0.04 a _{nom}
plastered	0.02 a _{nom}	0.02 a _{nom}
Structural timber		
sawn beam or strut	0.05 a _{nom}	2 mm
laminated beam, planed	≈ 0	1 mm

Table 8.4 – Mean values and standard deviations for deviation of cross-section dimensions from their nominal values

8.2.1.1.2 Permanent non-structural actions

Permanent non-structural actions are determined by the evaluation of self-weight of the nonstructural elements include road pavement, sidewalks, acoustic safety barriers, road safety



Table 8.3 – Mean value and coefficient of variation for weight density (Extract from [63])



barriers, parapets, finishes, water disposal system, road equipment, side walls and similar. Their value depends on the specific weights of the materials used.

The upper and lower characteristic values of the weights per unit volume of non-structural parts, such as ballast on railway bridges, or backfill on buried structures such as manholes, should be updated if the material is expected to consolidate, become saturated or otherwise change its properties during use.

Hence, the nominal thickness of the ballast on railway bridges should be specified and the upper and lower characteristic values of ballast thickness on railway bridges should be determined considering a deviation from the nominal thickness of $\pm 30\%$.

In particular, in order to determine the upper and lower characteristic values of the self-weight of waterproofing, decking and other cladding on bridges, where the variability of their thickness may be high, a deviation of the total thickness from the nominal value or other specified value should be taken into account.

According to [61], unless otherwise specified, this deviation should be considered as follows: $\pm 20\%$ if the nominal value includes a coating executed after construction and between $\pm 40\%$ and $\pm 20\%$ if this coating is not included.

For the self-weight of other non-structural elements, such as:

- handrails, safety barriers, parapets, kerbs, and other finishes of bridges;
- joints/connections;
- lightening elements.

The characteristic values should be taken as equal to the nominal values unless otherwise specified.

8.2.1.2 Variable actions

8.2.1.2.1 Snow actions

In [64] guidance on how to determine the values of loads due to snow is provided. Snow loads are usually classified as variable actions and are assumed to act vertically and to be referred as horizontal projections of the area.

Snow can be deposited on a roof in many different patterns, depending on:

- the shape of the roof;
- its thermal properties;
- the roughness of its surface;
- the amount of heat generated under the roof;
- the proximity of nearby buildings;
- the surrounding terrain;
- the local meteorological climate, in particular its windiness, temperature variations, and likelihood of precipitation (either as rain or as snow).

According to [64], snow load can be determined by the following relation:

$$S = \mu_i C_e C_t s_k \qquad [8-3]$$

Where:

- μ_i is the snow load shape coefficient;
- s_k is the characteristic value of snow load on the ground;
- C_e is the exposure coefficient;
- C_t is the thermal coefficient.





Concerning the characteristic value of snow load on the ground s_k , it should be determined in accordance with [62]([4.1.2]): in case of a small variability, one single value G_k may be used, otherwise an upper value $G_{k,sup}$ and a lower $G_{k,inf}$ value shall be used. In this respect, the National Annex specifies the characteristic values to be used, though different characteristic values might be allowed to cover unusual local conditions. For the European ground snow load map, see Annex C of [64].

Alternatively, according to [63], the following relation might be used:

$$S_r = S_q r k^{h/h_r}$$
 [8-4]

Where:

- S_g is the snow load on ground at the weather station and it is time dependent;
- r is a conversion factor of snow load on ground to snow load on roofs;
- *h* is the altitute of the building site;
- h_r is a reference altitude (= 300 m);
- k is 1.25 for coastal regions and 1.5 for inland mountainous regions.

Hence, [63] suggests determining the characteristics of the ground snow load S_g based on observations from weather stations.

[63] also includes the description of a probabilistic model for S_q , which is presented by:

- A probability distribution function for the total duration *T* of the load;
- A probability distribution function for the maximum load S_{amax} within one year.

The distribution functions $F_{S_{gmax}}$ in case of maritime and continental climates are gamma distributions whose parameters should be based on local observations.

Approaches provided by the references above are summarized here.

8.2.1.2.1.1 Eurocode approach

[64] recommends increasing snow loads values in regions with possible rainfalls on the snow and consecutive melting and freezing, especially in cases where snow and ice can block the drainage system of the roof.

The characteristic value of snow load on the ground s_k may be refined using an appropriate statistical analysis of long records taken in a well sheltered area near the site, given that record periods of less than 20 years are not generally suitable for the variability of maximum winter values. [64] also recommends considering the future development around a specific site of C_e . Provisions about the adjustment of the ground snow load according to return period are given in Annex D and summarized here.

 s_k definition is based on annual probability of exceedance of 0,02. Nevertheless, If the available data show that the annual maximum snow load can be assumed to follow a Gumbel probability distribution, then the relationship between the characteristic value of the snow load on the ground and the snow load on the ground for a mean recurrence interval of n years is given by the formula:

$$s_n = s_k \left\{ \frac{1 - V \frac{\sqrt{6}}{\pi} [ln(-ln(1-P_n)) + 0.57722]}{(1+2.5923 V)} \right\}$$
 [8-5]

Where:

- s_k is the characteristic snow load on the ground (with a return period of 50 years, in accordance with [62]
- s_n is the ground snow load with a return period of *n* years;







- P_n is the annual probability of exceedance [equivalent to approximately 1/n, where n is the corresponding recurrence interval (years)];
- *V* is the coefficient of variation of annual maximum snow load.

This expression is represented graphically in Figure 8.1, where *X* is the return period in years and $Y = s_n/s_k$.



Figure 8.1 – Adjustment of the snow load on the ground in relation to changes of the return period (Extract from [64])

8.2.1.2.1.2 JCSS approach

The characteristics of the ground snow load S_g should be determined based on observations from weather stations, which can provide data in terms of water-equivalents of snow or depths of snow. In the first case the values can be used directly to determine the ground snow load, while in the second case the data on snow depth must be converted to snow load by the following relation:

$$S_g = d \gamma(d) \qquad [8-6]$$

Where:

- *d* is the snow depth;
- $\gamma(d)$ is the average weight density of the snow, that derives from

$$\gamma(d) = \frac{\lambda\gamma(\infty)}{d} ln \left\{ 1 + \frac{\gamma(0)}{\gamma(\infty)} \left[exp\left(\frac{d}{\lambda}\right) - 1 \right] \right\}$$
 [8-7]

For further information, see [63].

8.2.1.2.2 Wind actions

In [65] and [63] guidance on the determination of natural wind actions is provided.

Wind effects on buildings and structures depend on the exposure of buildings, structures and their elements to the natural wind, but also on the dynamic properties, the shape and dimensions of the building/structure. Wind actions are classified as variable fixed actions and fluctuate with time and may act directly as pressures on the external surfaces, indirectly on the internal surfaces and directly on the internal surface of open structures.





The evaluation of wind actions is based on wind velocity or velocity pressure, which are composed of a mean and a fluctuating component.

Wind mean velocity v_m can be determined from the basic wind velocity v_b , defined as:

$$v_b = c_{dir} c_{season} v_{b,0} \qquad [8-8]$$

Where

- c_{dir} is the directional factor
- c_{season} is the season factor
- $v_{b,0}$ is the fundamental value of the basic wind velocity, which is defined as the characteristic 10 minutes mean wind velocity, irrespective of wind direction and time of year, at 10 m above ground level in open country terrain with low vegetation such as grass and isolated obstacles with separations of at least 20 obstacle heights.

Wind actions on structures and structural elements should be determined taking into account both external and internal wind pressures.

The fluctuating component of the wind is represented by the turbulence intensity. For further information on the coefficients for the evaluation of wind actions, see [65].

Parameter	Subject Reference
peak velocity pressure q _p	
basic wind velocity v _b	4.2 (2)P
reference height ze	Section 7
terrain category	Table 4.1
characteristic peak velocity pressure qp	4.5 (1)
turbulence intensity Iv	4.4
mean wind velocity v _m	4.3.1
orography coefficient $c_o(z)$	4.3.3
roughness coefficient c _r (z)	4.3.2
Wind pressures, e.g. for cladding, fixings and structural parts	
external pressure coefficient cpe	Section 7
internal pressure coefficient c _{pi}	Section 7
net pressure coefficient c _{p,net}	Section 7
external wind pressure: we=qp cpe	5.2 (1)
internal wind pressure: w _i =q _p c _{pi}	5.2 (2)
Wind forces on structures, e.g. for overall wind effects	
structural factor: c _s c _d	6
wind force $F_{\rm W}$ calculated from force coefficients	5.3 (2)
wind force F _W calculated from pressure coefficients	5.3 (3)

Table 8.5 – Calculation procedures for the determination of wind actions (Extract from [65])

Wind actions on bridges, in particular, produce forces in the x, y and z directions, as shown in Figure 8.2.







Figure 8.2 – Directions of wind actions on bridges (Extract from [65])

In particular, the wind force in x-direction may be obtained with the following equation:

$$F_W = \frac{1}{2} \rho v_b^2 C A_{ref,x}$$
 [8-9]

Where

- v_b is the basic wind speed
- \tilde{C} is the wind load factor
- $A_{ref,x}$ is the reference area
- ρ is the density of the air

The wind force in z-direction may have significant effects only if it is of the same order as the dead load, whilst the one in y-direction is not usually taken into account.

For further information on wind actions on bridges and on combination of wind and traffic actions in case of simultaneity for road and railway bridges, see chapter 8 of [65].

The influence of dynamic effects related to wind loads, typical in the case of high-rise buildings, should be considered, since they cause oscillations in the along wind, across wind, and torsional directions. These effects are described in annex E of [65].

In [63] the correspondent probabilistic model is given, in which it is assumed that mean wind velocities, for any terrain category, height above the ground and averaging time interval have a Weibull distribution.

$$F_{\overline{U}}(x) = 1 - exp\left[-\frac{1}{2}\left(\frac{x}{\sigma}\right)^k\right]$$
 [8-10]

With k close to 2.

8.2.1.2.3 Traffic actions

8.2.1.2.3.1 ULS models

In [66] guidance on models and representative values for traffic loads on roadway bridges, footbridges and railway bridges is provided.

Loads due to road traffic consisting of cars, trucks and special vehicles, give rise to vertical and horizontal forces, static and dynamic. Vehicle traffic can differ between bridges by:

- composition (e.g. percentage of trucks and trucks types);





- density (e.g. average number of vehicles per year);
- maximum weight of vehicle weights;
- axle load;
- presence of traffic signs limiting the weight of vehicles.

Loads are applied to conventional lanes, whose location and numbering should be determined in accordance with the recommendation given here.

The width of the carriageway, *w*, must be measured between the curbs of pavement or between interior boundaries of vehicle containment systems and must not include the distance between the vehicle containment systems or the curbs of the traffic dividers, nor the widths of such vehicle containment systems.

The number of lanes is represented in the following table:

Carriageway width w	Number of notional lanes n _l	<i>Width of a</i> notional lane	Width of the remaining area
w<5.4 m	1	3 m	w-3 m
5.4 m ≤w<6 m	2	0.5 w	0
6 m ≤w	Int(w/3)	3 m	w- $3 \times n_1$

Table 8.6 – Subdivision of the carriageway (Extract from [66])

When the carriageway of a bridge deck is physically divided into two parts, separated by a central reservation, then:

- if the two parts are separated by a road containment system permanent, each part, including the emergency lanes or the platforms, are divided into conventional lanes;
- if the two parts are separated by a road containment system temporary, all carriageway, including the central reservation, are divided into conventional lanes.

The conventional lanes are arranged and numbered in order to induce the most unfavourable design conditions: the lane that gives the most unfavourable effect is numbered Lane Number 1, the one that gives the second most unfavourable effect is numbered Lane Number 2, etc (see Figure 8.3).

Legend:

W= Width of the carriageway

- w_i= Lane width
 - 1= Lane Number 1
 - 2= Lane Number 2
 - o 3= Lane Number 3
 - 4= Remaining surface







Figure 8.3 – Example of the numbering of the Lanes in the most general case (Extract from (EN-1991-2, 2005))

For each individual verification, the load models, on each notional lane, should be applied on such a length and so longitudinally located that the most adverse effect is obtained, as far as this is compatible with the conditions of application defined below for each model.

Load Models for vertical loads represent the following traffic effects:

- Load Model 1 (LM1): concentrated and uniformly distributed loads, which they cover most of the effects of truck and motor vehicle traffic.
- Load Model 2 (LM2): single axle load applied to a specification tire footprint, which covers the dynamic effect of normal traffic on structural elements of small span.
- Load Model 3 (LM3): a set of vehicle axle loads special (for example for industrial transport) that can travel on roads enabled for exceptional loads. It is to be used for global and local verifications.
- Load Model 4 (LM4): a crowd load, to be used only for checks global.

8.2.1.2.3.1.1 Load Model 1

Load Model 1 consists of two parts:

• Double-axle concentrated loads (tandem system: TS), each axle having the following weight:

$$\alpha_0 Q_k \qquad [8-11]$$

where α_{Q} are correction coefficients.

• A uniformly distributed load (UDL system), with the following weight per square meter of lane:

$$\alpha_q q_k$$
 [8-12]

where α_q are correction coefficients.





Location	Tandem system Axle loads Q _{ik} (kN)	UDL system q _{ik} (kN/m ²)
Lane Number 1	300	9
Lane Number 2	200	2.5
Lane Number 3	100	2.5
Other Lanes	0	2.5
Remaining Area	0	2.5





Figure 8.4 – Application of load model 1 (Extract from [66])

For local verifications, a tandem system is applied in the most unfavourable position. If two tandem systems are considered on adjacent lanes, these can be close together at a distance between the axles of the wheels less than 0.5 m (see Figure 8.5).







Figure 8.5 – Application of tandem systems for local verifications (Extract from [66])

8.2.1.2.3.1.2 Load Model 2

Load Model 2 consists of a single axle load $\beta_Q Q_{ak}$ with Q_{ak} equal to 400 kN, including dynamic amplification. The model 2 is considered traveling in the direction of the longitudinal axis of the bridge and should be apply in any location on the roadway. If necessary, only one wheel load of β_Q ·200 kN should be considered. The contact surfaces of the wheel, if not otherwise specified, is a rectangle of sides 35 x 60 cm.



Figure 8.6 – Application of load model 2 (Extract from [66])

8.2.1.2.3.1.3 Load Model 3 (special vehicles)

Special vehicle models must be defined and taken into account when significant.

8.2.1.2.3.1.4 Load Models 4 (crowd load)

The crowd load, if significant, must be represented by a model of load consisting of a uniformly distributed load (which includes amplification dynamic) equal to 5 kN/m².





8.2.1.2.3.1.5 Rail traffic actions

General rules are given for the calculation of the associated dynamic effects, centrifugal forces, nosing force, traction and braking forces and aerodynamic actions due to passing rail traffic.

Rail traffic actions are defined by means of load models, which are the following:

- Load Model 71 (and Load Model SW / 0 for continuous bridges) to represent normal rail traffic on the main railway lines;
- SW / 2 Load Model to represent heavy loads;
- HSLM Load Model to represent the loads of trains used for transport passengers for speeds greater than 200 km / h;
- "Unloaded train" loading model to represent the effects of an unloaded train.

For further information on traffic load on road and rail bridges, see [66].

8.2.1.2.3.1.6 Probabilistic traffic load models

The traffic load models given in codes of practice are intentionally made conservative in order to be valid for a wide range of bridge types and loading conditions, and because the marginal cost of providing additional capacity is low. Load models for bridge assessment tend to be less conservative. However, in most countries the same bridge assessment principles are applied equally to bridges carrying dense traffic with heavily loaded trucks and those carrying sparse traffic with lighter trucks. In some cases, bridges may result to be structurally deficient according to these conservative load models. A more accurate representation of the current loading conditions on the bridge considered can be obtained considering the traffic weight and volume statistics for a specific bridge site. Real traffic loads can be measured using suitable techniques, for instance WIM technologies. Calibrated and cleaned data, then, serve as a basis for the use of probabilistic or simplified methods aimed at assessing load effects/values.

For further information, see 8.4.

8.2.1.2.3.2 Fatigue models

Traffic running on bridges produces a stress spectrum which may cause fatigue. The stress spectrum depends on the geometry of the vehicles, the axle loads, the vehicle spacing, the composition of the traffic and its dynamic effects.

Concerning calculation of fatigue lives, the separate models for road and railway bridges are described in paragraph [4.6] and [6.9] of [66] respectively.

In particular, there are 5 fatigue load models: 1, 2 and 3 aim to determine the maximum and minimum stresses resulting from the possible load arrangements on the bridge of any of these models, whilst 4 and 5 are intended to be used to determine stress range spectra resulting from the passage of lorries on the bridge. Fatigue Load Model 5 is the most general model, since it uses actual traffic data.

In general, a traffic category on a bridge should be defined, for fatigue verifications, at least,

by:

- The number of slow lanes
- The number of heavy vehicles observed or estimated, per year and per slow lane.

8.2.1.2.3.2.1 Load Model 1

Fatigue Load Model 1 has the configuration of the characteristic Load model 1 described in 8.2.1.2.3.1.1, with the values of the axle loads equal to 0,7 Q_{ik} and the values of the uniformly distributed loads equal to 0,3 q_{ik} and (unless otherwise specified) 0,3 q_{rk} .







The maximum and minimum stresses ($\sigma_{FLM,max}$ and $\sigma_{FLM,min}$) should be determined from all possible load arrangements of the model on the bridge. The design value of the resulting stress range $\Delta \sigma_{FLM1} = \sigma_{FLM1,max} - \sigma_{FLM1,min}$ should be equal to or smaller than the design value of the constant amplitude fatigue limit of the applicable S-N curve:

$$\Delta \sigma_{FLM1} \gamma_{F,fat} \le \frac{\Delta \sigma_D}{\gamma_{M,fat}}$$
 [8-13]

8.2.1.2.3.2.2 Load Model 2

Fatigue Load Model 2 consists of a set of idealised lorries, called "frequent" lorries: the maximum and minimum stresses should be determined from the most severe effects of different lorries, separately considered, travelling alone along the appropriate lane.

According to [66], each "frequent lorry" is defined by:

- the number of axles and the axle spacing,
- the frequent load of each axle,
- the wheel contact areas and the transverse distance between wheels.

This information are summarized in Table 2.1 – Damage processes for reinforced concrete and steel structures

1	2	3	4
LORRY	Axle	Frequent	Wheel
SILHOUETTE	spacing	axle loads	type (see
	(m)	(kN)	Table 4.8)
	4,5	90	Α
		190	В
	4,20	80	А
	1,30	140	В
0 00		140	В
	3,20	90	А
	5,20	180	В
0 0 0 0 0 0 0	1,30	120	С
	1,30	120	С
		120	С
	3,40	90	А
	6,00	190	В
0 0 00	1,80	140	В
		140	В
	4,80	90	A
	3,60	180	В
0 0 0 00	4,40	120	С
	1,30	110	С
		110	С

Table 8.8 – Set of "frequent" lorries [66]

The maximum and minimum stresses ($\sigma_{FLM2,max}$ and $\sigma_{FLM2,min}$) should be determined from all possible load arrangements of the model on the bridge.





The design value of the resulting stress range $\Delta \sigma_{FLM2} = \sigma_{FLM2,max} - \sigma_{FLM2,min}$ should be equal to or smaller than the design value of the constant amplitude fatigue limit of the applicable S-N curve:

$$\Delta \sigma_{FLM2} \gamma_{F,fat} \le \frac{\Delta \sigma_D}{\gamma_{M,fat}}$$
 [8-14]

8.2.1.2.3.2.3 Load Model 3

This model consists of four axles, each of them having two identical wheels. The geometry is shown in Figure 8.7. The weight of each axle is equal to 120 kN, and the contact surface of each wheel is a square of side OAO m.





Figure 8.7 – Fatigue Load Model 3 [66]

8.2.1.2.3.2.4 Load Model 4

Fatigue Load Model 4 consists of sets of standard lorries which produce effects equivalent to those of typical traffic on European roads. A set of lorries appropriate to the traffic mixes predicted for the route as defined in Table 8.9 and Table 8.10 should be taken into account. This model, based on five standard lorries, simulates traffic which is deemed to produce fatigue damage equivalent to that due to actual traffic of the corresponding category.




VEHICLE TYPE		TRAFFIC TYPE				
1	2	3	4	5	6	7
			Long distance	Medium distance	Local traffic	
LORRY	Axle spacing (m)	Equivalent axle loads (kN)	Lorry percentage	Lorry percentage	Lorry percentage	Wheel type
	4,5	70	20,0	40,0	80,0	А
0 0		130				В
	4.20	70	5.0	10.0	5.0	A
	1.30	120				В
	1,00	120				В
0- 00						
	3,20	70	50,0	30,0	5,0	A
	5,20	150				В
	1,30	90				C
0-0-0-000	1,30	90				С
·		90				C
	3,40	70	15,0	15,0	5,0	A
	6,00	140				В
	1,80	90				В
00-0- 00		90				В
	4,80	70	10,0	5,0	5,0	A
	3,60	130				В
	4,40	90				C
0 0 00	1,30	80				C
		80				C

Table 8.9 – Set of equivalent lorries [66]



Table 8.10 – Definition of wheels and axles [66]

According to [66], each standard lorry is defined by:

- the number of axles and the axle spacing,
- the equivalent load of each axle
- the wheel contact areas and the transverse distances between wheels.

The stress range histogram and the corresponding number of cycles from each fluctuation in stress during the passage of individual lorries on the bridge should be the Rainflow or the Reservoir counting method. This histogram contains all stress cycles $\Delta \sigma_i$. The fatigue damage, *D*, should be determined using the following equation:







$$D_{d} = \frac{1}{5*10^{6}} \sum_{i=1}^{n} min \left[\left(\frac{\gamma_{M,fat} * \gamma_{F,fat} * \Delta \sigma_{i}}{\Delta \sigma_{D}} \right)^{3}, \left(\frac{\gamma_{M,fat} * \gamma_{F,fat} * \Delta \sigma_{i}}{\Delta \sigma_{D}} \right)^{5} \right]$$
[8-15]

The design damage should be equal to or smaller than 1.

8.2.1.2.3.2.5 Load Model 5

Fatigue Load Model 5 consists of the direct application of recorded traffic data, supplemented, if relevant, by appropriate statistical and projected extrapolations.

A stress history can be obtained by analysis using recorded representative real traffic data, multiplied by a dynamic amplification factor, which should take into account the dynamic behaviour of the bridge and depends on the expected roughness of the road surface and on any dynamic amplification already included in the records.

8.2.1.2.3.2.6 Probabilistic fatigue load models

According to [67], the current fatigue load models applied in Europe are based on traffic load measurements in 1986 and the most frequently used fatigue load model is unable to represent the fatigue action effects of today's European traffic. Hence, [67] proposes a new fatigue load model whose parameters can be calibrated on WIM data, in order to have a significant improvement in accuracy compared to the existing models. This approach in summarized in 8.4.

8.2.1.2.4 Thermal actions

In [68] and [63] guidance on the determination of thermal actions is provided.

Thermal actions can be classified as variable and indirect actions. The temperature distribution within an individual structural element may be split into the following four essential constituent components, as illustrated in Figure 8.8:

- a) A uniform temperature component, ΔT_u ;
- b) A linearly varying temperature difference component about the z-z axis, ΔT_{MY} ;
- c) A linearly varying temperature difference component about the y-y axis, ΔT_{MZ} ;
- d) A non-linear temperature difference component, ΔT_E . This results in a system of selfequilibrated stresses which produce no net load effect on the element.



Figure 8.8 – Diagrammatic representation of constituent components of a temperature profile (Extract from [68])

The strains and therefore any resulting stresses, are dependent on the geometry and boundary conditions of the element being considered and on the physical properties of the





material used. When materials with different coefficients of linear expansion are used compositely the thermal effect should be taken into account.

Representative values of thermal actions should be assessed by the uniform temperature component and the temperature difference components. In particular, the vertical temperature difference component should generally include the non-linear component.

8.2.1.2.4.1 Uniform temperature component

The uniform temperature component depends on the minimum and maximum temperature which a bridge achieves. This results in a range of uniform temperature changes which, in an unrestrained structure would result in a change in element length, though the effects related to restraint of associated expansion or contraction due to the type of construction, to friction at roller or sliding bearings, to non-linear geometric effects ad to the interaction between the track and in the bridge (in case of railway bridges) due to the variation of the temperature of the deck and of the rails, which may induce supplementary horizontal forces on the bearings, should be taken into account if relevant.

8.2.1.2.4.2 Range of uniform bridge temperature component

The values of minimum and maximum uniform bridge temperature components for restraining forces shall be derived from the minimum (T_{min}) and maximum (T_{max}) shade air temperatures.

The correlation between minimum/maximum shade air temperature (T_{min}/T_{max}) and minimum/maximum uniform bridge temperature component $(T_{e,min}/T_{e,max})$ is given in figure 6.1 of [68].

The characteristic value of the interval of maximum contraction of the uniform temperature component of the bridge is:

$$\Delta T_{N,con} = T_0 - T_{e,min} \qquad [8-16]$$

while that of maximum expansion is:

$$\Delta T_{N,exp} = T_{e,max} - T_0 \qquad [8-17]$$

where T_0 is the initial bridge temperature at the time that the structure is restrained.

8.2.1.2.4.3 Vertical component

According to the [68], the effect of vertical temperature differences should be considered by using an equivalent linear temperature difference component with $\Delta T_{M,heat}$ and $\Delta T_{M,cool}$, whose values should be applied between the top and the bottom of the bridge deck.

Specific values for each country should be used. Some examples are given in Table 8.11.





Type of Deck	Top warmer than bottom	Bottom warmer than top	
Type of Beek	$\Delta T_{M,heat}$ (°C)	$\Delta T_{M,cool}$ (°C)	
Type 1: Steel deck	18	13	
Type 2: Composite deck	15	18	
Type 3: Concrete deck: - concrete box girder - concrete beam - concrete slab	10 15 15	5 8 8	
NOTE 1: The values given in the table represent upper bound values of the linearly varying temperature difference component for representative sample of bridge geometries.			
NOTE 2: The values given in the table are based on a depth of surfacing of 50 mm for road and railway bridges. For other depths of surfacing these values should be multiplied by the factor k_{sur} . Recommended values for the factor k_{sur} is given in Table 6.2.			

Table 8.11 - Recommended values of linear temperature difference component for different types of
bridge decks for road, foot and railway bridges (Extract from [68])

8.2.1.2.4.4 Horizontal component

The temperature difference component generally needs only to be considered in the vertical direction. In particular cases, however (for example when the orientation or configuration of the bridge results in one side being more highly exposed to sunlight than the other side), a horizontal temperature difference component should be considered.

8.2.1.2.4.5 Probabilistic model

Probabilistic models for thermal actions are described in section 2.14 of [69], in which the basic variables that may significantly influence effects of thermal actions on structures are listed:

- Climatic agents
 - Shade air temperature (daily and seasonal changes)
 - Solar radiation (direct and diffused)
 - Wind speed (influenced by regional wind climate and orography)
- Operating process temperatures
 - o Inner environment of the structure
- Characteristics of construction works
 - Space orientation of the structure
 - Shape of the structure, dimensions and cross-sectional geometry
 - o Joints of the structure, types of materials and used colours
 - Structural system
 - Thermal properties of materials
 - $\circ\,$ Initial temperature at which the structure is restrained, properties of atmosphere and terrain
- Geographical location of site
- Properties of atmosphere and terrain.
 - Emissivity of the atmosphere and terrain, location near some water source





According to [69], development of temperatures in a structure may be obtained either theoretically by means of the heat conduction method or experimentally by measuring the temperatures in many points of relevant structure with adequate frequency. A reliable theoretical prediction of the temperature fields in a structure is a quite difficult problem because it needs the integration of the Fourier heat conduction equation on a generally complex shaped domain under appropriate initial conditions, and with non-linear and time-dependent boundary conditions.

Given that the thermal exchange between the structure and its environment is influenced by three basic factors which may appear simultaneously (solar radiation and irradiation, thermal convection, and conduction), for bridges the evaluation of the vertical difference temperature component should take into account the measurements of shade air temperatures and effects of solar radiation.

The statistical characteristics of the model of temperature difference component may be assessed on the basis of the extreme value probabilistic distribution. For further information, see [69].

8.2.1.3 Accidental actions

In [70] and [63] guidance on the determination of accidental actions is provided. This category includes actions due to impact and explosions, though, in case of bridges, explosions might be disregarded.

In practice, the occurrence and consequences of accidental actions can be associated with a certain risk level, which may lead to additional measures, when necessary. It should be noted, however, that zero risk level is impracticable and that in most cases it is necessary to accept a certain level of risk. For bridge structures such emergency measures may involve the closure of the road or rail service within a specific limited period.

Hence, the risk analysis of the accidental actions to be considered depend upon the measures to be taken for preventing or reducing the severity of an accidental action, the probability of occurrence and the consequence of failure of the identified action, the public perception, and the level of acceptable risk.

8.2.1.3.1 Impact

[70] defines accidental actions due to the following events:

- impact from road vehicles (excluding collisions on lightweight structures);
- impact from forklift trucks;
- impact from trains (excluding collisions on lightweight structures);
- impact from ships;
- the hard landing of helicopters on roofs.

According to [63], Impact is an interaction phenomenon between the object and the structure. In the case of bridges, the actions due to impact and the mitigating measures provided should consider, amongst other things, the type of traffic on and under the bridge and the consequences of the impact, therefore actions due to impact from road vehicles, trains and ships must be taken into account.

In general, actions due to impact may be determined either by a dynamic analysis or by an equivalent static force, where the forces at the interface of the impacting object and the structure depend on their interaction.

The parameters to be considered are:

• impact velocity of the impacting object;





- mass distribution;
- deformation behaviour;
- damping characteristics (of both the impacting object and the structure);
- angle of impact;
- construction and movement of the impacting object after collision.

The basic model for impact loading consists of:

- Potentially colliding objects;
- The occurrence of a human or mechanical failure that may lead to a deviation of the intended course;
- The course of the object after the initial failure;
- The mechanical impact between object and structure.



Figure 8.9 – Probabilistic collision model (Extract from [63])

Using the assumption of rigid structure and colliding object modelled as an elastic single degree of freedom system, with equivalent stiffness k and mass m, the maximum possible resulting interaction can be calculated as:

$$F_c = v_c \sqrt{km} \qquad [8-18]$$

Where v_c is the object velocity at impact and the duration of this load is:

$$\Delta t = m v_c / F_c \qquad [8-19]$$

8.2.1.3.1.1 Impact from road vehicles

Accidental actions caused by road vehicles include the ones due to impact on supporting substructures and to impact on superstructures.

[70] provides examples for the value of equivalent static force to use.





Category of traffic	Force F _{dx} ^{a)} [kN]	Force F _{dy} ^{a)} [kN]
Motorways and country national and main roads	1 000	500
Country roads in rural area	750	375
Roads in urban area	500	250
Courtyards and parking garages with access to: 50 25 - Cars 50 25 - Lorries ^{b)} 150 75		
 a) x = direction of normal travel, y = perpendicular to the direction of normal travel. b) The term "lorry" refers to vehicles with maximum gross weight greater than 3,5 tonnes. 		

Table 8.12 – Indicative equivalent static design forces due to vehicular impact on members supporting structures over or adjacent to roadways (Extract from [70])

Category of traffic	Equivalent static design force F _{dx} ^{a)} [kN]
Motorways and country national and main roads	500
Country roads in rural area	375
Roads in urban area	250
Courtyards and parking garages	75
a) x = direction of normal travel.	

 Table 8.13 – Indicative equivalent static design forces due to impact on superstructures (Extract from [70])

For further information on impact loads on kerbs and parapets and impact for traffic on bridges, see [66], while accidental actions caused by road vehicles on bridges also carrying rail traffic are described in UIC leaflet 777.1R.

Concerning the condition of impact from road vehicles, [70] suggests using the following recommendations:

- for impact from lorries the collision force F may be applied at any height h between 0,5 m to 1,5 m above the level of the carriageway or higher where certain types of protective barriers are provided. The recommended application area is a = 0,5 m (height) by 1,50 m (width) or the member width, whichever is the smaller.
- for impact from cars the collision force F may be applied at h = 0,50 m above the level of the carriageway. The recommended application area is a = 0,25 m (height) by 1,50 m (width) or the member width, whichever is the smaller.





Key

- a Is the height of the recommended force application area. Ranges from 0,25 m (0,50 m (lorries)
- *h* Is the location of the resulting collision force *F*, i.e. the height above the level carriageway. Ranges from 0,50 m (cars) to 1,50 m (lorries)
- Is the centre of the lane



Figure 8.10 – Collision force on supporting substructures near traffic lanes for bridges and supporting structures for buildings (Extract from [70])

For $h_0 \le h \le h_1$, these values may be multiplied by a reduction factor r_F , where



Figure 8.11 – Recommended value of the factor r_F for vehicular collision forces on horizontal structural members above roadways, depending on the clearance height h (Extract from [70])

- h Is the physical clearance between the road surface and the underside of the bridge deck;
- *h*₀ Is the minimum height of clearance between the road surface and the underside of the bridge deck below which an impact on the superstructure need to be taken into account. The recommended value of *h*₀ is 5,0 m;
- *h*₁ Is the value of the clearance between the road surface and the underside of the bridge deck. For values of *h*₁ and above, the impact force F need not be





considered. The recommended value of h1 is 6,0 m (+ allowances for future resurfacing, vertical sag curve and deflection of bridge);

• b is the difference in height between h_1 and h_0 , i.e. $b = h_1 - h_0$. The recommended value for b is 1,0 m. A reduction factor for F is allowed for values of b between 0 and 1 m, i.e. between h_0 and h_1 .

On the underside surfaces of bridge decks the same impact loads as above with an upward inclination may have to be taken into account: the conditions of impact may be given in the National Annex. The recommended value of upward inclination is 10°, see Figure 8.12.



- X Direction of traffic
- h Height of the bridge from the road surface measured to either the soffit or the structural members



Figure 8.12 – Impact loads on members of the superstructure (Extract from [70])

[70] also considers accidental actions caused by derailed rail traffic under or adjacent to structures, using the following classification of structures.

Class A	Structures that span across or near to the operational railway that are either permanently occupied or serve as a temporary gathering place for people or consist of more than one storey.
Class B	Massive structures that span across or near the operational railway such as bridges carrying vehicular traffic or single storey buildings that are not permanently occupied or do not serve as a temporary gathering place for people.

Table 8.14 – Classes of structures subject to impact from derailed railway traffic (Extract from [70])

For further information, see [70].

8.2.1.3.1.2 Impact from ships

Accidental actions due to collisions from ships should consider:

- the type of waterway,
- the flood conditions,
- the type and draught of vessels and their impact behaviour, and
- the type of the structures and their energy dissipation characteristics.

According to [70], the action due to impact should be represented by two mutually exclusive forces:

- a frontal force (in the direction of the normal travel, usually perpendicular to the longitudinal axis of the superstructure (deck))
- a lateral force with a component F_R parallel to F_{dx} .





The following equation may be used for the evaluation of the impact force due to friction F_R acting simultaneously with the lateral impact force:

$$F_R = \mu F_{dy} \qquad [8-20]$$

Where μ is the friction coefficient.

The position and area over which the impact force is applied depend upon the geometry of the structure and the size and geometry (e.g. with or without bulb) of the vessel, the vessel draught and trim, and tidal variations.



Figure 8.13 – Indicative impact area for ship impact. (Extract from [70])

Moreover, the forces on a superstructure should be determined by taking account of the height of the structure and the type of ship to be expected. In general, the force on the superstructure of the bridge will be limited by the yield strength of the ships' superstructure.

8.2.1.3.2 Explosions

An explosion is defined as a rapid chemical reaction of dust, gas, or vapour in air, which results in high temperatures and high overpressures. Explosion pressures propagate as pressure waves. The pressure generated by an internal explosion depends primarily on the type of dust, gas or vapour, the percentage of dust, gas or vapour in the air and the uniformity of the dust, gas or vapour air mixture, the ignition source, the presence of obstacles in the enclosure, the size, the shape, and the strength of the enclosure in which the explosion occurs, and the amount of venting or pressure release that may be available.

Explosion pressures on structural members should be determined considering reactions transmitted to the structural members by non-structural members. The explosive pressure should be assumed to act effectively simultaneously on all the bounding surfaces of the enclosure in which the explosion occurs.

For further information, see [70].

8.2.1.4 Seismic actions

In [71] guidance on design and construction of buildings and civil engineering works in seismic regions is provided. In cases of low seismicity, reduced or simplified seismic design procedures for certain types or categories of structures may be used.

Seismic actions evaluation depends on the "basic seismic hazard" of the construction site, since they are defined in terms of the maximum horizontal acceleration expected in free field





conditions on a rigid reference site with a horizontal topographic surface, as well as in terms of the ordinates of the elastic response spectrum in acceleration corresponding to it, with reference to predetermined probabilities of exceedance in the period of reference.

Hence, the first step of the evaluation consists of the identification of the category of soil to which the site belongs.

Ground type	Description of stratigraphic profile	Parameters		
		v _{s,30} (m/s)	N _{SPT} (blows/30em)	c _u (kPa)
А	Rock or other rock-like geological formation, including at most 5 m of weaker material at the surface.	> 800	_	-
В	Deposits of very dense sand, gravel, or very stiff clay, at least several tens of metres in thickness, characterised by a gradual increase of mechanical properties with depth.	360 - 800	> 50	> 250
С	Deep deposits of dense or medium- dense sand, gravel or stiff clay with thickness from several tens to many hundreds of metres.	180 - 360	15 - 50	70 - 250
D	Deposits of loose-to-medium cohesionless soil (with or without some soft cohesive layers), or of predominantly soft-to-firm cohesive soil.	< 180	< 15	< 70
E	A soil profile consisting of a surface alluvium layer with v_s values of type C or D and thickness varying between about 5 m and 20 m, underlain by stiffer material with $v_s > 800$ m/s.			
S ₁	Deposits consisting, or containing a layer at least 10 m thick, of soft clays/silts with a high plasticity index (P1 > 40) and high water content	< 100 (indicative)	-	10 - 20
<i>S</i> ₂	Deposits of liquefiable soils, of sensitive clays, or any other soil profile not included in types $A - E$ or S_1			

Figure 8.14 – Ground types (Extract from [71])

The earthquake motion at a given point on the surface is represented by an elastic ground acceleration response spectrum, henceforth called an "elastic response spectrum".

The elastic response spectrum in acceleration is expressed by a spectral form referred to a conventional damping of 5%, multiplied by the value of the maximum horizontal acceleration on a rigid horizontal reference site.

The defined spectrum can be used for structures with fundamental period less than or equal to 4.0 s. For structures with higher fundamental periods, it must be defined by specific analyses, otherwise seismic actions should be described by accelerograms.

The elastic response spectrum of the horizontal component is defined by the following expressions:

$$0 \le T < T_B \qquad \qquad S_e(T) = a_g \cdot S \cdot \left(1 + \frac{T}{T_B} \cdot (\eta \cdot 2.5 - 1)\right)$$

Page | 119





$$T_B \le T < T_C \qquad \qquad S_e(T) = a_g \cdot S \cdot \eta \cdot 2.5$$

$$T_C \le T < T_D$$
 $S_e(T) = a_g \cdot S \cdot \eta \cdot 2.5 \left(\frac{T_C}{T}\right)$

$$T_D \le T$$
 $S_e(T) = a_g \cdot S \cdot \eta \cdot 2.5\left(\frac{T_C T_D}{T^2}\right)$

[8-21]

where

- $S_e(T)$ is the elastic response spectrum;
- *T* is the vibration period of a linear single-degree-of-freedom system;
- a_q is the design ground acceleration on type A ground;
- T_B is the lower limit of the period of the constant spectral acceleration branch;
- T_C is the upper linlit of the period of the constant spectral acceleration branch;
- *T_D* is the value defining the beginning of the constant displacement response range of the spectruln;
- *S* is the soil factor;
- η is the damping correction factor with a reference value of $\eta = 1$ for 5 % viscous damping.

8.2.2 Tunnels

There are four main types of actions acting on tunnels:

- Permanent actions
 - Structural
 - o Non-structural
- Soil actions
- Hydraulic actions
- Landslide actions

According to [72], the most common loads distribution model is shown in the following figure.







Figure 8.15 – Loads on the segmental ring (Extract from [72])

Where:

- pw1 is the vertical water pressure;
- pe1 is the vertical earth pressure;
- pr is the ground reaction due to vertical load;
- pg is the self-weight;
- q_{w1} is the lateral water pressure at tunnel top;
- q_{w2} is the lateral water pressure at tunnel bottom;
- q_{e1} is the lateral earth pressure at tunnel top;
- q_{e2} is the lateral earth pressure at tunnel bottom;
- q_r is the ground reaction due to lateral deformation;
- H_w is the depth of groundwater level;
- H is the buried depth of tunnel;
- k is the coefficient of ground reaction;
- δ is the lateral deformation;
- P₀ is the surcharge;
- t is the thickness of segment;
- R_o is the outer radius of segmental ring;
- R_c is the calculative radius of segmental ring, i.e. radius of segmental ring used in calculations)





The description of the actions listed above is given in the following paragraphs.

8.2.2.1 Permanent actions

According to [73], dead load is vertical load acting along the centroid of the cross-section of tunnel and is calculated in accordance with the following equation:

$$p_g = \frac{W}{2 \pi R_c}$$
 [8-22]

Where:

- W is the weight of lining per meter in longitudinal direction;
- R_c is radius of centroid of the linings.

8.2.2.2 Soil actions

The load distribution model consists of applying uniform vertical ground pressures, linearly varying lateral earth pressures, and a triangularly distributed horizontal ground reaction. Their effect can be analysed using elastic equations, beam-spring models, finite element methods (FEM) and discrete element methods (DEM). The elastic equation method is a simple method for calculating lining internal forces for circular tunnels.

The magnitude and distribution of the upward ground reaction due to vertical load P_r are assumed to be the same as the downward earth in the upper part. The lateral earth pressures are assumed, instead, to increase uniformly with depth. The values of lateral earth pressure can be evaluated multiplying the vertical loads by a coefficient of lateral earth pressure. [73] suggested that the value of the coefficient of lateral earth pressure (λ) to be used in the design calculation should be between the value of the coefficient of lateral earth pressure at rest (K₀) and the value of the coefficient of lateral active earth pressure (K_a). It was proposed by JSCE (2006) that the value of K₀ should be regarded as λ when the horizontal ground reaction is difficult to be obtained, and that the value of K_a or a reduction of K₀ should be taken as half of the sum of K₀ and K_a:

$$\lambda = \frac{1}{2} \times (K_0 + K_a)$$
 [8-23]

 K_a is the coefficient of lateral active earth pressure. K_a can be calculated using equation proposed by Rankine:

$$K_a = \tan^2 \left(\frac{\pi}{4} - \frac{\emptyset}{2}\right)$$
 [8-24]

To estimate vertical loads, it can be referred to the concept of Terzaghi's solid [74], formulations and [75], used respectively for the soil and rock conditions. In soft ground condition, the vertical ground pressure should be equal to the overburden pressure, if the designed tunnel is a shallow tunnel (H < 2D). If it is a deep tunnel (H \geq 3D), the reduced earth pressure can be adopted in accordance with Terzaghi's formula considering the wet unit weight for soil above groundwater table and the submerged unit weight for one below groundwater table.

$$\sigma_{v} = \frac{B_{1}(\gamma - c/B_{1})}{k_{0} \tan \phi} \left(1 - e^{-K_{0} \tan \phi H/B_{1}} \right) + p_{0} e^{-K_{0} \tan \phi H/B_{1}}$$
[8-25]

On the other hand, in rock ground condition, the rock load (P) based on Unal concept is calculated as [75]:

Page | 122





$$P = \frac{100 - RMR}{100} \gamma D$$
 [8-26]

where RMR is Bieniawski's Rock Mass Rating, and *D* is the diameter of the tunnel.

The ground reaction q_r correspond to the lateral tunnel deformation multiplied by a coefficient of ground reaction k. The coefficient of ground reaction k is an empirical coefficient and its value ranges widely in different kinds of soil. Its value will be suggested in the geotechnical or design report for a specific project and can be estimated with Galerkin Method:

$$K_n = \frac{E}{R_t(1+\gamma)}$$
; $K_t = \frac{1}{3}K_n$ [8-27]

where *E* is rock mass deformation modulus, v is the ground Poisson coefficient, and R_t is the radius of tunnel.

8.2.2.3 Hydraulic actions

The water pressure acting on lining is represented by a hydrostatic pressure. The resultant of water pressure acting on tunnel linings is the buoyancy. If the resultant of the vertical earth pressure at crown and the dead load is greater than the buoyancy the difference between them acts as the vertical earth pressure at bottom (subgrade reaction). If the buoyancy is greater than the resultant of the vertical earth pressure at crown and the dead load, the tunnel would float. According to [73], water pressure load can be estimated as:







Figure 8.16 – Hydrostatic pressure (Extract from [73])

$$P_{w1} = \gamma_w H_w \text{ [At tunnel crown]}$$
 [8-28]

$$P_{w1} = P_{w1} + \gamma_w R_c (1 - \cos\theta)$$
 [At the bottom] [8-29]



Figure 8.17 – Load scheme (Extract from [73])

8.2.2.4 Landslide actions

In the planning of underground infrastructures, geological, geomorphological, as landslides, and hydrogeological conditions of the area interested by the works should be carefully considered to evaluate, in term of deformation response of the rock mass, the interaction between the excavation and the existing structures at the surface. Landslide location and morphology, active or quiescent in nature, must be evaluated in detail to define the tunnels' alignment. Landslides are particularly sensitive to changes of the stress-strain in the rock mass induced by underground excavation (both in conventional and mechanized system) and to the tunnel size and shape, slope inclination and shape of the sliding surface; the excavation of a tunnel can lead to a rapid evolution of landslides. For this reason, tunnel design must include, especially in a landslide area, an extensive investigations and geotechnical monitoring, as well as the development of numerical models (FEM or FDM analyses) to forecast the evolution of landslides during tunnel excavation in the short and long term (e.g., an investigation campaign with boreholes fairly deepened down the future tunnel invert needs to be planned, in addition to the studies of the area and to the bibliographic research).

The delimitation of landslides bodies in the depth of the slopes allows to define a conceptual "kinematic model" to be used to evaluate slope stability before, during and after tunnel construction. The excavation, if driven inside a landslide or close to a major slipping surface, may increase the natural movement of the landslide or even reactivate quiescent bodies. The presence of a landslide, even inactive, on a slope where a tunnel must be driven, must be carefully considered also for the potential damages to the tunnel itself that may be caused by the induced ground movements. At the ground level these movements usually start when the





face of the excavation is at a distance between one and two times the tunnel overburden and stop when the face is passed through at the same distance. Different tunnel locations – in plan and elevation - should be analysed, to find the geometric solution compatible with the slope and landslide stability. In this stage, empirical approaches and numerical evaluation in "closed form" are generally used, in order to obtain "qualitative" responses, to be compared in term of cost and time and environmental sustainability.

8.3 Actions evaluation

As per Chapter 8.2 structures are subjected to different kinds of actions. These actions are sometimes specific to their type, such as wind loads on bridges. However, they can also be consistent among several, such as self-weight. This chapter aims to explain the different methodologies that can quantify the action that affect a given structure. This evaluation is directly involved with the design of the asset, as it must consider the possible actions that will influence it in a certain location or across seasons. To better summarize this information, the chapter presents a table that collects the different methods or sensors used to evaluate a given action. These items will then be briefly described following the separation between bridges and tunnels.

Туре	Actions	Evaluation	
Permanent	Structural	Load cells	
Permanent	Non-structural	Weight estimation via density	
Variable	Thermal	Infrared thermography Thermocouples Thermistors	
Variable	Snow	Rods Snow pillows/scales Passive gamma radiation	
Variable	Wind	Anemometer	
Variable	Traffic	Weight in motion Cameras	
Accidental	Accidental	Simulation	
Seismic	Seismic	Accelerometers	

Table 8.15 – Most common sensors for bridge action evaluation

Туре	Actions	Evaluation
Permanent	Structural	Load cells
Permanent	Non-structural	Weight estimation via density.
Variable	Soil actions	Strain gauges Pressure cells Weight estimation via soil stratification
Variable	Hydraulic actions	Strain gauges Pressure cells Piezometers
Accidental	Landslide	Slope clinometers

Table 8.16 – Most common sensors for tunnel action evaluation

8.3.1 Bridges

Permanent loads: For the measurement of structural and non-structural loads the most common sensor in structures are the load cells. A load cell is a device that converts a force

Page | 125





into a measurable output (e.g., voltage). They can be used for directly measuring the permanent load G (i.e., self-weight) of the desired component such as the deck, the pavement, the non-structural elements on the superstructure of the bridge, etc.

There are different types of load cells according to their measurement technique. Most common load cells are the strain-gauge load cells. These sensors calculate the weight of the bodies on them by measuring the intensity of the electrical resistance variation in the strain gauges, which is proportional to the intensity of the applied force on them. Due to their measurement technique, these load cells can be glued to any beam, but it is recommended to placed them onto the ones that are critical in the structure e.g., arches, stringers, etc.

Other load cell types are the hydraulic load cells that measure the weight as a change in pressure in the internal filling fluid; the pneumatic load cells that operate on the force-balance principle or the inductive load cells whose output is proportional to the displacement of a ferromagnetic core. All these load cells must be placed between the zone where the desired permanent load is being applied and a base. In order to have the most possible accuracy, these two surfaces must be parallel, and the load has to be applied vertically on the loading surface.

Besides, there are a relationship between the permanent loads G and mass of the structural elements $m_{element}$ using the gravity constant g.

$$G = m_{element} \cdot g \qquad [8-30]$$

Therefore, if a measuring of the whole structure is required, weight estimations using the volume of all the elements and the density of their materials must be done. The volume can be calculated by using the dimensions of the structure by looking at historical documents such as the drawings of the structure, or by measured them with instruments like terrestrial laser scanners or callipers. On the other hand, density of the constituent material can be measured directly using some instruments such as solid density meters or by testing an extracted sample from the bridge (measuring both, its volume and mass).

Thermal loads: Thermal loads, as it is introduced in Chapter 8.2.1.2.4, are provoked by temperature differentials in the structure. These differentials can be measured through electric devices such as thermocouples, thermistors or RTD (resistance temperature detectors). Thermocouples are devices based on the Seebeck effect [76], which relates the open circuit voltage ΔV between semiconductors and the temperature differential ΔT between the hot and the cold point of the circuit, using the Seebeck coefficient α_S .

$$\Delta V = \alpha_S \cdot \Delta T \qquad [8-31]$$

Thermal actions can derive from uniform or non-uniform temperature differentials along the cross-section height. Therefore, for non-uniform temperature differentials, ends of the thermocouple must be placed both on deck intrados and extrados, so that non-uniform variations can be measured.

On the other hand, thermistors and RTD are temperature-sensitive resistors, so the temperature differential can be estimated by the difference in the resistance value.

Nowadays, it was verified that thermographic methodologies are also suitable for assessing the temperature effects in different structures [77]. Furthermore, it is possible to estimate the temperature variations in a structure by observing the effects that this differential provokes in some structural responses such as the dynamic behaviour. Thus, accelerometers can be used for measuring the temperature differentials in the bridge.

Snow loads: Recently, the European Cooperation in Science & Technology and some partners [78] conducts a survey about the different measurement techniques related with snow







in Europe. In the document, four different measurements were asked about: snow depth, snow presence, depth of snowfall and water equivalent of snow cover. The snow depth and the depth of snowfall quantify the height of the snow in the ground in any time or after a specific period of time (usually 24 hours) respectively, so they can be measured by using rods and rulers. On the other hand, the snow presence identifies the existence of snow in a defined area, which can be checked by cameras or satellite images. Although relevant, these measurements do not provide information related to snow loads. Nevertheless, the water equivalent of snow cover can be expressed as a pressure that can be easily extrapolated to a snow load. In order to measure this data, manual and automatic approaches can be undertaken. In manual measurements, a worker must conduct a snow course. The snow course consists in take from 5 to 10 SWE sample locations spaced 30 meters apart. A SWE sample consists in filling a metallic tube with known dimensions to calculate the density of the snow in it. After weighting the sample, the density of the snow is calculated using the following equation.

$$\rho_{snow} = \frac{m_{sample}}{V_{tube}}$$
 [8-32]

Subsequently, the water equivalent of snow cover is calculated using the equation below.

$$SWE = \sum_{samples} L \cdot \rho_{snow}$$
 [8-33]

Where ρ_{snow} , m_{sample} and V_{tube} are referred to the density of the snow, the mass of the SWE sample and the volume of the metallic tube. And *L* is the height of the snow of the snow sample.

Concerning the automatic approaches, the water equivalent of snow cover can be directly measured using weighing mechanisms such as snow pillows or scales or passive gamma radiation instruments. The snow pillows are synthetic rubber or stainless-steel bladders filled with an antifreeze fluid that measure the pressure of the snow on them by measuring the hydro-static pressure in the bladder. On the other hand, the snow scales performed the weight measurement using an electronic load cell. Both, snow pillows and scales, must be placed on a ground surface where the snow is going to be accumulated. Finally, the passive gamma radiation instruments are able to calculate the water equivalent of snow cover by measuring the attenuation of the gamma radiation by the water in the snowpack and the soil. However, this last technique is only adequate for measuring low SWE values. These instruments consist of a gamma radiation source separated from a gamma radiation detector. The source must be positioned on the ground surface, while the detector must be positioned above the snow surface pointing towards the gamma radiation source.

Wind loads: In Chapter 8.2.1.2.2 is commented that the wind loads are calculated by considering the direction and the speed of wind. Anemometers are the most common sensors for measuring these variables. There are a lot of different anemometer types, but the windmill and the ultrasonic anemometers are the most used ones in bridges. The first one calculates the wind speed by measuring its angular velocity when its propellers are pushed by the air. On the other hand, ultrasonic anemometers have an ultrasonic emitter and receiver separated one from each other. The emitter sends ultrasonic pulses to the receiver, but these may be disrupted by the air if this is moving quickly. Hence the velocity of the wind is calculated by measuring this disruption. Besides, there are some measuring techniques for measuring the wind pressure that some authors are studying recently. Concerning the ones that could be implemented in civil engineering structures like bridges, in [79] researchers develop a capacitive pressure sensor based on a membrane of dielectric elastomer that correctly identifies the wind pressure when it is placed on a surface.

Traffic loads: In order to quantify the traffic loads a lot of bridges have installed Weight in Motion (WIM) systems. There are several types of WIM systems such as the pavement WIM systems, the Bridge WIM (BWIM) systems and the Dynamic On-Board WIM (OBW) systems.

Page | 127





All of them are thoroughly described in WP2. Pavement WIM systems are embedded in the roadway or under/on the bridge deck, they can be installed before or on the bridge. On the other hand, the BWIM systems comprehend different sensors attached to the soffit the bridge to measure its stresses and strains so results can be obtained by inverse modelling. Finally, the OBW systems have its sensors installed on the vehicles instead of the infrastructure. All these WIM systems measure the frequency and the weight of the vehicles that cross it. Besides, they are also able to measure the tire impact forces, the strain forces and the velocity of the vehicles. All this information is studied for establishing the load models that correctly describe the traffic loads of a bridge.

Nevertheless, some authors like [80] said that most Weigh in Motion technologies do not work correctly in congested traffic. Hence, they have proposed a methodology to estimate the traffic loads using digital cameras and image analysis. This methodology consists of taking several pictures of a zone of the bridge using the digital camera in order to capture data of the vehicles crossing the structure e.g., dimensions and type of vehicles, speed and volume of traffic, etc. Moreover, the authors have defined a statistical correlation between these data (dimensions and type of vehicles, allowing the calculation of traffic loads.

Accidental loads: An accidental load has a complicated nature. The load will depend on the mass, velocity and direction of the vehicle but also on the mechanical properties of the crashed body and the structural element where it collapses. For instance, the magnitude of the load will be different if the collision is plastic or elastic. Besides, accidental loads have a low frequency of occurrence by definition. Therefore, it is complicated to install some sensors in the structure for quantifying them. In fact, the current standards provide several tables that shows the most appropriate load to consider in each specific collision, as it is showed in the Chapter 8.2.1.3. Hence, a recommendable approach for assessing the damage that an accidental load provokes is by modelling the structure and simulate its response for every accidental action showed in the tables of the standards.

Seismic loads: According to the standards [81], the seismic loads have to be represented by an elastic ground acceleration response spectrum, but they can also be modelled using accelerograms and spatial models. Therefore, the main surveying instrument for quantifying them in bridges are the accelerometers, which have been thoroughly described in WP2. Some authors [82] claim that it is also interesting to capture the dynamic behaviour of the structure while measuring the seismic loads. In addition, they also comment that these measurements can be improved by including some sensors such as strain and displacement gauges, weather stations, etc. for recording all the data related to the structural vibrations under strong motions and the data that can interferes in the measuring of the accelerometers. This information is useful for the structural health monitoring and the study of the combined effects of the soil-structure interactions.

8.3.2 Tunnels

Permanent loads: Similar to bridges, the most common sensor used for quantifying the permanent loads of the structural and non-structural elements in the tunnels are the load cells, particularly the strain-gauge load cells. Besides, if a quantification of the permanent load of the whole structure is required, it can also be estimated by measuring the dimensions of all the structural elements and check the density of their materials by looking at the standards or by measuring it using devices such as solid density meters. The permanent loads can be calculated from the measured or calculated weights by multiplying them by the standard earth gravity.

Soil actions: The soil actions act on the walls of the tunnels pressing them. Therefore, it is possible to quantifying them by measuring the deformations that they provoke in these walls using strain-gauge pressure plates. As it was commented before, the strain gauges calculate the applied forces into them using the proportionality between them and the intensity of the





electrical resistance variation in the gauges. Some authors [83] have implemented a system consisting in pressure plates and fiber-optic to quantify these loads.

Besides, the soil actions can be also estimated by knowing the density of the soil above the tunnel. However, soils have stratums made of different materials. Therefore, it is necessary to study the soil stratification in order to identify them and, consequently, their depths and densities. These stratums can be identified by looking at the bibliographical data (e.g., historical documents), by studying the soil stratification in the laboratory or by performing insitu surveys. Concerning the surveys, the cone penetration test is the most used technique to identify and classify the stratums of a ground. It consists of vertically penetrate a cone into the ground at a constant low velocity to measure the resistance of the soil to this penetration. It should be noted that it is necessary to carry out a preliminary study to identify the survey positions. These are defined according to the depth of the ground and the continuity of its stratums. On the other hand, some non-destructive measuring systems can be also used for measuring the soil stratification. The most commonly used is the Ground Penetrating Radar (GPR), which is thoroughly described in WP2.

Hydraulic actions: Similar to the soil actions, these can be measured by looking at the strains that they provoke in the walls of the studied tunnel. Hence, pressure plates explained before can be also implemented for measuring hydraulic actions.

Nevertheless, these actions can also be quantified by measuring the pore water pressures. These pressures can be obtained by piezometers. A piezometer is a sensor capable of measuring these water pore pressures and the groundwater level. It can also be used to measure the pressure of the water in aquifers. Piezometers have to be driven into the ground, so a borehole has to be drilled beforehand. The position of these boreholes must be previously defined in a preliminary study. There are different types of piezometers depending on their measuring principle such as the hydraulic piezometers, pneumatic piezometers or the open standpipe piezometers.

Landslide actions: In order to monitoring the land slope and predict its possible collapses, a measurement of its displacements and accelerations is required. Besides, these displacements can also be useful to determine the mass of displaced debris if a landslide occurs. Therefore, the most common sensors for measuring them are the slope clinometers. The clinometers are sensors that measure the slope gradient and the shear displacements of the ground over time. They are also used for measuring this gradient during activities such as tunnelling, excavation, etc. Inclinometers are the servo clinometers and the MEMS clinometers. Last ones have been thoroughly explained in WP2. There are several papers [84] that describe different monitoring of landslides systems that are based on these types of sensors.





8.4 Use of data for actions modelling [SAC]



Load models in 8.2 describe the temporal, spatial and directional properties of the actions across the structure. This section aims to provide information on how to include available data in the models described above, identifying parameters and variables which may be updated in order to have a reasonably accurate representation of the current loads. In this respect, each section will include the following scheme:

- Surveying technology
- Site-specific data provided by each technology
- Parameters for actions evaluation which can be updated based on the site-specific data.

8.4.1 Bridges

8.4.1.1 Permanent actions

Concerning the evaluation of permanent actions, paragraph 8.3.1 lists the following technologies:

- Load cells
- Terrestrial laser scanner
- Calliper
- Solid density meters
- In-situ/lab tests on samples.

In this section the measured quantity and the associated parameter to be updated (element weight, volume, weight density) related to each technology is given. Load cells allow to evaluate the weight of desired element, whilst terrestrial laser scanners and callipers provide volume measurements. Weight density can be updated based on data coming from solid density meters or in situ/lab tests on samples.







Figure 8.18 – Use of data for permanent actions

8.4.1.2 Variable actions

8.4.1.2.1 Snow actions

Concerning the evaluation of snow actions, paragraph 8.3.1 lists the following technologies:

- Rods
- Rulers
- Cameras
- Satellite images
- Snow pillows/scales
- Passive gamma radiations

In this section the measured quantity and the associated parameter to be updated is given. Cameras and satellite images allow to detect the presence of snow, but cannot provide a direct input for the evaluation of snow loads, whilst rods and rulers can be used for the quantification of snow depth and therefore of the value of the snow load on ground S_a .



Figure 8.19 – Use of data for snow actions-rods/rulers

Snow pillows/scales and passive gamma radiations are used for the evaluation of the waterequivalent of snow cover, which is directly related to S_a .



Figure 8.20 - Use of data for snow actions-snow pillows/scales, passive gamma radiations





As outlined in 8.2, indeed, the characteristics of the ground snow load S_g can be determined based on observations from weather stations, either water-equivalents of snow or depths of snow, given that in the first case the values can be used directly to determine the ground snow load, while in the second case the data on snow depth must be converted to snow load by the relation:

$$S_g = d \gamma(d) \qquad [8-34]$$

8.4.1.2.2 Wind actions

Wind actions evaluation depends on the location and on the availability and quality of meteorological data, the type of terrain, etc. Mean wind velocities may vary over the year: if data are available, it is possible to consider updated values of $v_{b,0}$ in the calculation.

Anemometers allow the measurement of the direction and the speed of wind, based on which wind actions can be evaluated.



Figure 8.21 – Use of data for wind actions

8.4.1.2.3 Traffic loads

8.4.1.2.3.1 Ultimate loads

Load models included in [66] do not describe actual loads: they have been selected and calibrated so that their effects (with dynamic amplification included where indicated) represent the effects of the actual traffic in the year 2000 in European countries.

Hence, structure-specific traffic load models may be developed in order to have consistent data with real traffic loads, which can be measured using suitable techniques, for instance WIM technologies. Calibrated and cleaned data, then, serve as a basis for the use of probabilistic or simplified methods aimed at assessing load effects/values. The general scheme is provided in figure 8.21 and includes:

- WIM DATA: as the first step of the procedure, the measured WIM data must be calibrated and cleaned, as WIM systems may have some measurements errors and are thus sensitive to environmental conditions such as the variation of temperature, which can cause the subsequent process to produce distorted results.
- **SIMULATION MODEL**: key random variables representing the traffic environments of the structure are identified and fitted to probability distribution models. Then, using the developed simulation model and appropriate parameter values of the key random variables, a Monte Carlo simulation is performed to generate artificial WIM data for the time period of interest.
- TRAFFIC LOAD EFFECT SIMULATION: load effects/values are calculated.





Figure 8.22 – General scheme (Extract from [4])

For further information on how to build site-specific traffic load models, see [4].

8.4.1.2.3.2 Fatigue loads

The fatigue load models (FLMs) described in 8.2 were calibrated using traffic load measurements, in terms of traffic composition, heavy vehicle construction, intervehicle distances, number of traffic jams, legislation and number of heavy vehicles. But, as these measurements have changed a lot since 1986, FLMs should be updated based on actual traffic data. In this respect, traffic data measured with a WIM station may provide a good basis for the calibration of fatigue load models for road bridges. An example of how WIM data can be used for the creation of a WIM database to be directly used for fatigue verifications instead of using random simulations with vehicle weight distributions-based measurements is provided in [67] and summarized here.

- The first step of the procedure consists of the creation a WIM database, obtained from a WIM measurement station. WIM databases should allow an accurate representation of the fatigue loads.
- Because of the differences in traffic composition between different countries, sitespecific load models can be derived, in which the stress ranges $\Delta\sigma$ defined by Eurocodes are used to derive one equivalent range:

$$\Delta \sigma_{FLM}^{*} = \begin{cases} \Delta \sigma_{l} & \text{if } n_{l} = 1 \\ \left[\Delta \sigma_{l}^{5} + \left(1 + a \frac{L}{m} \right) \sum_{l=1}^{n_{l}} \Delta \sigma_{l}^{5} \right]^{1/5} & \text{if } n_{l} > 1 \end{cases}$$
 [8-35]

Where:

- n_l is the number of cycles encountered by crossing the vehicle over the influence line;



Page | 133



- $\Delta \sigma_l$ is the largest stress range encountered by crossing the vehicle over the influence line;
- *L* is the span (in case of a single span bridge, otherwise it is the average of the adjacent spans)
- *a* is a factor depending on the density of traffic, whose value must be calibrated with WIM data.

For further information, see [67].

8.4.1.2.4 Thermal actions

The magnitude of the thermal effects depends on local climatic conditions, together with the orientation of the structure, its overall mass, finishes (e.g. cladding in buildings).

Moreover, an additional contribution assessed on the basis of weather station data or data from temperature sensors can be considered, which consists of a linear variation between extrados and intrados of the element under investigation.

Thermal differentials can be measured through electric devices such as thermocouples, thermistors or RTD (resistance temperature detectors). Based on the available data, the minimum and maximum uniform bridge temperature components $T_{e.min}$ and $T_{e.max}$ might be determined.

As outlined in 8.3.1, thermal actions can derive from uniform or non-uniform temperature differentials along the cross-section height. Therefore, for non-uniform temperature differentials, ends of the thermocouple must be placed both on deck intrados and extrados, so that non-uniform variations can be measured.



Figure 8.23 – Use of data for thermal actions

8.4.1.3 Accidental actions

The strategies recommended by [70] for accidental design situations are illustrated in Figure 8.24.

In the perspective of prevention, accidental actions should be evaluated considering the most unfavourable scenario within the analysis carried out for the specific structure. This implies that the value of the action cannot in general be modified on the basis of the available data, unless exceptional situations which are not foreseen by regulations are taken into account.







Figure 8.24 – Strategies for accidental design situations (Extract from [70])

As stated in 8.3.1, a recommendable approach for assessing the damage that an accidental load provokes is by modelling the structure and simulate its response for every accidental action showed in the tables of the standards.

8.4.1.4 Seismic actions

National territories are subdivided into seismic zones, depending on the local hazard. Hence, by definition, the hazard within each zone is assumed to be constant. Nevertheless, seismic maps might be updated either based on the occurrence of unusual seismic events or depending on the acceleration histories recorded by seismic stations in the surrounding area of the site under investigation.

Hence, the main surveying instrument for quantifying them in bridges may be the accelerometers, since they allow the evaluation of the dynamic behaviour of the structure while measuring the accelerations provoked by seismic events.

8.4.2 Tunnels

8.4.2.1 Permanent actions

The considerations made in paragraph 8.4.1.1 can also be applied to tunnels.

8.4.2.2 Soil actions

As outlined in 8.3.2, soil actions act on the linings of the tunnels pressing them. Hence, they may be quantified by measuring the deformations that they provoke using strain-gauge pressure plates.



Figure 8.25 – Use of data for soil actions-pressure plates

Moreover, they depend on the stratigraphic characteristics of the soil/rock crossed by the tunnel, whose geotechnical and mechanical parameters can be updated on the basis of the





results of investigation campaigns. Therefore, it is necessary to study the soil stratification in order to identify them and, consequently, their depths and densities.



Figure 8.26 - Use of data for soil actions-GPR, investigation campaigns bibliographical data

8.4.2.3 Hydraulic actions

Hydraulic actions can be measured by looking at the strains that they provoke in the linings of the tunnel under investigation.



Figure 8.27 - Use of data for hydraulic actions-pressure plates

Nevertheless, water pressure load can be estimated based on H_w , which is the depth of groundwater level. The theoretical value of H_w is provided by the tunnel design documents, but it may be updated based on a piezometric survey campaign in the proximity of the section to be analysed.



Figure 8.28 – Use of data for hydraulic actions-piezometers

8.4.2.4 Landslide actions

To consider the effect derived from the activation of an accidental load, as a landslide, it's necessary to develop a numerical FEM or FDM model of the slope and the tunnel which consider the results of a preliminary and extensive structural and geotechnical investigation. Geotechnical investigation may include campaign with piezometer to measure the heigh of groundwater table at specific point, with slope inclinometers to detect zones of movement and establishing whether a movement is constant, accelerating, or responding to remedial measure.





9 Model updating based on diagnostic load testing

9.1 General

Structures and structural components are designed in such a way to ensure that the performance level remains above the one required for the structural safety and serviceability during their entire life cycle. To allow effective and efficient life cycle management, though, the following factors should be taken into account:

- deterioration mechanism(s), acting or suspected;
- aging of materials;
- environmental conditions;
- increase/decrease in loads over time;
- change in standards performance requirements.

Hence, the factors which may influence the deterioration and progress of deterioration are related to changes in loads acting on structures and to the properties of the structures themselves. Figure 9.1 shows the probability density function of resistance and load respectively: assuming that variables have a Gaussian distribution resistance, it should be noted that resistance is characterized by a reduced mean value and by an increased standard deviation, because of the higher level of uncertainty, whilst loads increase over time without substantial changes in standard deviation values.



Figure 9.1 – Changes over time of resistance and loads (Extract from [85])

Accordingly, it is necessary to determine the deterioration mechanisms (see chapter 2), present deterioration levels and deterioration rates of materials and/or structural performance using appropriate models based on information obtained during from inspection, testing and monitoring activities (chapter 3 and 4), from the design and construction records, information upon previous interventions and the environmental conditions. Model updating allows to evaluate the condition of structures and, therefore, to make a prognosis of current performance.

In particular, loads acting on structures may vary over time due to several reasons:

- Standards update (to account for the increase of actions);
- change of the intended use;
- change of traffic volume;
- change of permanent structural actions.

Resistance variation, instead, is related to changes in the properties of materials, sections, structural components and structural systems (e.g., aging, delamination, spalling, etc).

The variation in the resistance of structures can be determined from the analysis of design documentation, drawings, results of visual inspections and testing on elements and materials,





and from information obtained through a load test, either a proof load test or a diagnostic load test. Design documentation and drawings can be used for a first estimate of the resistance, whilst inspections and testing are the basis for the updating over time of parameters related to the structural behaviour. Load test, instead, can prove that the structure's load bearing capacity is actually adequate, i.e. it can reveal its hidden capacity, and it always enables a check if the response of the structure is according to the objective of its design.



Figure 9.2 – Model updating methods

The difference between proof loading and diagnostic load testing is that proof loading focuses on improving the analytical assessment of an existing structure revealing the potential hidden safety reserve, giving important information about the effective structural performance and its actual level of safety. It is defined as the assessment of a structure under a given limit state by applying an equivalent load. Diagnostic load testing, instead, focuses on confirming the response of the structure against the service loads. Results from the model and the observed behaviour of the structure under a certain percentage of the design live loads are examined in order to verify the suitability of the design/analytical model. It can be static and/or dynamic.

For further information on proof loading and diagnostic load test, see [4].

In the following paragraphs reference is made only to diagnostic load testing.

Load tests may be performed in order to update the failure probability of a structure and, therefore, the reliability index β , in order to implement a maintenance and interventions prioritization plan.

In particular, the actual β should be compared to the target reliability levels proposed for the assessment of existing structures, based on which actions to be taken may vary:

- *β*₀ level below which the existing structure is considered unreliable and should be upgraded;
- *β_{up}* level indicating an optimum upgrade strategy while upgrading of existing structures.

A detailed framework for the data-informed safety assessment is thoroughly described in [10].

9.2 Model updating

Model updating techniques are widely used in the SHM field and allow for the calibration of the system properties (parameters in numerical models), based on actually observed behaviour of the system of interest. Structural FE model updating is used for design verification and validation, to obtain improved predictions of structural response quantities, or simply to identify unknown system characteristics [86]. It is assumed that localized structural damage results in a local reduction of stiffness, thus, updating stiffness parameters of the FE model in several damage states provides a (non-destructive) means to thoroughly and accurately investigate the condition of the structure [87].

Both data acquired from dynamic analysis and static diagnostics load tests are used for updating purposes, as they provide detailed information regarding the global and local





behaviour of the structure of interest and can be measured in an operational state of the structure.

Starting from an initial numerical model of the structure, diagnostic load tests are conducted in order to verify the compliance of the numerical model with the real structural response by examining, on a comparative basis, the results from the model and the observed behaviour of the structure under a certain percentage of the design live loads. In the static diagnostic load test, the load is applied following application systems and is placed in each intermediate and final phase long enough for the measurements to stabilize. The structural behaviour is controlled by selected measurement parameters (flexures, deformations, etc.), which must be constantly monitored and evaluated. Test should be considered successful when the structure carries the applied load without signs of distress and therefore fulfils the code requirements.

The dynamic test, instead, is performed to evaluate the response of the structure to the real dynamic load. In this test the dynamic amplification factor is evaluated, that is the coefficient that amplifies the structural response when the load acting from the static to the dynamic mode is changed. For further information on static and dynamic diagnostic load tests, see [4]. In this specific case, the loads used in this test are equal to those used in the static test (heavy trucks or some type of railway vehicle, depending on the type of the bridge). The loading protocol comprises a set of vehicles equal to the number of lanes driving from one end of the bridge to the other side by side at a certain speed. Usually, a few speed levels are used which are determined according to the speeds that can appear in real traffic.

In this section, a framework for the model updating related to diagnostic load testing is described, starting from the identification of the state of the structures up to the experimental validation of the model.







Figure 9.3 – Model updating framework





The framework consists of the following steps:

- 1. **OBJECTIVES DEFINITION**: includes the definition of the elements of the structure to be tested (whole structure/ selected spans), the parameters to be measured and the type of test (static/dynamic) to be performed.
- 2. **MODEL INITIALIZATION**: information, e.g., from design documents, drawings, inspection and intervention reports, is collected and used for a simplified modelling, which can be based either on beam elements or on shell elements.
- 3. **TEST PREPARATION AND EXECUTION**: based on the objectives identified in the first step, the instrumentation plan is developed. In particular, the following information is required:
 - a. Number and type of sensors
 - b. Sensors layout
 - c. *Test execution mode* In the case of static load tests, load weight, load positions have to be defined. In the case of dynamic load tests in which moving loads are applied, the test duration, vehicle speed, load path must be specified too. Otherwise, the environmental noise must be specified.
 - d. Load cases to be considered (symmetric/ asymmetric).
 - e. Load test report- summary of the information regarding the load test procedure and results.
- 4. **DATA ANALYSIS**: systematic errors compensation, due to sensors positioning error and/or to environmental influences, and extraction of synthetic parameters.
- 5. **MODEL UPDATING**: model is updated with the actual positions and values of the loads. At the model level, a cross-correlation should also be made between data from the tests. During this phase an experienced engineer must compare field data to numerical and analytical data but also identify which parameters should be adjusted to yield an accurate field-validated model (best-fitting), e.g., flexural and torsional stiffness (both in terms of sections and materials).
- 6. STRUCTURAL ASSESSMENT AND FINAL LOAD TEST REPORT: structural assessment, in order to evaluate the reliability index and eventually to assess potential deterioration processes which may be occurring, and summary of the results in a final report, providing information on the load test procedure, the model calibration method and the health status condition.

9.3 Methods of calibration of models: Deterministic and probabilistic models

There are several methods for model calibration. According to [16], it is possible to distinguish between:

• Deterministic models: the basic idea is to calibrate a model trying to minimize an objective function (objective function) capable of expressing the residuals (algebraic differences) between numerical data and experimental data. For the process of updating the FE model, modal data are typically used, in particular natural frequencies or eigenvalues. Given a set of measured modal data of the real system, the update problem is that the model realistically approximates the mass and stiffness matrices that will produce modal data as close to real systems as possible.

Deterministic models can be divided into *direct* (one step) or *parametric* (multi-step).

a) Direct deterministic methods: non-iterative methods in which the solution is reached by applying the variation of the parameters directly on the matrices that describe the system. Direct methods are the fastest and easiest ones, but sometimes the solution found has no physical meaning. To obtain valid and





physically compatible results, it is necessary to start from a very detailed finite element model, as close as possible to reality and minimized errors.

b) **Parametric deterministic methods**: iterative methods which modify the values of the parameters until convergence is reached.

Parametric methods were introduced to correct the weaknesses of direct methods. There are three main problems when comparing measured data and the corresponding numerical estimates: the first one is that the natural frequencies and the experimental and theoretical modal forms are physically relevant in the same way; for this purpose it is useful to use MAC (Media Access Control).

The second problem concerns the scaling of modal shapes. Once calculated eigenvectors are usually normalized to mass. The modal forms do not all intervene with the same scale, therefore each one is independently scaled; it is, then, possible to use a modal scaling factor (MSF):

$$MSF = \frac{\Phi_i^T \Phi_{mi}}{\Phi_{mi}^T \Phi_{mi}}$$
[9-1]

Moreover, in the uncommon case damping is taken into account in the finite element model, the eigen solutions of experimental origin can be obtained in complex form.

- **Probabilistic models**: The parameters to be updated and are not treated neither with direct formulation nor with multi-step techniques. During the model calibration procedure, errors are taken into account. The most widely applied approach to merging models with monitoring data is using a Bayesian approach, which leverages Bayesian statistics to treat parameters as random variables, thus assigning probability distributions to them. The Bayesian update leads to updated probability distributions of the parameters for use in the updated numerical model. When the analysis is complete, a model is obtained in which each of the parameters has a probability density function and an error relative to its evaluation.
- 9.3.1 Direct deterministic methods

Method of Lagrange multipliers:

The Lagrange multiplier method is a simple tool to minimize an objective function in the presence of well-defined constraints and boundary conditions on the independent variables. In this method, one set of parameters is considered fixed and immutable, while other sets of parameters are updated separately until an objective function is minimized.

An example is that of Berman and Nagy [88], who minimized the following objective function:

$$J_M = \left\| M_A^{-1/2} (M - M_A) M_A^{-1/2} \right\| + \sum_{i=1}^m \sum_{j=1}^m \lambda_{ij} (\phi^T M \phi - 1)$$
 [9-2]

Where:

- *M* are the updated masses;
- M_A are the masses extracted from the mass matrix of the model;





• λ_{ij} are the Lagrange multipliers used to force the orthogonality of the vectors with respect to the updated masses.

The results of the minimization procedure led to the following expression for the updated masses:

$$M = M_A + M_A \phi m_a^{-1} (1 - \phi^T M_A \phi) (\phi^T M_A \phi) \phi^T M_A$$
 [9-3]

 $\Phi \in \mathbb{R}^{P \times M}$ is an incomplete modal matrix because M<P and P is the order of the updated numerical model. After the calculation of *M*, the matrix *K* can also be calibrated by minimizing an additional objective function:

$$J_{k} = \left\| M_{A}^{1/2} (K - K_{A}) M^{1/2} \right\| + \sum_{i=1}^{m} \sum_{j=1}^{m} \lambda_{K_{ij}} (K\phi - M\phi\Lambda)_{ij} + \sum_{i=1}^{m} \sum_{j=1}^{m} \lambda_{O_{ij}} (\phi^{T} K\phi - \Lambda)_{ij} + \sum_{i=1}^{m} \sum_{j=1}^{m} (K - K^{T})_{ij}$$
[9-4]

Where Λ represents the spectral matrix. The stiffness updating equation can be expressed as:

$$K = K_A + (\Delta - \Delta^T)$$
[9-5]

Where Δ is obtained with the following equation:

$$\Delta = \frac{1}{2} M \Phi (\Phi^T K_A \Phi + \Lambda) \Phi^T M - K_A \Phi \Phi^T M \qquad [9-6]$$

In this case the Lagrange multipliers are used to force the respect of the equations of motion, in conditions of orthogonality, and the symmetry of the stiffness matrix.

"Matrix Mixing" approach:

This method assumes that modal forms are orthonormal and can be used only if modal forms for each degree of freedom are given.

$$\begin{split} \Phi^{T} M \Phi &= 1; \\ M &= \Phi^{T^{-1}} \Phi^{-1}; \\ M^{-1} &= \Phi \Phi^{T} = \sum_{i=1}^{m} \Phi_{i} \Phi_{i}^{T} \end{split}$$

[9-7]

The same approach can be applied to the stiffness matrix:

$$\begin{split} \Phi^{T} K \Phi &= \Lambda; \\ \Phi^{T^{-1}} (\Phi K \Phi) \Phi^{-1} &= \Phi^{T^{-1}} \Lambda \Phi^{-1}; \\ K &= \Phi^{T^{-1}} \Lambda \Phi^{-1}; \\ K^{-1} &= \Phi \Lambda^{-1} \Phi^{T} = \sum_{i=1}^{m} \frac{1}{\omega_{i}^{2}} \Phi_{i} \Phi_{i}^{T} \end{split}$$

[9-8]

The matrix-mixing approach uses modes for which some experimental data are missing.



Page | 143



$$M^{-1} = \sum_{i=1}^{m} \phi_{T_i} \phi_{T_i}^{T} + \sum_{i=m+1}^{p} \phi_{A_i} \phi_{A_i}^{T};$$

$$K^{-1} = \sum_{i=1}^{m} \frac{(\phi_{T_i} \phi_{T_i}^{T})}{\omega_{T_i}^2} + \sum_{i=m+1}^{p} \frac{(\phi_{A_i} \phi_{A_i}^{T})}{\omega_{A_i}^2}$$

[9-9]

Where subscripts A and T indicate numerical data and experimental data respectively.

If *m* is the number of measured modal shapes and *p* is the number of degrees of freedom, the issue with assembling the mass and stiffness matrices from the experimental data is that the number *m* of components of the eigenvectors of experimental origin is usually way smaller than the order *p* needed by the model. The eigenvectors extracted from the numerical and experimental analyses ϕ_{A_i} and ϕ_{T_i} , must have dimension *p*.

Structural matrices assembled with criteria m < p self-solutions are incomplete.

The matrix-mixing method generally provides matrices of complete mass and stiffness, but which have little consideration for physical connectivity.

Eigenstructure Assignment approach

The assumption of this method is that equations of motion are expressed as first or second order ordinary differential equations.

Stiffness and damping matrices are updated, while the mass matrix of the model does not vary.

The equation of motion in terms of displacement is:

$$M\ddot{x} + C\dot{x} + Kx = B_0 u$$
 [9-10]

Where:

- *M*, *C*, *K* are positive definite matrices;
- *u* is the input of the vector of control forces;
- *B*₀ is the matrix that correctly distributes the exciting forces on the displacement degrees of freedom.

It is hardly ever possible to measure all displacement variables. The measured vector containing displacement and velocity information is called *y*:

$$y = D_0 x + D_1 \dot{x}$$
 [9-11]

 D_0 and D_1 are the matrices governing displacements and velocities in the state space. They depend on the excitation positions and on the positions and type of measurements. In the updating procedure the matrices B_0 , D_1 , D_0 are initially imposed. One sets u = Gy where G is the gain matrix, on which the calibration depends. The equation of motion is:

$$M\ddot{x} + C\dot{x} + Kx = B_0(D_0Gx + D_1G\dot{x}); M\ddot{x} + (C - B_0GD_1)\dot{x} + (K - B_0GD_0)x = 0$$
 [9-12]

The gain matrix G induces perturbations on the damping and stiffness matrices by updating them in such a way that, once the real components of G are identified, the model is able to






reproduce the eigen solutions of experimental origin. The updated stiffness and damping matrices are:

$$K = K_A + B_0 G D_0$$

$$C = C_A + B_0 G D_1$$
[9-13]

The matrix B_0 can be chosen arbitrarily, C_0 and C_1 are chosen so that the sum $C_1 \Phi \Lambda + C_0 \Phi$ is not singular and is therefore invertible. The matrices Φ and Λ contain incomplete experimental eigenvectors and eigenvalues. Only m < p eigenvectors of experimental origin are needed.

9.3.2 Parametric deterministic methods

Methods based on penalty functions

In general, penalization functions are defined by considering an additional term to the objective function, which penalizes the violation of the constraints. Penalization functions are generally non-linear functions of parameters that need to take into account boundary conditions and constraints. Methods based on penalty functions are based on the use of an iterative procedure that, due to the nature of the cost function, requires the evaluation of the congruence, compatibility and significance of the modal model at each iteration. If the variation of the parameters in successive iterations is small the convergence of the solution has been reached.

Approach based on Sensitivity Analysis

The relationship between the calculated measurable quantities (natural frequencies, modal shapes or displacements) and the parameters of the model for which the correction is sought, is generally non-linear. The method is based on the linearization of this relationship and uses a Taylor series expansion truncated after the linear term:

$$\varepsilon_x = x_m - x_A(\theta) \approx r_i - G_i(\theta - \theta_i);$$

$$r_i = x_m - x_{Ai}$$

$$G_i = \left[\frac{\partial x_j}{\partial \theta_k}\right]_{\theta = \theta_i}$$

Where:

- x_m is the vector of the experimental results (frequencies and modal forms, etc.);
- x_{Ai} is the vector of the results estimated numerically at iteration *i*;
- r_i represents the residual between x_m and x_{Ai} .
- *G_i* is the sensitivity matrix;
- θ_i is the vector of the parameters of the model at iteration i^{th} .

Natural frequencies and modal shapes can be used in the updating procedure. Modal shapes often contain large measurement errors, and individual vector components can be up to 20% uncorrect. Conversely, natural frequencies can be accurately measured close to 1% and can be confidently used. In many practical cases, the number of unknown parameters exceeds the number of measured data. The system is under-determined and there are infinite groups of parameters able of satisfying the equations, in accordance with the Rouché-Capelli



[9-14]

Page | 145



Theorem. In this case, the choice must take into account the physical significance of the parameters found.

Minimum variance methods

These methods can be considered as based on penalty functions in which the weighting matrices change in a particular way from one iteration to another. This approach uses Bayesian statistics to manage a large amount of data. Both the measured data and the estimates of the initial parameters are affected by errors that can be expressed in terms of variance matrices.

Methods that use frequency domain data

In these methods the Frequency Response Function (FRF) is optimized through a penalty function that directly involves the FRF itself. FRF data can be used directly without extracting natural frequencies and modal shapes. In these methods the damping, in particular the proportional damping, is modeled in the finite element model.

9.3.3 Probabilistic methods

Probabilistic estimators

Through the Maximurn Likelhood Estimation (MLE) it is possible to determine the best estimate of the parameters θ of the model $f(\theta)$ starting from the observed measurements y and the likelihood function $L(y|\theta)$, which provides the probability of observation of the measurements given the parameters. The best parameter estimate provided by the MLE satisfies the following equation:

$$\theta_{mle}(y) = \frac{\arg\max}{\theta} L(y|\theta)$$
 [9-15]

The value of θ_{mle} defined in [9-15] can be determined with the least squares method by minimizing the error function

$$S = [y - f(\theta)] \times [y - f(\theta)]^T$$
[9-16]

In the determination of parameters θ it is possible to introduce a priori information in the data analysis with the use of an a priori distribution $g(\theta)$. The posterior distribution $p(\theta)$ is therefore

$$p(\theta) \propto /(y|\theta) \times g(\theta)$$
 [9-17]

The type of a priori distribution and its uncertainty are generally based on information provided by documented cases or the literature of the sector [Kass and Wasserman 1994] In the Bayesian inference of the model parameters, the best estimate of the parameters θ is the set of the same θ_{ML} to which the maximum posterior probability (MAP) corresponds:

$$\theta_{ML}(y) = \frac{\arg \max}{\theta} \left(L(y|\theta) \times g(\theta) \right)$$
[9-18]

For the estimation of parameters in a continuous domain, it is necessary to assess the posterior distribution of the parameters and the related uncertainties The uncertainty can be determined through direct formulas for the propagation of the error or it can be obtained







through the application of numerical methods such as Markov chain Monte Carlo methods (MCMC).

If the following assumptions are satisfied:

- the state parameters are defined in a continuous domain;
- the value of the parameters θ_k at a certain instant k is a linear function of the value of the parameters θ_{k-1} at the instant k-1, and of the realization of the model uncertainties w_k, that is:

$$\theta_k = F_k \theta_{k-1} + w_k \tag{9-19}$$

Where F_k is a matrix representing the temporal relationship of the parameters.

• the measurements y_k observed at instant k can be considered a linear function of the state θk at instant k, and of the realization of the instrumental error v_k , that is:

$$v_k = H_k \theta_k + v_k \tag{9-20}$$

where H_k is a matrix representing the relationship between parameters and measures.

 the probabilistic distributions of the realizations w_k and v_k are multinormal with zero averages;

then the estimation of the parameters at the instant k and the corresponding covariance matrix can occur through an application of the Kalman filter.

Monte Carlo methods

Through the application of Monte Carlo methods, numerical results can be obtained on the basis of repeated random sampling. This method is used when it is difficult or impossible to analyze variables in closed form or it is not practical to apply a deterministic algorithm. In the Monte Carlo method the following particular scheme is followed:

- define a domain of possible inputs;
- randomly generating inputs by means of a probability distribution on the domain;
- perform the deterministic calculation on the basis of the inputs;
- aggregate the results using a probabilistic function.

Deterministic evaluation is the direct solution of a system of equations based on each given input. In this way, once input and output data are given, it is possible to understand which are the variables on whose variability the output depends.

For further information on Monte Carlo methods, see [89].

Multi-Model Approach

This methodology is based on the generation of a large number of candidate models: by comparing many direct solutions to well-determined problems, the solution of an inverse problem is found. The most significant advantage is that each of the generated models can certainly respect the physical-mechanical compatibility since its creation, which in classical methods are verified a posteriori.





Each model is associated with the value of an objective function and is identified by its value. The "right" model is the one associated with the minimum value of the objective function.

The first thing to do is to create the model and then the optimization procedure starts. A global stochastic search and optimization algorithm is used to select a population of candidate models. A global stochastic research algorithm called PSGL (Probabilistic Global Search Lausanne) is used to minimize the cost function. Initially, it is assumed that the probability density function is uniform for each parameter. There is no model generation during this phase. The second step is the sampling cycle. In this cycle, several models are generated and the cost function (objective function) is calculated for each model. At the end of this cycle, only the model associated with the minimum of the cost function, which is called Best Sample (BS), is stored. To edit the PDF for each variable, a probability update cycle is used, which works as follows:

- identification of the range in which the BS is located;
- multiplying the interval by a factor greater than one;
- normalization of PDF.

This probability updating cycle is applied multiple times to find the range containing the Current Best Point. The current best point at the end of this focus cycle is called CBEST. Since both measurements and models contain errors, they are taken into account all models whose objective function is below an adequate threshold. The threshold value therefore depends on the rough estimate of modeling and measurement errors. Applicable models are grouped into classes using known data mining techniques. Each class has a representative "flag" model. The final comparison takes place between the "flag" models.

The objective function to be minimized is:

$$f(p) = \sum_{i=1}^{n} \left[\left[1 - MAC \left(\varphi_{id,i} \varphi_{fe,i}(p) + \left\| \frac{\omega_{id,i} - \omega_{fe,i}(p)}{\omega_{id,i}} \right\| \right) \right] \right]$$
[9-21]

Where:

- ω represents the modal angular frequencies;
- the subscripts "id" refers to the quantities identified experimentally;
- the subscripts "fe" refers to the quantities extracted from the finite element model.

9.4 Procedures for identifying KPIs from diagnostic testing

Once model updating has been performed through the procedures described in the previous paragraphs, it is possible to identify the PIs and the related KPIs among the parameters which have been initially selected.

In general, there is no clear distinction between PI and KPI. The KPIs are determined by a number of performance indicators collected at the operational level.

According to [2], Key Performance indicators are:

- Reliability
- Availability
- Safety
- Economy
- Environment





IM-SAFE proposal, though, is based on the **RAMSSHE€P** approach [90], which includes:

- Safety, Reliability and Security (S, R, S) a combined KPI
- Availability and Maintainability (A, M) a combined KPI
- Economy € (i.e. Costs)
- Environment (E)
- Health and Politics (H, P) a combined KPI



Figure 9.4 – RAMSSHE€P approach (Extract from [90])

Since some of the KPIs above are difficult to be assessed, it should be noted that within the IM-SAFE project only KPIs relating to structural safety are taken into consideration.

Hence, starting from a diagnostic load test, it is possible to obtain information only in relation to the Safety, Reliability and Security combined KPIs: measurements deriving from the performance of diagnostic load testing includes displacements, deformations, frequency, vibrations/oscillations and eventually, the presence of cracks.

The parameters obtained from the measurements carried out during the diagnostic load test are compared with the field parameters. The latter are not necessarily parameters directly measured by the sensors or may be derived quantities. Given that, the parameters can be direct or indirect, depending on whether they already have the required information inside them or whether they need to be reprocessed to obtain the desired measurement. An example of an indirect parameter can be the frequency derived from the acceleration measured by the accelerometers. Measurements listed above can be identified as Performance Indicators: among them it is possible to collect which PIs may be considered also as KPIs.

Among parameters in Figure 9.5 – Performance Indicators (PIs) and related Key Performance Indicators (KPIs) from diagnostic load testing., which lists a series of PIs, it is possible to select KPIs that can be derived from a diagnostic load test. Figure 9.5 shows PIs and the related KPIs.







Figure 9.5 – Performance Indicators (PIs) and related Key Performance Indicators (KPIs) from diagnostic load testing.





10 Conclusions

This report aims to provide technical background for the formulation of the proposal for the mandate to CEN for a further amendment to the existing EU standards on diagnostics of structures based on survey data and for a new standard for monitoring strategies. For this purpose, this document contains an overview of the damage and degradation mechanisms that can affect bridges and tunnels, focusing only on concrete structures (non-reinforced/reinforced/prestressed). For each damage process, Damage Indicators (Dis) and Performance Indicators (PIs) have been identified, as well as the surveying technologies which may be used for its detection.

Based on the information above, a four-step damage detection procedure has been developed: it consists of damage identification and localization, damage quantification, damage modelling and prediction and damage monitoring. This procedure accounts for type, size and location of defects or other relevant issues depending on the type of structure, the actions on structure, and the risks that may potentially affect the structure in the future.

Since damage identification and localization is based, among other things, on inspection outcomes, a classification of inspection methodologies is given, describing each type of inspection with respect to its objectives, frequency of execution, data collection methods and outcomes.

Principles of structural monitoring are also provided, introducing the objectives and requirements of monitoring systems and their classification based on the period of execution and on the purpose. Monitoring systems architecture and design processes are also described, as well as a guidance for monitoring systems installation and management and for data acquisition, processing and treatment.

This document also includes a procedure for damage classification and for the development of a damage classification database.

Since damages that might appear in a given structure are directly linked to the actions that structure is bearing, a description of load models provided by standards on both bridges and tunnels is given. Furthermore, this document provides procedures for determining actions on structures based on condition survey data, which can be used for the development of representative and accurate load models.

Lastly, procedures for assessing KPIs from diagnostic testing are given, including the description of a decision-making process for model updating purposes and a review of the existing model calibration methods.





11 Bibliography

- ^[1] IMSAFE Glossary, "IMSAFE Terminology," 2022.
- ^[2] CA1406-WG3, COST ACTION 1406 Establishment of a Quality Control Plan, 2018.
- ^[3] FIB-TG3.3, "Existing Concrete Structures Life Management, Testing and Structural Health Monitoring," 2021.
- ^[4] IM-SAFE-WP3, Progress Report D3: Appraisal of methods for safety evaluation and risk management (under development), 2022.
- ^[5] Haydl-Nikiel, "DESIGN AND CONSTRUCTION ERRORS—CASE STUDIES," *ASCE LIbrary*, 2000.
- ^[6] FIB Model Code 2020, FIB Model Code 2020 (under review), 2022.
- ^[7] SAMCO, Guideline for Structural Health Monitoring, 2006.
- ^[8] FIB-Bulletin59, Condition control and assessment of reinforced concrete structures, 2011.
- ^[9] FIB-Bulletin62, Structural concrete-Textbook on behavior, design and performance, 2012.
- ^[10] IMSAFE-D3.2, "D3.2 BACKGROUND MATERIALS FOR IMPLEMENTATION OF DECISION-MAKING REGARDING MAINTENANCE STRATEGIES," 2022.
- ^[11] IMSAFE-D2.1, "ONLINE INTERACTIVE CATALOGUE OF SURVEYING TECHNOLOGIES FOR TRANSPORT INFRASTRUCTURE," 2022 (under development).
- ^[12] L. Deng, W. Wang and Y. Yu, "State-of-the-Art Review on the Causes and Mechanisms of Bridge Collapse," *Journal of Performance of Constructed Facilities*, no. 04015005-1, 2015.
- ^[13] J.-S. Tan, K. Elbaz, Z.-F. Wang, J. Shui Shen and J. Chen, "Lessons Learnt from Bridge Collapse: A View of Sustainable Management," *Sustainability*, vol. 12(3), no. 1205, 2020.
- ^[14] T. D. Everett, P. Wykamp, H. A. Capers, W. R. Cox, T. S. Drda, L. Hummel, P. Jensen, D. A. Juntunen, T. Kimball and G. A. Washer, "Bridge Evaluation Quality Assurance in Europe," American Trade Initiatives, Alexandria, 2008.
- ^[15] FIB Model Code 2010, FIB Model Code 2010, 2010.
- ^[16] UNI, Guidelines for structural health monitoring, 2016.
- ^[17] MIT, Linee guida per la classificazione e gestione del rischio, la valutazione della sicurezza e il monitoraggio dei ponti esistenti, 2020.
- ^[18] Fib_TG3.2, "Modelling of structural performance of existing concrete structures," 2020.
- ^[19] "building SMART," [Online]. Available: https://www.buildingsmartitalia.org/utenti/pubblicazioni/guida-ifc-per-i-ponti/casostudio-b-ponte-con-sezione-mista-acciaio-calcestruzzo/.
- ^[20] M. G. Alberti, J. Alonso Trigueros, Á. Moreno Bazán and M. G. Alberti, "New Perspectives for BIM Usage in Transportation Infrastructure Projects," *Applied sciences*, 2020.
- ^[21] "SIMSCALE," [Online]. Available: https://www.simscale.com/projects/tags/nonlinear/#viewMode=thumbView&sortBy=pr oject_popular&time=all.
- ^[22] "Construction Specialties," [Online]. Available: https://www.csgroup.com.au/products/allway-expansion-joint-covers/parking-joint-covers/gasketedparking-joint-covers/pda-pds-series/.
- ^[23] T. Suntharavadivel and T. Aravinthan, "Overview of external post-tensioning in bridges," *Engineering*, 2005.





- ^[24] "BASALT.TODAY," [Online]. Available: https://basalt.today/2020/06/43759/.
- ^[25] "AUTODESK," [Online]. Available: https://blogs.autodesk.com/infrastructurereimagined/building-the-fanli-bridge-with-vertical-and-horizontal-bim/.
- ^[26] G. Barla, *Tunnelling under squeezing rock conditions,* 2002.
- ^[27] T. R. c. (E-C129), Use of Inclinometers for Geotechnical Instrumentation on Transportation *Projects*, 2008.
- ^[28] H. Konietzky, "Overview about field measurements in rock masses," 2021.
- ^[29] COST 345, "COST 345: Procedures Required for Asessing Highway Structures," European Commission, Directorate General Transport, 2001.
- ^[30] Highways England, Transport Scotland, Llywodraeth Cymru Welsh Government, Department for Infrastructure, "CS 450 Inspection of highway structures version 0.1.0," Highways England, 2021.
- ^[31] M. Artus and C. Koch, "State of the art in damage information modeling for bridges," in *26th International Workshop on Intelligent Computing in Engineering, EG-ICE 2019,* Leuven, 2019.
- ^[32] M. Artus and C. Koch, "Modelling Geometry and Semantics of Physical Damages using IFC," in 27TH INTERNATIONAL WORKSHOP ON INTELLIGENT COMPUTING IN ENGINEERING, Berlin, Germany (Digitally), 2020.
- ^[33] J. Silva, Y. Garbatov and C. G. Soares, "Ultimate strength assessment of rectangular steel plates subjected to a random localised corrosion degradation," *Engineering Structures*, pp. Vol 52, Pages: 295-305, 2013.
- ^[34] H. Tran-Ngoc, S. Khatir, G. De Roeck, T. Bui-Tien, L. Nguyen-Ngoc and M. Abdel Wahab, "Model Updating for Nam O Bridge Using Particle Swarm Optimization Algorithm and Genetic Algorithm," *Sensors*, vol. 18, p. 4131, 2018.
- ^[35] Y. Garbatov, C. G. Soares, J. Parunov and J. Kodvanj, "Tensile strength assessment of corroded small scale specimens," *Corrosion Science*, pp. Vol 85, Pages: 296-303, 2014.
- ^[36] F. Wetscher, R. Stock and R. Pippan, "Changes in the mechanical properties of a pearlitic steel due to large shear deformation," *Materials Science and Engineering: A*, pp. Vol 445–446, Pages: Pages 237-243, 2007.
- ^[37] Y. Wang, S. Xu, H. Wang and A. Li, "Predicting the residual strength and deformability of corroded steel plate based on the corrosion morphology," *Construction and Building Materials*, vol. 152, pp. 777-793, 2017.
- ^[38] Y. Wang, S. Xu and A. Li, "Flexural performance evaluation of corroded steel beams based on 3D corrosion morphology," *Structure and Infrastructure Engineering*, vol. 16, pp. 1562-1577, 2020.
- ^[39] J. Neter, M. Kutner, C. Nachtsheim and W. Wasserman, Applied Linear Statistical Models, New York: McGraw-Hill Irwin, 1996.
- ^[40] G. A. F. Seber and A. Lee, Linear Regression Analysis, Hoboken: John Wiley & Sons, 2003.
- ^[41] V. N. Vapnik, The Nature of Statistical Learning Theory, New York: Springer-Verlag, 1995.
- ^[42] A. J. Smola and B. Schölkopf, "A tutorial on support vector regression," *Statistics and Computing*, vol. 14, pp. 199-222, 2004.
- ^[43] S. Marelli, N. Lüthen and B. Sudret, UQLab User Manual Polynomial Chaos Expansions, Zürich: ETH Zurich, 2022.
- [44] R. G. Ghanem and P. D. Spanos, Stochastic Finite Element Method: Response Statistics, New York: Springer New York, 1991.





- ^[45] S. Marelli and B. Sudret, "An active-learning algorithm that combines sparse polynomial chaos expansions and bootstrap for structural reliability analysis," *Structural Safety*, vol. 75, pp. 67-74, 2018.
- ^[46] C. Lataniotis, D. Wicaksono, S. Marelli and B. Sudret, UQLab User Manual Kriging (Gaussian Process Modeling), Zürich: ETH Zurich, 2022.
- ^[47] C. E. Rasmussen and C. K. I. Williams, Gaussian Processes for Machine Learning, Cambridge: The MIT Press, 2006.
- ^[48] V. Dubourg, Adaptive surrogate models for reliability analysis and reliability-based design optimization, Clermont-Ferrand: Université Blaise Pascal, 2011.
- ^[49] A. Shrestha and A. Mahmood, "Review of Deep Learning Algorithms and Architectures," *IEEE Access*, vol. 7, pp. 53040-53065, 2019.
- ^[50] P. P. Shinde and S. Shah, "A Review of Machine Learning and Deep Learning Applications," *Fourth International Conference on Computing Communication Control and Automation (ICCUBEA)*, pp. 1-6, 2018.
- ^[51] Y. Bengio, "Learning Deep Architectures for AI," *Foundations and Trends® in Machine Learning*, vol. 2, pp. 1-127, 2009.
- ^[52] F. C. R and W. Keith, "An introduction to structural health monitoring," *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences,* pp. Vol 365, Pages: 303-315, 2007.
- ^[53] V. Patidar, S. Lab, K. C. Sinha, P. D. Thompson, A. Shirolé and W. Hyman, "Performance Measures for Enhanced Bridge Management," *Transportation Research Record*, pp. Vol: 1991 Pages: 43-53, 2007.
- ^[54] Z. M. Angelo, F. Flora and C. J. Ramon, "State-Of-Research on Performance Indicators for Bridge Quality Control and Management," *Frontiers in Built Environment*, 2019.
- ^[55] EUROSTRUCT, "COST TU 1406 WG2: Perfomance Goals for Roadway Bridges," 2017.
- ^[56] R. Elmasri and S. B. Navathe, Fundamentals of Database Systems, 2007.
- ^[57] C. A. Tokognon and G. Y. T. a. Y. Y. B. Gao, "Structural Health Monitoring Framework Based on Internet of Things: A Survey," *IEEE Internet of Things Journal*, pp. Vol:4 Pages: 619-635, 2017.
- ^[58] S. Jeong, R. Hou, J. P. Lynch, H. Sohn and K. H. Law, "An information modeling framework for bridge monitoring," *Advances in Engineering Software*, pp. Vol: 114, Pages: 11-31, 2017.
- ^[59] M. Theiler and K. Smarsly, "IFC Monitor An IFC schema extension for modeling structural health monitoring systems," *Advanced Engineering Informatics*, pp. Vol: 37, Pages: 54-65, 2018.
- ^[60] CEN, CEN/TS 17440:2020 Assessment and retrofitting of existing, 2020.
- ^[61] EN-1991-1, Actions on structures-Part 1-1: General actions Densities, self-weight, imposed loads for buildings, 2004.
- ^[62] EN-1990, Basis of structural design, 2002.
- ^[63] JCSS_part2, "JCSS probabilistic model code- Part 2: load models," 2001.
- ^[64] EN-1991-1-3, Actions on structures- Part 1-3: General actions Snow loads, 2004.
- ^[65] EN-1991-1-4, Actions on structures-Part 1-4: General actions Wind actions, 2010.
- ^[66] EN-1991-2, Actions on structures-Part 2: Traffic loads on bridges, 2005.
- ^[67] Maljaars, "Evaluation of traffic load models for fatigue verification of European road bridges," *Elsevier*, 2020.
- ^[68] EN-1991-1-5, Actions on structures- Part 1-5: General actions Thermal actions, 2004.





- ^[69] JCSS, JCSS probabilistic model code- Thermal actions, 2015.
- ^[70] EN-1991-1-7, Actions on structures- Part 1-7: General actions Accidental actions, 2006.
- ^[71] EN-1998, Eurocode 8: Design of structures for earthquake resistance, 2009.
- ^[72] Qing-Yong-Sankaran-Xiaomo, "Probabilistic degradation modeling of segmental linings assembled circular tunnels," *Structural Concrete,* 2015.
- ^[73] ITA-Working-Group-Research, Guidelines for Design of Shield Tunnel Lining, 1999.
- ^[74] Terzaghi, Rock defects and loads on Tunnel Supports. In: Introduction to rock tunnelling with steel support, 1946.
- ^[75] Unal, Development of design guidelines and roof-control standards for coal-mine roofs, 1983.
- ^[76] S. Kasap, "Thermoelectric effects in metals: thermocouples," *Canada: Department of Electrical Engineering University of Saskatchewan,* 2001.
- ^[77] I. Garrido, S. Lagüela and P. Arias, "Infrared thermography's application to infrastructure inspections.," *Infrastructures*, p. 3(3), 2018.
- ^[78] A. Haberkorn, European Snow Booklet, 2019.
- ^[79] L. An, T. Lu, J. Xu, Z. Wang, M. Xu and T. Wang, "Soft sensor for measuring wind pressure," *International Journal of Mechanical Sciences*, pp. 386-392, 2018.
- ^[80] A. Micu, E. J. O'Brien and E. Sevillano, "Application of image processing to the analysis of congested traffic.," *Civil Engineering Research in Ireland*, 2016.
- ^[81] E. C. f. Standardization, EN 1998-1 Eurocode 8: Design of structures for earthquake resistance. Part 1: General rules, seismic actions and rules for buildings., Brussels, 2013.
- ^[82] Y. Kaya and C. Ventura, "Seismic Structural Health Monitoring of Bridges in British Columbia, Canada," in *Seismic Structural Health Monitoring. From Theory to Successful Applications*, Switzerland, Springer Tracts in Civil Engineering, 2019, pp. 31-50.
- ^[83] T. Xu, M. Wang, L. Yu, C. Lv, Y. Dong and Y. Tian, "Research on the Earth Pressure and Internal Force of a High-Fill Open-Cut Tunnel Using a Bilayer Lining Design: A Field Test Using an FBG Automatic Data Acquisition System," *Sensors*, p. 19(7):1487, 2019.
- ^[84] L. Simeoni and L. Mongiovì, "Inclinometer Monitoring of the Castelrotto Landslide in Italy," *Journal of Geotechnical and Geoenvironmental Engineering*, p. 133:6(653), 2007.
- ^[85] ASCE, Bridge Load Testing: State-of-the-Practice, 2021.
- ^[86] K. Worden, C. Farrar, G. Manson and P. G., "The fundamental axioms of structural health monitoring," *Proceedings of the Royal Society of London Series A, Mathematical, physical and engineering sciences, V. 463,* pp. 1639-1644, 2007.
- ^[87] E. Simoen, G. De Roeck and G. Lombaert, "Dealing with uncertainty in model updating for damage assessment: A review," *Mechanical Systems and Signal Processing Vol. 56-57*, pp. 123-149, 2015.
- ^[88] Berman-Nagy, "Improvement of a Large Analytical Model Using Test Data," *AIAA Journal*, 1983.
- ^[89] Mares-Mottershead-J.E-Friswell, "Stochastic model updating : Part 1 theory and simulated example," *Engineering*, 2006.
- ^[90] CA1406-WG2, "COST-ACTION-1406-WG2 Performance goals for roadway bridges," 2017.
- ^[91] Hajdin-Kušar-Mašović-Linneberg-Amado-Tanasić, "COST ACTION 1406-WG3: Technical Report-Establishment of a Quality Control Plan," 2018.
- ^[92] FIB-G3.3, "Existing Concrete Structures Life Management, Testing and Structural Health Monitoring," 2021.





- ^[93] IM-SAFE, Progress Report D3: Appraisal of methods for safety evaluation and risk management (under development), 2022.
- ^[94] E. Horszczaruk, Abrasion resistance of high-strength concrete in hydraulic structures, vol. 259, 2005, pp. 62-69.
- ^[95] ASTM, Designation: G40 21a.
- ^[96] P. A. Johnson, R. D. Hey, M. W. Horst and A. J. Hess, "AGGRADATION AT BRIDGES," 2001.
- ^[97] P. P. Shrestha, M. Asce and K. P. Neupane, Identification of Geotechnical-Related Problems Impacting Cost, Schedule, and Claims on Bridge Construction Projects, 2020.
- ^[98] A. Kalwik, P. Postawa and M. Nabiałek, Analysis of Ageing Processes of Semi-Crystalline Materials Effect of processing conditions on injection moldings properties View project.
- ^[99] "EN 1993-1-9: Eurocode 3: Design of steel structures Part 1-9: Fatigue," 2005.
- ¹⁰ M. Mehrjoo, N. Khaji, H. Moharrami and A. Bahreininejad, Damage detection of truss)] bridge joints using Artificial Neural Networks, vol. 35, 2008, pp. 1122-1131.
- ¹⁰ Y. Wang and Y. Zheng, Research on the damage evolution process of steel wire with pre-1] corroded defects in cable-stayed bridges, vol. 9, MDPI AG, 2019.
- ¹⁰ D. Zhou, R. Li, J. Wang and C. Guo, Study on Impact Behavior and Impact Force of Bridge 2] Pier Subjected to Vehicle Collision, vol. 2017, Hindawi Limited, 2017.
- ¹⁰ ISO 8100-20:2018(en). 3]
- ¹⁰ "TIPTON AND SOREM ON CONTROLLED OVERLOADING 585".
- 4]
- ¹⁰ "Deterioration processes in reinforced concrete: an overview".
- 5]
- ¹⁰ I. Sims and B. Brown, "Concrete Aggregates".
- 5]
- ¹⁰ M. P. Singh and H. Nandan, Effects of environmental temperature fluctuations on 7] vibration properties of concrete bridge structures, Taylor and Francis/ Balkema, 2008, p.
- 331.
- ¹⁰ G. P. A. G. Van Zijl and V. Slowik, "RILEM State-of-the-Art Reports A Framework for Durability Design with Strain-Hardening Cement-Based Composites (SHCC) State-of-the-Art Report of the RILEM Technical Committee 240-FDS".
- ¹⁰ ISO 8044:1999(en) Corrosion of metals and alloys Basic terms and definitions.
- 3]
- ¹¹ The Concrete Society, Biological growth Concrete Society.
- 2]
- ¹¹ N. Mokni, S. Olivella, J. Carrera and B. Otto, Surface movements in a rock massif induced 1]
- by drainage associated to tunnel excavation, vol. 37, 2013, pp. 1162-1188.
- ¹¹ S. Abbas, A. M. Soliman and M. L. Nehdi, Experimental study on settlement and punching 2] behavior of full-scale RC and SFRC precast tunnel lining segments, vol. 72, Elsevier BV, 2014, pp. 1-10.
- ¹¹ "DEPARTMENT OF THE ARMY Engineering and Design TUNNELS AND SHAFTS IN ROCK," 3] 1997.
- ¹¹ X. Liu, C. Zhang, C. Zhang and Y. Yuan, Ultimate load-carrying capacity of the longitudinal 4] joints in segmental tunnel linings, vol. 18, Wiley-Blackwell, 2017, pp. 693-709.
- ¹¹ Chen Yunhua, Fatigue rebounding at the interface of reinforcement concrete.
- 5]
- ¹¹ N. Bettini, R. Felicetti, G. Lilliu, A. Meda and P. Riva, "FIRE ANALYSIS OF REINFORCED
 - 5] CONCRETE PRECAST TUNNEL LINING CONSIDERING THE SPALLING EFFECT," 2015.
 - ¹¹ K. Tsuno and K. Kishida, Evaluation of spalling of concrete pieces from tunnel lining 7] employing joint shear model, vol. 103, Elsevier Ltd, 2020.





- ^{[11} Austrian Society for Construction Technology, Austrian internal Guideline for Inspection ^{3]} of Tunnels and the Guideline, 2013.
- ¹¹ NZ Transport Agency and Opus International Consultants., Bridge manual : manual ³ number: SP/M/022.
- A. Usmani and S. K. Jain, "SEEPAGE IN TUNNELS UNDER DIFFERENT FIELD CONDITIONS".
- ^{[12} J. Xiaoguang, L. Hongwei and Z. Shixiang, Research of water inflow impact on ecological ^[1] environment for Cimushan tunnel, Vols. 594-597, 2012, pp. 1263-1268.
- ¹² C. C. Lu and J. H. Hwang, Damage analysis of the new Sanyi railway tunnel in the 1999
- ^{2]} Chi-Chi earthquake: Necessity of second lining reinforcement, vol. 73, Elsevier Ltd, 2018, pp. 48-59.
- M. Li, W. Xu, Y. Wang, Q. Tian and J. Liu, Shrinkage crack inhibiting of cast in situ tunnel
 concrete by double regulation on temperature and deformation of concrete at early age,
 vol. 240, Elsevier Ltd, 2020.
- J. Liu, J.-X. Fang, J.-J. Chen and G. Xu, Evaluation of Design Provisions for Interface Shear
 Transfer between Concretes Cast at Different Times, vol. 24, American Society of Civil Engineers (ASCE), 2019, p. 06019002.
- H. Ye, X. Gao and L. Zhang, Influence of time-dependent rheological properties on distinct-layer casting of self-compacting concrete, vol. 199, Elsevier Ltd, 2019, pp. 214-224.
- B. Zima and M. Rucka, Wave propagation in damage assessment of ground anchors, vol.
 628, Institute of Physics Publishing, 2015.
- 7. Author, P. J. Sabatini, D. G. Pass and R. C. Bachus, "GEOTECHNICAL ENGINEERING
 CIRCULAR NO. 4 Ground Anchors and Anchored Systems 4. Report Date 13 Type of Report and Period Covered Unclassified," 1999.





12 Appendixes

Annex A - Damage and degradation processes

Appendix A1. Damage and degradation processes identification

Appendix A2. Damage indicators

Appendix A3. Surveying technologies

Appendix A4. Data analysis methods for specific surveying technologies Fibre optic sensors

Annex B - Damage characterization procedure

Appendix B1. Damage and degradation processes identification





REVIEW OF METHODOLOGIES AND INSTRUMENTS FOR DIAGNOSTICS OF TRANSPORT INFRASTRUCTURE APPENDIXES

Annex A - Damage and degradation processes





This project has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 958171. © 2021 IM-SAFE-Proiect | TNO. The Netherlands. All right reserved |



Appendix A1 Damage and degradation processes identification

A1.1 Bridges

- A1.1.1 Environmental exposure or external factors
- A1.1.1.1 Physical processes
- A1.1.1.1.1 Abrasion

Abrasion is defined as the process by which relative motion between a surface and hard particles or protuberances on an opposing surface produces abrasive wear of the surface [91]. This process involves the removal of material from a solid object when loaded against hard particles which have equal or greater hardness; these particles can originate externally or from fractures of asperities. Abrasive wear of materials changes the surfaces and dimensions of the components of a structure. All construction materials have abrasion resistance which mainly depends on their constitutive properties.

Properties which mainly define the abrasion resistance of concrete structures are the surface finishing and the curing conditions [92]. Nevertheless, the abrasion resistance of concrete can be affected by environmental conditions (e.g. by the action of solid particles carried by the water or frictional forces due to ice formation), dosage of aggregates, or the use of supplementary cementitious materials. Abrasion can be generated by a number of sources; the most common is the action of airborne or waterborne particles although the collision of vehicles with the soffit and/or superstructure of bridges is also an important source of abrasion.

A1.1.1.1.2 Aggradation (alluviation)

Aggradation results when the bed-material load supplied to a reach of river from upstream exceeds the rivers capacity to transport it. Aggradation at bridges is critical because the deposited material (which can lead to biological growth in different areas) can significantly decrease the waterway opening beneath a bridge, thereby increasing contraction scour, upstream flooding, and bridge overtopping [93]. Long-term aggradation may become a major problem at some bridges, especially in multiple-span bridges, because in this typology aggradation at one or more spans can lead to the potential for increased contraction scour at the remaining spans.

A1.1.1.1.3 Erosion

Erosion is a progressive loss of original material (concrete) from a solid surface due to mechanical interaction between that surface and a fluid, a multicomponent fluid, or impinging liquid or solid particles [91], although wind impact can lead also in erosion.

In bridges, erosion is generated by water inflowing under bridge foundations (scour). Scour occurs in three main forms, namely, general scour, contraction scour and local scour. Firstly, general scour occurs naturally in river channels, may occur because of changes in the hydraulic parameters governing the channel form such as changes in the flow rate or changes in the quantity of sediment in the channel. Secondly, contraction scour occurs because of the reduction in the channel's cross-sectional area that arises due to the construction of structures such as bridge piers and abutments. Finally, local scour occurs around individual bridge piers and abutments. Downward flow is induced at the upstream end of bridge piers, leading to very localized erosion in the direct vicinity of the structure. Erosion in bridges is quantified in terms of rate, that means loss of material (erosion) with exposure duration.





A1.1.1.1.4 Changing geotechnical properties

Geotechnical properties (e.g cohesion, angle of internal friction, capillarity, permeability) are defined as all specifications of the soil material that are considered to properly characterize the soil under the structure. Geotechnical properties are highly important to investigate the impact of soil-structure interaction and to evaluate the structure stability.

Geotechnical properties are prone to change due many reasons, most common are due seepage, ground water, encountering materials different in classification from those predicted, and replacement of supplementary inappropriate material. However, failures identifying subgrade conditions, aggradation, abrasion, and inaccurate estimation of erosion materials are also important causes of changing geotechnical properties [94].

A1.1.1.5 Aging of material

Degradation contributes to the formation of irreversible or much less reversible changes in the structure and physical properties of construction materials. Under natural conditions, most of construction materials undergo aging. Aging of the material is the result of the interaction of different types of degradation with different intensities [95].

There are many types of degradations due aging. Between more relevant are thermal degradation, which occurs under the influence of the increased operating temperature or its rapid changes. Chemical degradation, caused by the effect of chemical compounds. Radiation degradation, which occurs under the impact of high-energy radiation. Biodegradation, caused by the activity of microorganisms and enzymes on the material. Finally, mechano-degradation occurring under the influence of the breakage of macromolecules as a result of exceeding the cohesion forces.

Aging of material starts in the surface of materials, causing fogging or microcracks, their presence has a destructive effect on the material since it allows the penetration of volatile substances in their deeper layers, which facilitates the intensification strength (tensile or bending strength) affecting the structure stability.

A1.1.1.1.6 Fatigue

Fatigue is understood as the process of initiation of cracks through a structural element due to the action of fluctuating stress [96]. Loads that produce fatigue could be a set of actions or parameters based on typical loading events. Fatigue damage often occurs at joints, starting with micro-cracks that finally lead to holes. In the case of steel bridges, fatigue often occurs in the welded type of joints. Micro-cracks may grow larger when bridges are exploited and subjected to iterative loading and unloading. As a result, the cross-sectional area of members may decrease in joints location [97]. In steel bridges, corrosion can significantly reduce the fatigue life of steel wires. This phenomenon leads because the corrosion substantially affects the fatigue crack initiation life of the wire during the service period, thus reducing the steel wire's service life and residual strength [98].

A1.1.1.1.7 Impact due to an accident

Accident occurrences has significant influences on the safety of bridge structures, impact accident includes all damages arising from accidental human activity including impact, collisions from wheeled vehicles, ships, derailed trains, gas explosions.

All impacts may affect bridge in different magnitudes some collision accidents resulted in slight damage to the piers, such as concrete cracking at the impact location. While others caused severe damage to the bridge structures, such as pier fracture and bridge collapse [99]. Effect





of impact velocity and impact mass is very important, because a larger impact velocity (and greater impact mass) can result in higher impact energy, causing a larger peak impact force.

Piers are the principal components affected under impact due to an accident. Potential failure modes due this damage process are the spalling of concrete cover, the plastic hinge formation, the breakage of the piers and the rebars fracturing. Mentioned failure modes lead in a big increment on maintenance costs. Impact is also influenced by the bridge material. In case of concrete, the peak impact force and the impact duration are independent of concrete strength, because concrete is brittle material, and it is difficult for the brittle material to perform the high shear capacity in the short impact duration [99]. In case of steel higher steel strength can resist the larger impact energy, resulting in a larger impact force.

A1.1.1.1.8 Overloading of an element

When a load (in the load-carrying unit) that exceeds the rated load is applied to an element, it is overloading [100]. Same time, stresses are induced throughout the element, unless both the components geometry and the distribution of the applied loading are extremely uniform, the distribution of stresses throughout the component will be nonuniform [101]. This could produce damages to elements that are no prepared to carry the increased loads.

Overloading can be the result of static loading, in the short or long term, including creep effects or as a result of dynamic loading, resulting from impacts or seismic loading. In both instances, once the load is cancelled, the material retains some memory of the previous overload and cannot go back to its initial state [102]. Internal defects could arise creating a weak point for further aggressions.

A1.1.1.1.9 Freeze-thaw

Freeze-thaw affects concrete structures inducing an internal phase change and flow of pore water in concrete. When the porous space of concrete is highly saturated, these processes can generate internal stress high enough to rupture the material during freezing. It can result either in surface scaling and spalling, or in material volume expansion which usually induces a network of cracks [102]. Moreover, with the presence of salts the deterioration process can be greatly reduced [103]. Accordingly, the action intensity of freeze-thaw environment is graded by the basic factors contributing to the concrete damage by freezing, that is, the climate frost intensity, concrete saturation degree and presence of salts.

Concrete resists freeze-thaw damage when free water can move through capillary poresuntil reaching 'bubbles' where the ice is able to freely expand, without creating excessive internal stresses. The freeze/thaw resistance of a material is understood as the ability to withstand repeated water exposure and subsequent freeze cycling for 100 freeze/thaw cycles. This ability will be quantified in terms of the % retention of the mechanical property of interest as compared to samples exposed to water for a comparable time without subsequent freeze/thaw cycling.

A1.1.1.1.10 High temperature

A bridge structure is continuously exposed to fluctuating environmental temperature with distinct annual and diurnal trends. Changes of temperature have several effects in structures. The modulus of the material can change with the temperature thus affecting its rigidity [104]. Modal properties change with temperature gradients induced in the structural elements, usually the top of the bridge exposed to direct sun could induce temperature gradients and associated thermal stresses which lead in changes in geometric stiffness.





The internal thermal stresses could also be introduced by a change in boundary constraints such as frozen bearings or change in end fixity constraints caused by freezing and thawing of the foundations of the piers rigidly connected to the super structure [104].

Overall, hight temperature can induce changes in material stiffness, changes in modal parameters, increased stresses, changes in bearing behaviour, problems on expansion joints, between others.

A1.1.1.2 Chemical processes

A1.1.1.2.1 Alkali aggregate reaction (alkali-silica reaction)

Alkali aggregate reaction (AAR) is an expansive chemical deterioration process that is caused by the chemical reaction between reactive aggregate and the inherently alkaline cementbased composites. Symptoms of AAR include volume increase, the mapping of crack pattern formation, gel seeping from cracks, and pop-outs of aggregates [105]. Apart from the unsightly appearance that results from the presence of ASR, it can also cause significant deterioration in the mechanical properties of especially stiffness (i.e. Youngs modulus) and strength. Crack formation typically presents access to deleterious substances, which subsequently ingress into the matrix, and which may lead to various other deterioration processes, such as to the chloride-induced corrosion of embedded steel reinforcement.

Certain types of siliceous rock and mineral aggregate constituents can react with alkali hydroxides in the concrete pore solution to form a gel product which can expand to cause cracking and other damage to the hardened concrete [103]. These AAR aggregates can therefore be undesirable, but the possibility of reaction occurring and the magnitude of any resultant damage is dependent upon a combination of critical factors.

A1.1.1.2.2 Sulphate reaction

Sulphate attack affect concrete causing either softening, decay of the concrete matrix and expansive cracking. These reactions largely involve the cement paste of concrete, however, in some unusual circumstances, the sulphate action may derive from a constituent of the aggregate, or the aggregate itself might be vulnerable to attack [103]. Sulphate attack can have endogenous origin (developing without any contribution from the environment) or exogenous origins (such as sulphates contained in the soils or in liquids). In both cases, the consequence is some volume expansion owing to the delayed formation of ettringite, which is an expansive component [102].

A1.1.1.2.3 Chemical attack

A chemical attack involves all dissolution of substances or chemical reactions between substances and components that could cause problems of the concrete, due to dissolution or expansion.

The ability of concrete to resist chemical attack is primarily dependent upon the properties of the hydraulic binder, including the cement type, the presence and type of any mineral additions, the water/cement ratio and the degree of compaction. Aggregate properties only rarely directly influence the overall chemical resistance of concrete [103].

A1.1.1.2.4 Corrosion

Corrosion is an electro-chemical process by which the cross-section of steel reinforcement is reduced either reasonably uniformly or locally (that is, through pitting). Corrosion is generated by an interaction between a metal and its environment that results in changes in the properties





of the metal, and which may lead to significant impairment of the function of the metal [106]. By other hand, corrosion results from the fact that steel tends towards finding their natural form, which is oxidized. The corrosion rate of passivated steel can be less than 1 μ m per year.

Corrosion of ordinary structural steel reinforcements commences quite soon following exposure to humid air, and can then proceed rapidly in honeycombed or extensively cracked concrete. However the process is relatively slow in the absence of aggressive species, such as chloride and sulphate ions. Initially there are no visible traces of corrosion on the concrete surface, but with time staining followed by cracking, spalling and delamination become evident [106]. The development of active corrosion in reinforced concrete results from two mechanisms whose common feature is the diffusion of external agents through the pores in the concrete. These mechanisms are the carbonation process and the chloride diffusion process. In reinforcement concrete structures, the steel is normally protected by the alkalinity of the cement pore solution (pH around 13). At lower pH levels, steel attains a high corrosion potential that leads to passivity, with the formation of a thin surface film, about two and three nanometres thick, of iron hydroxides, which provides corrosion resistance [102].

Corrosion of prestressing reinforcement will also commence in a humid environment (as may be present at inadequately filled joints between precast segments) due to carbonation of the concrete, and in the presence of aggressive species - such as chloride ions. Corrosion due to chloride ions can occur in partially grouted or open ducts when contaminated water comes in contact with the tendon. Prestressing steel may also fail through stress corrosion and/or hydrogen embrittlement [106].

Corrosion can also affect metallic components, this type of corrosion can be initiated and promoted in several ways, the main ones are environment corrosion, stray electric currents, stress corrosion cracking, galvanic corrosion, crevice corrosion and bacteriological corrosion. Finally, Aluminium structures can suffer from pitting corrosion, but its effects are rarely serious. However aluminium culverts and underpasses are particularly susceptible to pitting corrosion and so are not permitted in some States [106].

A1.1.1.3 Biological processes

A1.1.1.3.1 Biological growth

Biological growth could be defined as an excessive organism growth that may affect structures inducing damages caused by itself. Biological growth usually starts with primary colonisation by micro-organisms, algae, fungi and various types of bacteria. This is followed by visible growths of algae and lichens. Dirt then collects on the surface and, together with the decaying remains of dead organisms, provides an environment suitable for more advanced biological activity [107]

In advanced cases of biological growth problems associated to concrete structures, dead lichen can provide a footing for mosses and larger plants, which can affect drainage and lead to more serious damage. The roots of plants can grow into cracks and weak spots in the concrete, resulting in bursting stresses that can increase the size of cracks and may lead to spalling [107].

A1.1.2 Damage processes related to design and construction issues

Deterioration related with design and construction errors can lead in many problems of durability in structures.

One of the commonest problems met with concrete structures is inadequate cover to the reinforcement. The evidence of this may not be immediately evident but it will show itself with time through the corrosion of the reinforcement. Problems related to stresses that arise during





construction (due construction in multiple stages) are also common. Other defects at the construction stage may include:

- o Inadequate load capacity due to a design error
- o Reinforcement incorrectly placed
- Poor reinforcement detailing and concrete specification
- Inadequate cover to the reinforcement
- Honeycombing due to aggregate grading or poor compaction
- Poor quality control during construction
- Use of reinforcing/structural steel that did not meet design requirements metallic
- o Use of metallic fasteners that promotes corrosion of the reinforcement
- The properties of materials are untested or not well understood at the time of construction

A1.2 Tunnels

- A1.2.1 Environmental exposure or external factors
- A1.2.1.1 Physical processes
- A1.2.1.1.1 Continuous vertical rock movement

In general, settlement is defined as a downward vertical movement of a point. Usually, vertical rock movements are mainly due to the following two factors: stress changes and pressure variations. When the tunnel is shallow, movements caused by total stress changes are larger than movements owing to pore pressure variations [18].

By other hand, the dissipation of pore pressure within the rock matrix, caused by drainage through the tunnel walls, and resulting consolidation processes are also responsible of continuous vertical rock movements in tunnels. Significant surface movements can also occur through consolidation processes in crystalline rock masses in which pore pressure is lowered. Indeed, in this case, neither porosity nor compressibility, are directly relevant for this problem [18]. To avoid several damages in tunnels due vertical rock movements, impermeabilization on the fractured sections is commonly used, that helps to reduce the movements on surface strongly.

A1.2.1.1.2 Local rock movement (punching)

Tunnels are usually subjected to punching and settlement loads during their service life due to local rock movements. These load effects need to be carefully studied to avoid tunnels instability problems.

Punching effect due to local rock movements in tunnels derive from a close relationship between the penetration rate and brittleness index of the rock (ratio of uniaxial compressive strength and tensile strength of rock) [19]. There is an inverse correlation between these two parameters, as brittleness index decreases, the penetration rate increases. In harder rocks with lower rock brittleness, the expansion of fragmentation area decreases, and the number and length of main cracks are also reduced outside the fracture.

A1.2.1.1.3 Higher horizontal actions (underestimation of lateral action)

Horizontal actions in tunnels can be generated by the settlement of the foundation; excessive earth pressure; failures of earthworks adjacent to a structure; water pressures produced by





inadequate or blocked drains; and by changes in the strength or degree of consolidation of the subsoil or backfill [20].

The underestimation of lateral action can lead in transverse horizontal movements, such as tilting, bulging, and sliding of the tunnel ring. This underestimation may also produce bearings and rotational movements which usually results from unsymmetrical settlement or lateral movements [20].

A1.2.1.1.4 Bending stress

Excessive circumferential bending stress can lead in excessive distortions and cracking in tunnel concrete linings [21]. In low resistance bending zones (poor rock zones or low shear zones) there is an important effect in the structural stability of tunnels due to bending stresses; in those cases bending reinforcement is usually required even though the remainder of the tunnel may have received no reinforcement or only shrinkage reinforcement. Failure due to bending stress involves formation of a single lobe parallel to the axis of the tunnel.

Bending stress is especially relevant in tunnels joints; the effect of bending on joints is generally considered by applying a reduction factor to the bending stiffness of the tunnel lining [22]. Based on the bending sign, the detectable damage will be different: the failure due to negative bending moment is just terminated by the crushing of intrados concrete, while failure due to positive bending moment is usually initiated by the yield of tunnel bolts.

A1.2.1.1.5 Debonding

Debonding could be defined as a separation at the interface between the substrate and the near-surface mounted or externally bonded materials [23]. Debonding phenomena are strongly depending on the load transfer mechanisms at the concrete/matrix interface. Debonding failures can take place at the ends of the strengthening surfaces in presence of high stresses at the interface between the strengthening system and the concrete (end debonding) or away from the ends of the bonded strengthening surfaces when they are induced by flexural or flexural-shear cracks (intermediate crack induced debonding).

A1.2.1.1.6 Partial spalling of concrete cover

Spalling is defined as the breaking off fragments or solid particles from a surface, in consequence, spalling phenomena can leads to a reduction of the concrete section in tunnels.

Cracks seen on tunnel lining surfaces have a range in widths, from small to large, and it is assumed that closed cracks with greater widths have a greater risk of spalling. The structural effect of the spalling are two folds: a reduction of the concrete section leads to a decrease of the bearing capacity of the lining but also a stiffness decreases leading to a reduction of the actions due to thermal expansion [24].

In tunnels with plain concrete linings, for which there is no adhesion of the concrete to reinforcing bars, any closed cracks that run across the width of the lining can cause spalling. In practice, the risk of closed cracks causing concrete to spall off is reduced by the roughness of the mating faces of the cracks causing frictional resistance [25].

A1.2.1.1.7 Overloading (rock movement) of prestressing

Excessive deflections can be generated by higher than anticipated live loads and through a reduction in the load carrying capacity of the superstructure or substructures. Excessive deflections of prestressed superstructures can result from the use of prestressing reinforcement with a higher relaxation than assumed in design. In case of tunnels, anchor





systems are prone to be overloaded due to local movements or global mountains deformations, leading in excessive strengths and anchor elongations [26].

A1.2.1.1.8 Deformation of the ground

Some structures have a highest potential of ground deformations (e.g. due to groundwater changes, mining or liquefaction); their effects on the structures and on the performance requirements for the road link shall be taken into consideration in the development and design of appropriate mitigation measures [27].

There are many factors which influence the deformation of the ground in structures, some of them are horizontal, vertical forces and displacements induced on or within the structure, settlements, seismic actions, between others. Main causes of ground rupture, instability and soil deformation can result from earthquake shaking, earthquake induced liquefaction or cyclic softening, lateral spreading with or without associated liquefaction or cyclic softening, fault rupture associated with earthquakes and subsidence from other causes, such as groundwater changes, mining, etc [27].

A1.2.1.1.9 Water impact

Water inflow into the tunnels is highly dependent on the hydraulic conductivity of the rock mass in case of grouted and un-grouted tunnel zones. Furthermore, factors like joint roughness, percentage of contact areas and infillings can also control the seepage inflow into the tunnels.

Water inflow and water pressure controls have a big impact in the structural behaviour of tunnels. These factors need to be carefully designed. Uncontrolled water behaviour causes mechanical instability, hostile to construction progress and causes adverse environmental impacts in and around the surrounding area [28]. Environmental impact is controlled by factors including the depth of groundwater, groundwater mineralized degree, suspended moisture content and salt content etc [29].

A1.2.2 Damage processes related to design and construction issues

A1.2.2.1 Missing reinforcement

During the tunnel design, as long as the lining is seen as a structure loaded by the rock mass, moments and tensile stresses are projected to the lining and must be covered by reinforcement. Missing design reinforcement can lead in weak tunnels with low capacity resisting shear force and bending moment [30]. When an overloading as a strong earthquake happens, a large shear force and bending moment will be induced in the lining due to dynamic squeezing of the surrounding ground. Thus, the lining will be easily damaged, and can even collapse.

A1.2.2.2 Deformation due to shrinkage, temperature within the shell-blocks

Deformations due to shrinkage and temperature induce early-age cracking in concrete tunnels. Early age cracking occurs when thermal and shrinkage induced stresses in concrete exceed the tensile strength of concrete. The induced stress caused by restraint to early-age deformation in response to temperature reduction, moisture loss, and chemical reaction correspond to thermal, drying, and autogenous shrinkages. Although plastic and carbonatation shrinkage also can lead in additional stresses around the time [31].





Thermal and autogenous shrinkages dominate early age cracking, while drying shrinkage may occur when the concrete surface is exposed to environmental conditions (it usually occurs within 3–5 days even before the removal of formwork) [31]. Thus, more attention should be given to avoid excessive heat of hydration, reducing or compensating shrinkage and decreasing restraint to control early age cracking.

A1.2.2.3 Different casting times

The cyclic behaviour and effects of fatigue on interfaces between concrete cast at different times, subjected to shear stress is particularly relevant in structures subjected to important cyclic loads. In general, concrete-to-concrete interfaces are classified into three categories: (1) monolithically cast uncracked interface, (2) monolithically cast pre-cracked interface, and (3) interface between concretes cast at different times (that is, cold joint). The transfer of shear force across cold joints is a key factor to consider in concrete structures with concrete which was casting at different times [32].

A1.2.2.4 Different concrete qualities

The construction quality of concrete is possibly degraded by the distinct layer casting in real structures due to the time variable rheology of cement paste [33]. Many factors have influence in concrete quality of structures, so that is almost impossible to have an entire structure with same quality of concrete. More influenced factors in concrete quality are rheological properties of aggregates, cement type, additive type, water quality, binder ratio (w/b), between others.

When different layers of concrete are casted, a weak interface is induced (the interface bond strength decreases with the longer delay for casting the second layer). Main problems with weaknesses due different concrete qualities are related to shear and adherence strengths.

A1.2.2.5 Delamination of concrete layers (e.g. spreaded concrete)

Delamination is defined as a zone of weakness or separation along a plane parallel to the concrete surface caused by a either material, processing and/or environmental factors. The result of air voids or bleed water trapped under a dense polished concrete surface is the creation of zones of weakness (defects can be caused by several factors and mechanisms including premature finishing, excess air, delayed finishing, surface crusting, unusual bleeding and long-term factors) [20].

A1.2.2.6 Anchor failure

Most common damage mechanisms of the anchor systems are failures of the steel tendon caused by corrosion or failures of anchor body caused by weak condition of performed injected element [34]. Corrosion can occur when significant variations exist in the ground along the ground anchor length, particularly with variations in pH and resistivity. The potential for excessive loss of metal by corrosion in soil is high in the several environments, more relevant environment is soils near the groundwater table and soil exhibiting low pH, although in soils with high concentrations of aggressive loss of metal by corrosion in soil as chlorides and sites where stray currents are present the potential for excessive loss of metal by corrosion in soil is also important [35].





Appendix A2 Damage indicators

A2.1 Bridges

- A2.1.1 Environmental exposure or external factors
- A2.1.1.1 Physical processes
- A2.1.1.1.1 Abrasion

Abrasion resistance, Surface hardness, Compressive Strength.

A2.1.1.1.2 Aggradation (alluviation) Concrete spalling.

A2.1.1.1.3 Erosion Concrete spalling.

A2.1.1.1.4 Changing geotechnical properties Displacement rate, Vertical Stresses.

A2.1.1.1.5 Aging of material Wave propagation velocity, Path of deformation.

A2.1.1.1.6 Fatigue

Compressive strength, Displacement rate, Vertical Stresses.

A2.1.1.1.7 Impact due to an accident

Induced stresses due an impact, Induced displacements, Vibrations.

A2.1.1.1.8 Overloading of an element

Average crack density, Longitudinal crack density, Transverse crack density, Crack opening.

A2.1.1.1.9 Freeze-thaw

Relative dynamic modulus of elasticity, Fundamental transverse frequency, Water Saturation (with/without de-icing agent).

A2.1.1.1.10 High temperature

Modulus of elasticity of concrete, Compressive resistance, Tensile strength.





A2.1.1.2 Chemical processes

A2.1.1.2.1 Alkali aggregate reaction (alkali-silica reaction)

Phase composition, Gel presence, Content of alkali, Moisture, Presence of deformed areas, Internal expansion, Mechanical properties of concrete.

A2.1.1.2.2 Sulphate reaction

Internal heat distribution, Chloride ions content.

A2.1.1.2.3 Chemical attack

Waveform amplitude, Internal heat distribution, pH over cross section, Chloride ions content, Dielectric properties, Dielectric constant changes.

A2.1.1.2.4 Corrosion

Chloride ions content, Dielectric properties, Dielectric constant changes, Presence of defects.

A2.1.1.3 Biological processes

A2.1.1.3.1 Biological growth

Displacement ratios, Vertical stresses, Bulging.

A2.1.2 Damage processes related to design and construction issues

Displacement ratios, Deformation, Vertical stress, Horizontal stress, Chloride ions content, Dielectric properties, Dielectric constant changes, Presence of defects.

A2.2 Tunnels

- A2.2.1 Environmental exposure or external factors
- A2.2.1.1 Physical processes
- A2.2.1.1.1 Continuous vertical rock movement

Induced principal stress.

A2.2.1.1.2 Local rock movement (punching)

Vertical fault dislocation. Vertical stress. Horizontal stress. Deformation rate.

A2.2.1.1.3 Higher horizontal actions (underestimation of lateral action) Horizontal stress, Vertical stress, Displacement ratios.

A2.2.1.1.4 Bending stress

Stress distribution profiles.





A2.2.1.1.5 Debonding Strength measurement.

A2.2.1.1.6 Partial spalling of concrete cover Existence of round or oval depressions along surfaces, Ultrasonic pulse velocity.

A2.2.1.1.7 Overloading (rock movement) of prestressing Elongation at maximum load, Modulus of elasticity, Shear modulus.

A2.2.1.1.8 Deformation of the ground

Induced deformations, Induced stresses.

A2.2.1.1.9 Water impact Bulging, Mechanical instability, Induced stresses, Changes of concrete pH.

A2.2.2 Damage processes related to design and construction issues

A2.2.2.1 Missing reinforcement

Displacement ratios, Strength measurement.

A2.2.2.2 Deformation due to shrinkage, temperature within the shell-blocks Induced stresses, Shrinkage rate.

A2.2.2.3 Different casting times

Strength measurement.

A2.2.2.4 Different concrete qualities

Chemical composition, Strength, Compression rate, Hardness.

A2.2.2.5 Delamination of concrete layers (e.g. spreaded concrete)

Delayed finishing, Surface crushing, Superficial crushing.

A2.2.2.6 Anchor failure

Anchor weaknesses, Chloride ions content, Presence of defects.





Appendix A3 Surveying technologies

A3.1 Bridges

- A3.1.1 Environmental exposure or external factors
- A3.1.1.1 Physical processes

A3.1.1.1.1 Abrasion

GPR, LiDAR, Satellite, Schmidt Hammer test, Windsor Probe test, Abrasion resistance test, Boroscopy, Endoscopy, Guided Waves Propagation, Acoustic Emission, Fibre Optic Sensors. Compressive Strength Test.

	Surveying	Detectio	on Acc	uracy	Availabilit	y Versatil	ity
	Techniques	effectiven	ess				
	GPR						
	Lidar						
	Satellite						
	Schmidt hammer test						
	Windsor Probe test						
	Abrasion resistance test						
	Guided Waves Propagation						
	Acoustic Emission						
	Fibre Optic Sensors						
	Boroscopy, Endoscopy						
	Compressive Strength Test						
Ratin	a Scale	Low	Low Medium	Moder	ateN	Ioderate-High	High



A3.1.1.1.2 Aggradation (alluviation)

GPR, LiDAR, Satellite, Clinometers, Endoscopy, Boroscopy, Acoustic Emission, Fibre Optic Sensors.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
Satellite				
Clinometers				
Endoscopy, Boroscopy				
Acoustic Emission				
Fibre Optic Sensors				

Rating Scale Low Low Medium Moderate Moderate-High Hi	igh
---	-----

A3.1.1.1.3 Erosion

GPR, LiDAR, Satellite, Rpas-UAV.

Surveyir	g	Detection	Accuracy	Availability	Versatility
Techniqu	es	effectiveness			
GPR					
LiDAR					
Satellite	;				
Rpas-UA	V				
Rating Scale	Lov	w Low Me	dium Modera	ate Moderate-H	ligh High





A3.1.1.1.4 Changing geotechnical properties

GPR, LiDAR, Satellite, Rpas-UAV, Acoustic Emission, Boroscopy, Endoscopy, Guided Waves Propagation, Fibre Optic sensors, Clinometers, Accelerometers.

Surveying	Detectio	on Acc	uracy	Availability	Versatility	
Techniques	effectiven	ess				
GPR						
LiDAR						
RPAS UAV						
Satellite						
Boroscopy, Endoscopy						
FOS						
Guided Waves Propagation						
Acoustic Emission						
Clinometers						
Accelerometers						
Rating Scale	Low	Low Medium	Moderate	e Mode	rate-High	H

A3.1.1.1.5 Aging of material

GPR, LiDAR, Satellite, Rpas-UAV, Boroscopy, Endoscopy, Fibre Optic Sensors, Computed Tomography, Infrared Thermography, Infrared Spectroscopy, Half-Cell potential method, Neutron Radiography, Clinometers, Accelerometers, Mechanical tests on cored samples.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
Lidar				
RPAS UAV				
Satellite				
Boroscopy, Endoscopy				
Fibre Optic Sensors				
Computed Tomography				
Infrared Thermography				
Infrared Spectroscopy				





	Half-cell Pot methoo	tential d								
	Neutroi Radiogra	n phy								
	Clinomet	ers								
	Accelerom	eters								
	Mechanical on cored sa	tests mples								
								·		
Ratin	ig Scale	L	ow	Low Mediun	n	Moderate	Modera	ate-High	High	l

A3.1.1.1.6 Fatigue

GPR, LiDAR, Satellite, Rpas-UAV, Boroscopy, Endoscopy, Fibre Optic Sensors, Pull-out test, Fatigue test, Computed Tomography, Infrared Thermography, Mechanical tests on cored samples, Clinometers, Accelerometers.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
Rpas - UAV				
Satellite				
Boroscopy, Endoscopy				
FOS				
Pull-out test				
Fatigue test				
Computed Tomography				
Infrared Thermography				
Mechanical tests on cored samples				
Clinometers				
Accelerometers				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High
--------------	-----	---------------	----------	---------------	------

A3.1.1.1.7 Impact due an accident

WIM Systems, LiDAR.





Surveying technology	Detecti effective	on ness	Accuracy	Availability	Versat	ility
WIM Systems						
LiDAR						
Rating Scale	Low	Low Mediu	um Modera	i <mark>te M</mark> ode	rate-High	High

A3.1.1.1.8 Overloading of an element

LiDAR, Satellite, Weight-in-Motion, Fibre Optic Sensors, Accelerometers, Clinometers.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
Weight-in-Motion				
LiDAR				
Satellite				
FOS				
Accelerometers				
Clinometers				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High
--------------	-----	---------------	----------	---------------	------

A3.1.1.1.9 Freeze-thaw

GPR, LiDAR, Satellite, Spectral Analysis of Surface Waves, Water penetration test, Boroscopy, Endoscopy.

Surveying	Det	ection	Ac	curacy	Availability		Versatility
Techniques	effect	tiveness					
GPR							
Lidar							
Spectral Analys of Surface Wav	iis es						
Satellite							
Water penetration test	on						
Boroscopy, Endoscopy							
lon Chromatograph	ıy						
ng Scale	Low	Low		Voderate	Moderate-H	liah	l Hiah

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High





A3.1.1.1.10 Hight temperature

GPR, LiDAR, Rpas-UAV, Satellite, Infrared Thermography, Spectral Analysis of Surface Waves, Mechanical tests on cored samples.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
RPAS UAV				
Satellite				
Spectral Surface Waves Analysis				
Mechanical tests on cored samples				
Infrared Thermography				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.1.1.2 Chemical processes

A3.1.1.2.1 Alkali aggregate reaction

GPR, LiDAR, Potentiometric Titration, Water Penetration test, Mechanical tests on cored samples, Argentometric Titration, Ion Chromatography, pH indicators, Schmidt Hammer test, Windsor Probe test.

Surveying Techniques	Detection effectiveness	Accuracy	Availability	Versatility
GPR				
LiDAR				
Potentiometric Titration				
Satellite				
Boroscopy, Endoscopy				
Chloride Diffusion test/ Ion migration test				
Mechanical tests on cored samples				
lon Chromatography				
pH Indicators				
Schmidt hammer test				
Windsor Probe test				

	Rating Scale	Low	Low Medium	Moderate	Moderate-High	High
--	--------------	-----	------------	----------	---------------	------

Page | 19





A3.1.1.2.2 Sulphate reaction

GPR, LiDAR, Satellite, Half-cell Potential method, Ion Chromatography, pH Indicators, Boroscopy, Endoscopy, Infrared Spectroscopy, NMR.

Surveying Techniques	Detection	Accuracy	Availability	Versatility
	effectiveness			
GPR				
LiDAR				
Satellite				
Boroscopy, Endoscopy				
Argentometric Titration				
Potentiometric Titration				
lon Chromatography				
pH Indicators				
Half-cell potential				

Rating Low Low Scale	Medium Moderate	Moderate-High	High
-------------------------	-----------------	---------------	------

A3.1.1.2.3 Chemical attack

GPR, LiDAR, Satellite, Rpas-UAV, Infrared Spectroscopy, NMR, Boroscopy, Endoscopy, Argentometric titration, Potentiometric titration, Ion Chromatography, pH Indicators, infrared Spectroscopy.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
RPAS UAV				
Satellite				
Boroscopy, Endoscopy				
Argentometric Titration				
Potentiometric Titration				
lon Chromatography				
pH Indicators				
Infrared Spectroscopy				
NMR				





A3.1.1.2.4 Corrosion

Rating Scale	Low	Low	Moderate	Moderate-High	High
		Medium			

GPR, LiDAR, Satellite, Rpas-UAV, Infrared Thermography, Accelerometers, Acoustic Emission, Chloride Diffusion Test, Magnetic Flux Leakage, Pulsed Eddy Current, Boroscopy, Endoscopy, Galvanostatic Pulse Technique.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
Lidar				
RPAS UAV				
Satellite				
Boroscopy, Endoscopy				
Galvanostatic Pulse Technique				
Pulsed Eddy Current				
Acoustic Emission				
Chloride Diffusion Test				
Magnetic Flux Leakage				
Infrared Thermography				
Accelerometers				

Rating Scale

Low Medium Moderate

Moderate-High

High

A3.1.1.3 Biological processes

Low

A3.1.1.3.1 Biological growth

GPR, LiDAR, Satellite, Rpas-UAV, Satellite, Boroscopy, Endoscopy, pH Indicators.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
RPAS UAV				
Satellite				
Boroscopy, Endoscopy				
pH Indicators				

Page | 21





Rating Scale	Low	Low Medium	Moderate	Moderate-High	High
--------------	-----	---------------	----------	---------------	------

A3.1.2 Damage processes related to design and construction issues

GPR, LiDAR, Satellite, Rpas-UAV, Satellite, Boroscopy, Endoscopy, Accelerometers, Clinometers.

Surveying	Detection	Accuracy	Availability	Versatility
technology	effectiveness			
GPR				
LiDAR				
RPAS UAV				
Satellite				
Boroscopy, Endoscopy				
Accelerometers				
Clinometers				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.2 Tunnels

- A3.2.1 Environmental exposure or external factors
- A3.2.1.1 Physical processes
- A3.2.1.1.1 Continuous vertical rock movement

GPR, LiDAR.

Surveying technology	Detection effectiveness	Accuracy	Availability	Versatility
GPR				
Lidar				

Rating Scale Low	Low Medium	Moderate	Moderate-High	High
------------------	------------	----------	---------------	------

A3.2.1.1.2 Local rock movement

GPR, LIDAR.

Surveying	Detection	Accuracy	Availability	Versatility
technology	effectiveness			
GPR				
LiDAR				
Clinometers				

Page | 22




Moderate-High

High

Rating Scale Low Low Medium Moderate Moderate-High High

A3.2.1.1.3 Higher horizontal actions

LiDAR.

Surveying technology	Detection effectiveness	Accuracy	Availability	Versatility
LiDAR				

Moderate

Low Medium

A3.2.1.1.4 Bending stress

Rating Scale

LiDAR, Fibre Optic Sensors, Clinometers, Flat jacks.

Low

Detection effectiveness	Accuracy	Availability	Versatility
	Detection effectiveness	Detection Accuracy effectiveness	Detection Accuracy Availability effectiveness

Rating Scale Low Medium Moderate Moderate-High High

A3.2.1.1.5 Debonding

GPR, LiDAR, Guided Waves Propagation, Acoustic Emission, Impact–echo, Fibre Optic Sensors, Infrared Thermography, Magnetic Memory Method, Magnetic Flux Leakage, Magnetic Pulse Induction, Neutron Radiography.

Surveying	Detection	Accuracy	Availability	Versatility
technology	effectiveness			
GPR				
LiDAR				
Guided Waves Propagation				
Acoustic Emission				
Impact-echo				
FOS				
Infrared Thermography				
Magnetic Memory Method				
Magnetic Flux Leakage				





Magnetic Pulse Induction					
Neutron Radiograp	hy				
Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.2.1.1.6 Partial spalling of concrete cover

GPR, LIDAR.

Surveying technology	Detection effectiveness	Accuracy	Availability	Versatility
GPR				
Lidar				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.2.1.1.7 Overloading (rock movement) of prestressing

LiDAR, Clinometers.

Surveying technology	Detection effectiveness	Accuracy	Availability	Versatility
LiDAR				
Clinometers				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High
-					_

A3.2.1.1.8 Deformation of the ground

LiDAR, FOS.

	Surveying technology	Detection effectiveness	Accu	racy	Availabilit	ţy	Versatility	
	FOS							
	Lidar							
	Clinometers							
	Slope clinometers							
Rat	ing Scale	Low Lo	w Medium	Moder	rate N	/loderate	e-High	High

A3.2.1.1.9 Water impact

Acoustic Emission, LiDAR, Water resistance test, Water penetration test, Boroscopy, Endoscopy, Schmidt Hammer test, Ultrasonic wave propagation.





Surveying	Detection	Accuracy	Availability	Versatility
technology	effectiveness			
Acoustic				
Emission				
Lidar				
Water resistance test				
Water Penetration test				
Boroscopy, Endoscopy				
Schmidt Hammer				
Ultrasonic pulse wave propagation				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.2.2 Damage processes related to design and construction issues

A3.2.2.1.1 Missing reinforcement

Clinometers, LiDAR, GPR, Magnetic Memory Method, Magnetic Flux Leakage, Magnetic Pulse Induction.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
Magnetic Memory Method				
Magnetic Flux Leakage				
Magnetic Pulse Induction				
LiDAR				
GPR				
Clinometers				

```
Rating Scale Low Low Medium Moderate Moderate-High High
```

A3.2.2.1.2 Deformation due to shrinkage, temperature within the shell-blocks

Clinometers, LiDAR, GPR, Concrete block potential shrinkage test.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
Clinometers				





Concrete block potential shrinkage test					
Rating Scale	Low	Low Medium	Moderate	Moderate-High	Hiah

A3.2.2.1.3 Different casting times

LiDAR, GPR, Schmidt Hammer test, Windsor Probe test, Compressive strength test., Crackmeters.

Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
LiDAR				
Schmidt Hammer test				
Windsor Probe test				
Compressive strength test				
Crackmeters				

Rating Scale Low Medium Moderate Moderate-High High

A3.2.2.1.4 Different concrete qualities

LiDAR, GPR, Schmidt Hammer test, Windsor Probe test, Compressive strength test, Acoustic Emission, Neutron Radiography.

Surveying	Detec	tion	Accuracy		Availability	Versat	ility
Techniques	effectiv	eness					
GPR							
Lidar							
Schmidt Hammer test							
Windsor Probe test							
Compressive Strength test							
Acoustic Emission							
Neutron Radiography							
Rating Scale	Low	Low Medi	ium Modera	ate	Moderate-	High	Hig

A3.2.2.1.5 Delamination of concrete layers

GPR, LiDAR, Acoustic emission, Guided Waves propagation, Magnetic Memory Method, Magnetic Flux Leakage, Magnetic Pulse Induction, Boroscopy, Endoscopy, Infrared Thermography, Impact-echo, Schmidt Hammer test, Neutron Radiography.





Surveying	Detection	Accuracy	Availability	Versatility
Techniques	effectiveness			
GPR				
Lidar				
Acoustic Emission				
Guided Waves Propagation				
Magnetic Memory Method				
Magnetic Flux Leakage				
Magnetic Pulse Induction				
Boroscopy, Endoscopy				
Infrared Thermography				
Impact-echo				
Schmidt hammer test				
Neutron Radiography				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High

A3.2.2.1.6 Anchor failure GPR, LiDAR, Clinometers.

Surveying Techniques	Detection effectiveness	Accuracy	Availability	Versatility
GPR				
LiDAR				
Clinometers				

Rating Scale	Low	Low Medium	Moderate	Moderate-High	High



Appendix A4 Data analysis methods for specific surveying technologies

	Data analysis methods for the specific surveying technology
Surveying	- Best practice or state of the art
technology	- Examples of novel/recent approaches
Visual inspection	Task: Defect detection and segmentation on high resolution images of concrete structures Best practice/state-of-the-art: EN 13018:2016 Novel/Recent approaches: Literature is mostly focused on detection of cracks, also because public benchmark datasets exist mostly for this defect type. State-of-the art results are obtained using deep learning architectures. Lately a novel Crack Transformer network (CrackFormer) for fine-grained crack detection was proposed [1]. It is composed of novel attention modules in a SegNet-like encoder-decoder architecture and new scaling-attention modules to combine outputs from the corresponding encoder and decoder blocks to suppress non-semantic features and sharpen semantic ones. Further references: Deep hierarchical CNN (DeepCrack), to predict pixel-wise crack segmentation
	In an end-to-end method [2].
Boroscopy	Best practice/state-of-the-art: EN 13018:2016 Novel/Recent approaches: A deep learning framework [3] based on state- of-the-art Fully Convolutional Networks (FCN) was proposed to identify and locate damages from borescope images. The framework successfully detected two major types of damages (cracks and burns) in borescope images and extracted their regions with high prediction accuracy. The framework was further optimized to significantly reduce the amount of training data by applying fine-tuning methods.
Water penetration	Task : Determine maximum range of water penetration into concrete element from the surface
Weight-in-motion method	 Task: Determine the weight and wheel loads of the vehicles Focus: Assess the quality of data acquired by the WIM system in terms of accuracy, outlier detection and error analysis. Best practice/state-of-the-art: State of the art in the measurement of the accuracy referred to weights and static loads [4] is based on the application of statistical analysis with calculation of mean, standard deviation, confidence level, confidence interval and tolerance. WIM data for the accuracy test are acquired using instrumented calibration lorries. Novel/Recent approaches: This paper [5] describes a method based on determining the Steering Axle Load Spectra (SALS) to investigate the accuracy of 77 operative weigh-in-motion stations. In order to include several factors—such as the type of axle load sensor, pavement temperature and vehicle speed—various cases of SALS were derived using a series of filters. Means and standard deviations were then calculated on normally distributed SALS (Lilliefors test at 99.9% level). By analysing the impact of temperature and vehicle speed on the cumulative distributions of mean values of SALS, the authors concluded that systematic error in WIM measurement can be the result of temperature change.
Fibre optic sensors	Task: Measure the deformation of concrete surface Best practice/state-of-the-art; EN IEC 61757-1-1:2020-12





	Data analysis methods for the specific surveying technology
Surveying	- Best practice or state of the art
technology	- Examples of novel/recent approaches
	- Further references
	Novel/Recent approaches : Recently, Cross Correlation Analysis (CCA) and
	Robust Regression Analysis (RRA) were applied to data generated on a
	laboratory model of a four-span highway bridge under different global and
	local damage scenarios [6]. A system of Fibre Bragg Grating (FBG) strain
	sensors was used to capture SHM data under different temporal damage
	amongst all possible individual pairs of sensors were created for the baseline
	(un-damaged) and the damaged condition. Statistically significant changes of
	sensor pairs cross-correlation coefficients with respect to the baseline during
	monitoring helped detect and locate the damage. In RRA, only sensor pairs
	that have high correlation were chosen and then rearranged into a new
	matrix. Only correlations of these pairs were used during the monitoring
	phase to detect any abnormal behaviour within a robust regression model.
	Both methods detected the damages. RRA outperformed CCA in terms of
	time-to-detection but also resulted more computationally expensive.
	Further references:
	Comparison of acquired data vs model, calibration of the model,
	determination of the density function of continuous measured data with
	genetic algorithms [7].
Endoscopy	Rest practice/state_of_the_art: EN 13018:2016
	Task: Detect debonding loss of diameter in rebars
	Best practice/state-of-the-art: BS 1881, DIN 1045
	Novel/Recent approaches : The Magnetic Force Induced Vibration
	Evaluation [8] (M5) is a unique mechanical and electromagnetic method that
	allows examining the bond between rebars and concrete with higher
Magnetic and	sensitivity. The M5 method can be considered as a type of modal analysis
Electrical Methods	using magnetic coupling and electromagnetic excitation. Using a strong
	magnetic coupling (excitation element with alternating magnetic field, strong
	magnet attached to the detection element), vibrations are induced only in the
	inspection of the frequency spectrum of the measured signal. Signal changes
	hetween a healthy rehar and a de-honded one are observed
Radioactive and	Task : Structural integrity of concrete samples (Loss of section, delamination)
Nuclear Methods	Best practice/state-of-the-art: ASTM D2950-91. NDIS 1401-1992
	Task: Crack detection in the damaged area
Guided waves	Best practice/state-of-the-art: BS 9690-1:2011, ASTM E2775 –16, ISO
rechniques	18211:2016
Surface	Task : Concrete strength, Hardness of the surface
measurements	Best practice/state-of-the-art: EN 12 504-2, ASTM C805-97 STM, ISO/DIS
	8046
	Task: Location of active destructive processes from emission of acoustic
	Signals. Best prestice/state of the art: EN 12554:2011 ASTM E2100 17
	Novel/Recent approaches: The waveform-based analysis approach is
Acoustic Emission	believed to be better than traditional parameter-based approach in source
	discrimination
	Acoustic emission methods are usually based on a first denoising or signal
	conditioning step, in which the recorded waveforms are processed using
	Fourier (frequency domain), short-time Fourier (time-frequency domain),





O	Data analysis methods for the specific surveying technology
technology	- Best practice or state of the art - Examples of novel/recent approaches
	- Further references
	 wavelet or fast-wavelet transforms. Latest literature agrees with the superiority of the wavelet over Fourier Transform based filter techniques [9]. This signal feature extraction step is the eventually followed by location analysis, cluster analysis of locations in physical space, measurement of variance and energy of location clusters, cluster analysis of locations in feature space, visualisation for final fracture identification [10]. Since similar source mechanisms emit similar signals, search for similarity is used for source discrimination. Cross-correlation coefficients in time domain and magnitude squared coherence (MSC) in frequency domain [11] are two proposed algorithms for source discrimination. Further references: Method based on image correlation [10] Use of wavelet algorithms and Coherence Functions [9]
Water resistance	Task: Water absorption of concrete as a measure of resistance against carbonation and chloride migration Best practice/state-of-the-art : PN-EN 772-11:2011
Qualitative chemical methods	Task : Distribution of the pH changes on the sample, carbonation Best practice/state-of-the-art : EN 14630:2006, EN 1239-12:2020, ASTM C876
Quantitative chemical methods	 Task: determine the presence of cracks or the loss of section in rebars from chloride ion concentration in steel reinforced concrete (wet solution, x-ray fluorescence spectroscopy (XRF)) Best practice/state-of-the-art: AS 1012.20-1992, EN 14629:2008, PN-EN 1767:2008, ASTM C 1202 Novel/Recent approaches: Additional references: Big data in quantitative chemistry [12] (PCA, Factor Analysis, Wavelet transform, Machine learning) Recent approaches of Factor Analysis in chemistry [13] Machine learning in electrochemistry [14]
Mechanical tests on cored samples	Task: Best practice/state-of-the-art: EN 12390-3:2001, EN 12504-1:2002, ASTM C900-94 STM, ISO 5725-1 Novel/Recent approaches: Verification of statements concerning the random variations of mechanical behaviour of concrete cores, supported with statistical approaches, is important. Statistically significant data sets under such conditions are very rare. In this study [15], the issue is addressed by exploring the mechanical behaviour of cores using probabilistic concepts. Two-parameter Weibull, normal and log-normal distributions are used to fit the test data (strength, elastic modulus and strain at peak stress) of concrete specimens. The Weibull distributions most accurately describe the experimentally measured data. A basic theory of damage mechanics is introduced to deal with stress-strain behaviour of cores. In this theory, a statistical method is used to describe mechanical properties on a mesoscopic scale in order to generate realistic behaviour at a macroscopic scale. It is shown that there is a relatively good coincidence between the theoretical results and the measured data.
Mems	Task: Bost practice/state of the art:
accelerometer	Dest practice/state-or-the-art:





Surveying	Data analysis methods for the specific surveying technology - Best practice or state of the art
technology	 Examples of novel/recent approaches Eurther references
	 Vibrational parameters: 1. Statistical analyses can be performed at sensor level (evaluation of the sensor vibrations), component level (analysis of data collected by all the sensors installed on the same element), and system level (analysis of data collected by all the sensors installed on the same structure). 2. As a result, the following quantities can be estimated: Sensor, element or structure average vibrational levels Anomalous vibrations induced by localized damages (e.g., road pavements defect) Presence of exceptional loads or accidental actions Earthquake detection 3. Modal parameters [16,17]: 4. Structure modal shapes estimation 6. Damage detection in operational conditions, by evaluating changes in time of frequencies values/damping ratios/modal shapes 7. Structure health status evaluation after an exceptional event, as an earthquake. Novel/Recent approaches: Machine learning for the detection of vibrational levels
Mems clinometer	Task:Best practice/state-of-the-art:8.Data analysis methods on clinometers data mainly consist in statisticalmethods, aimed to evaluate structure residual deformations due todegradation/damage mechanisms (plastic deformations, settlements, bearingsdeformations, structural joints defects, etc.) and evolving deformations. It usedalso to estimate the structure deformed shape under load.Novel/Recent approaches:
Satellite	 Task: Change detection Best practice/state-of-the-art: Due to the limited resolution of optical satellite images, structural failures can only be detected as "changes" over infrastructure images acquired overt time. Heuristic methods are based on simple thresholds (decided by experts), usually applied to determine if there is a change in the infrastructure. This paper, for example, extracts joint planar-vertical features to delineate the structure of interest and applies a multi-temporal change detection model to simultaneously capture the structure change over time [18]. The detected changes are usually referred as rupture, deformation and displacement interchangeably. Novel/Recent approaches: Machine learning can be used to determine changes in the image over time. This paper [19] describes an optimized architecture (EffCDNet) which adopts a siamese-based pre-trained encoder with an Attention-based UNet decoder for semantic segmentation. The network is built with pre-trained EfficientNet architecture with shared weights to extract robust features and to overcome the limitations caused by insufficient training data. The attention-based UNet decoder uses the attention-gate layer mechanism right before concatenation to obtain more discriminative relevant features and improve segmentation.





Surveying	Data analysis methods for the specific surveying technology			
technology	- Examples of novel/recent approaches			
	- Further references			
	performance. This allows the user to reconstruct fine-grained feature maps with significant context information. To obtain enhanced information difference map, the Undecimated Discrete Wavelet Transform (UDWT) fusion was used as a post-processing technique for spatial and temporal analysis of multi-resolution images. Further references : Alternative machine learning approaches [20,21].			
	Task : Detect damages in concrete (cracks, spalling, etc.)			
LiDAR	 Best practice/state-of-the-art: Damages are detected using heuristic methods. Point clouds from reconstruction are compared to previously acquired point clouds of the same infrastructure or to the model of the structure [22]. Novel/Recent approaches: This paper [23] presents a framework for automated defect inspection of concrete structures, that comprises LiDAR data collection, defect detection, scene reconstruction, defect assessment and data integration stages. Deep learning algorithms are implemented to efficiently detect defects from the collected LiDAR images, and a simultaneous localization and mapping algorithm is adopted for site reconstruction. Based on the images of detected defects, assessment is conducted to evaluate the defect conditions, complemented with the defect dimensions estimated from the aligned image and LiDAR data. The defect position could also be mapped to the respective structural elements and the inspection results integrated into existing Building Information Modelling files. The proposed workflow was validated using a case study for determining concrete cracks and spalls in a real-world facility. Further references: Heuristic [24] 			
	Alternative machine learning approaches [25,26]			
GPR	Task: Best practice/state-of-the-art: Heuristic methods are based on comparing the measured GPR signal to synthetic GPR data [27]. GPR data are often integrated with other complementary NDT data. Some additional imaging techniques like textural analysis or Hilbert transforms are used to highlight some signal attributes (amplitude, frequency, attenuation). The most typical machine learning techniques, based on detecting hyperbolic reflections or pattern recognition, are Support Vector Machine (SVM), Genetic algorithms, Fuzzy logic and Hidden Markov model (HMM). Histogram of oriented gradients (HOG) may be used for feature extraction. Novel/Recent approaches : Machine learning approaches are based on object and damage detection through CNNs and RNNs. In this example (28), a Mask Region-based Convolutional Neural Network (R-CNN) is improved by incorporating a new loss function for detection and segmentation. This new loss function is called <i>distance guided intersection over union</i> (<i>DGIoU</i>) and considers the center distance between two bounding boxes, to overcome the weakness of intersection over union (<i>IoU</i>) in training and evaluation. In addition, a new method is proposed to extract data points from the segmented mask patches containing both object signatures and background noises. The extracted data points can be further processed for object localization and characterization. Experiments conducted using GPR scans			





Surveying technology	Data analysis methods for the specific surveying technology - - Best practice or state of the art - Examples of novel/recent approaches - Further references
	collected from a concrete bridge deck demonstrate that the hyperbolic signatures of rebars can be accurately detected and segmented using the proposed method.



Appendix A5 References

[1]. Liu H, Miao X, Mertz C, Xu C, Kong H. CrackFormer: Transformer Network for Fine-Grained Crack Detection. :10.

[2]. Liu Y, Yao J, Lu X, Xie R, Li L. DeepCrack: A deep hierarchical feature learning architecture for crack segmentation. Neurocomputing. 2019 Apr;338:139–53.

[3]. Shen Z, Wan X, Ye F, Guan X, Liu S. Deep Learning based Framework for Automatic Damage Detection in Aircraft Engine Borescope Inspection. In: 2019 International Conference on Computing, Networking and Communications (ICNC). 2019. p. 1005–10.

[4]. Jacob B. ACTION COST323: WEIGH-IN-MOTION OF ROAD VEHICLES. In 1998 [cited 2022 Jan 13]. Available from: https://trid.trb.org/view/512347

[5]. Rys D. Investigation of Weigh-in-Motion Measurement Accuracy on the Basis of Steering Axle Load Spectra. Sensors. 2019 Jul;19(15):3272.

[6]. Malekzadeh M, Catbas FN. A comparative evaluation of two statistical analysis methods for damage detection using fibre optic sensor data. International Journal of Reliability and Safety. 2014;8(2/3/4):135.

[7]. Del Grosso A, Bergmeister K, Inaudi D, Santa U. Monitoring of bridges and concrete structures with fibre optic sensors in Europe. IABSE REPORTS. 2001;238–9.

[8]. Frankowski PK, Chady T, Zieliński A. Magnetic force induced vibration evaluation (M5) method for frequency analysis of rebar-debonding in reinforced concrete. Measurement. 2021;182:109655.

[9]. Grosse CU, Finck F, Kurz JH, Reinhardt HW. Improvements of AE technique using wavelet algorithms, coherence functions and automatic data analysis. Construction and Building Materials. 2004 Apr;18(3):203–13.

[10]. Pullin R, Eaton MJ, Hensman JJ, Holford KM, Worden K, Evans SL. Validation of Acoustic Emission (AE) Crack Detection in Aerospace Grade Steel Using Digital Image Correlation. Applied Mechanics and Materials. 2010 Jun;24–25:221–6.

[11]. Kaphle MR, Tan ACC, Thambiratnam DP, Chan THT. STUDY OF ACOUSTIC EMISSION DATA ANALYSIS TOOLS FOR STRUCTURAL HEALTH MONITORING APPLICATIONS. Proceedings of the 20th International Acoustic Emission Symposium. 2010;8.

[12]. Dumancas GG, Bello GA, Hughes J, Murimi R, Viswanath LCK, Orndorff CO, et al. Visualization Tools for Big Data Analytics in Quantitative Chemical Analysis: A Tutorial in Chemometrics. In 2018 [cited 2022 Jan 17]. Available from: https://www.igi-global.com/chapter/visualization-tools-for-bigdata-analytics-in-quantitative-chemical-analysis/www.igi-global.com/chapter/visualization-tools-forbig-data-analytics-in-quantitative-chemical-analysis/198789

[13]. Wentzell PD, Giglio C, Kompany-Zareh M. Beyond principal components: a critical comparison of factor analysis methods for subspace modelling in chemistry. Analytical Methods. 2021;13(37):4188–219.

[14]. Mistry A, Franco AA, Cooper SJ, Roberts SA, Viswanathan V. How Machine Learning Will Revolutionize Electrochemical Sciences. ACS Energy Letters. 2021 Apr;6(4):1422–31.

[15]. Wu S, Zhou J. Analysis of mechanical properties of concrete cores using statistical approach. Magazine of Concrete Research. 2013 Dec;65:1463–71.

[16]. Pakzad SN, Fenves GL. Statistical Analysis of Vibration Modes of a Suspension Bridge Using Spatially Dense Wireless Sensor Network. Journal of Structural Engineering. 2009 Jul;135(7):863–72.

[17]. Ubertini F, Gentile C, Materazzi AL. Automated modal identification in operational conditions and its application to bridges. Engineering Structures. 2013 Jan;46:264–78.

[18]. Huang X, Cao Y, Li J. An automatic change detection method for monitoring newly constructed building areas using time-series multi-view high-resolution optical satellite images. Remote Sensing of Environment. 2020 Jul;244:111802.

[19]. Patil PS, Holambe RS, Waghmare LM. EffCDNet: Transfer learning with deep attention network for change detection in high spatial resolution satellite images. Digital Signal Processing. 2021 Nov;118:103250.





[20]. Zheng Z, Zhong Y, Wang J, Ma A, Zhang L. Building damage assessment for rapid disaster response with a deep object-based semantic change detection framework: From natural disasters to man-made disasters. Remote Sensing of Environment. 2021 Nov;265:112636.

[21]. Anantrasirichai N, Biggs J, Kelevitz K, Sadeghi Z, Wright T, Thompson J, et al. Detecting Ground Deformation in the Built Environment Using Sparse Satellite InSAR Data With a Convolutional Neural Network. IEEE Transactions on Geoscience and Remote Sensing. 2021 Apr;59(4):2940–50.

[22]. Kim MK, Sohn H, Chang CC. Localization and Quantification of Concrete Spalling Defects Using Terrestrial Laser Scanning. Journal of Computing in Civil Engineering. 2015 Nov;29(6):04014086.

[23]. Chow JK, Liu K fu, Tan PS, Su Z, Wu J, Li Z, et al. Automated defect inspection of concrete structures. Automation in Construction. 2021 Dec;132:103959.

[24]. Jiang H, Li Q, Jiao Q, Wang X, Wu L. Extraction of Wall Cracks on Earthquake-Damaged Buildings Based on TLS Point Clouds. IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing. 2018 Sep;11(9):3088–96.

[25]. Feng H, Li W, Luo Z, Chen Y, Fatholahi SN, Cheng M, et al. GCN-Based Pavement Crack Detection Using Mobile LiDAR Point Clouds. IEEE Transactions on Intelligent Transportation Systems. 2021;1–10.

[26]. Wu H, Yao L, Xu Z, Li Y, Ao X, Chen Q, et al. Road pothole extraction and safety evaluation by integration of point cloud and images derived from mobile mapping sensors. Advanced Engineering Informatics. 2019 Oct;42:100936.

[27]. Abouhamad M, Dawood T, Jabri A, Alsharqawi M, Zayed T. Corrosiveness mapping of bridge decks using image-based analysis of GPR data. Automation in Construction. 2017 Aug;80:104–17.

[28]. Hou F, Lei W, Li S, Xi J, Xu M, Luo J. Improved Mask R-CNN with distance guided intersection over union for GPR signature detection and segmentation. Automation in Construction. 2021 Jan;121:103414.





REVIEW OF METHODOLOGIES AND INSTRUMENTS FOR DIAGNOSTICS OF TRANSPORT INFRASTRUCTURE APPENDIXES

Annex B - Damage characterization procedure





This project has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 958171. © 2021 IM-SAFE-Proiect | TNO. The Netherlands. All right reserved |





Appendix B1 Surveying technologies – Phases

Phase	Phase 1	Phase 2	Phase 3
Accelerometers			0
Abrasion resistance test		0	
Acoustic Emission			0
Argentometric Titration		0	
Boroscopy	0		
Chloride Diffusion test/lon migration test		0	
Compressive strength test		0	
Computed Tomography		0	
Covermeters		0	
Crackmeters			0
Electrical resistivity tomography		0	
Endoscopy	0		
Fatigue test		0	
Fiber Optic Sensors			0
Galvanostatic Pulse Technique		0	
GPR	\bigcirc		
Gravimetric method		0	
Half-cell Potential method		0	
Impact-Echo method		0	
Infrared Spectroscopy		0	
Ion Chromatography		0	
LiDAR	0		0
Magnetic Flux Leakage Method		0	
Magnetic Memory Method		0	
Neutron Radiography		0	
NMR Spectroscopy		0	
Phenolphthalein test		0	
Potentiometric Titration		0	
Pull-out test		0	
Pulsed Eddy Current Method		0	
Rainbow test		0	
Satellite	\bigcirc		\bigcirc



Phase Technology	Phase 1	Phase 2	Phase 3
Schmidt hammer		\bigcirc	
Spectral Analysis of Surface Waves		0	
Tensile test		\bigcirc	
Thymolphthalein test		\bigcirc	
Torque test		\bigcirc	
UAV	0		0
Ultrasonic Pulse Velocity		\bigcirc	
Uranyl-Acetate treatment		\bigcirc	
Water penetration test		\bigcirc	
Water resistance test		0	
Weight-In-Motion Systems			\bigcirc
Windsor Probe test		\bigcirc	

